THE DEVELOPMENT AND APPLICATIONS OF SERIOUS GAMES IN THE PUBLIC SERVICES: DEFENCE AND HEALTH

By Ioannis Th. Paraskevopoulos

A Thesis Submitted for the Degree of Doctor of Philosophy School of Engineering and Design, Brunel University July 2014

ACKNOWLEDGEMENTS

First and foremost, I would like to thank my supervisor and mentor Dr. Emmanuel Tsekleves who apart from being a constant source of support, consultation, guidance, encouragement and wisdom throughout the PhD process, introduced me to the wonderful world of 3D Multimedia and believed in my potentials.

I would also like to thank my second supervisor, Prof. John Cosmas, for his valuable input and support. I would like to also thank my friend and colleague, Dr. Daylamani-Zad for his friendship as well as his support and help which was offered willingly whenever required.

I also need to acknowledge my gratitude to the financial support of the EPSRC, the MoD of UK, the NHS and Brunel University throughout my PhD course. I would also like to thank all the people that gave up their time to participate in my research.

Additionally, a special thanks to the support of my partner, Sara, who was there for me with her love and friendship. Finally, I would like to thank my parents for their unconditional support, love, encouragement and self-denying sacrifices during all those years of my studies.

Ioannis Th. Paraskevopoulos Uxbridge – UK July 2014

TABLE OF CONTENTS

ACKNOW	LEDGEMENTS	II	
TABLE OF CONTENTS			
LIST OF FIGURES			
LIST OF 7	LIST OF TABLES V		
LIST OF F	PUBLICATIONS	VII	
LIST OF A	ABREVIATIONS	IX	
ABSTRAC		X	
STATEMI	ENT OF CONTRIBUTION	XI	
СНАРТЕН	R I: INTRODUCTION	1	
1.1 Rese	earch Background	1	
1.2 Rese	arch Motivation	4	
1.3 Aim	s and Objectives	6	
1.3.1 Ai	ms	6	
1.3.2 Ot	ojectives	7	
1.4 Thes	sis Structure	8	
CHAPTER	(II: LITERATURE SURVEY ON SERIOUS GAMES	12	
2.1 Histo	orical origins rview of the application domains	12	
2.2.1 SC	is as an instructional paradigm in education	15	
2.2.2 SC	is for Sustainability promotion / awareness	17	
2.2.3 SC	is in the Defence domain	20	
2.2.4 SC	Is in the Healthcare domain	22	
2.2.4.1	Exergames	22	
2.2.4.2	Educational and training simulation and games	23	
2.2.4.3	SGs used as tools for diagnosis, management, recovery and rehabilitation of		
cognitiv	e and motor impairments	23	
2.3 Rela	ted Work	24	
Defence Sim	ulations with Sustainability: Virtual Reality Simulations and Guidelines for	• the	
Integration of	f Photovoltaic Technology on the Modern Infantry Soldier	25	
2.3.1 511	VB in Defence applications	25	
2.3.1.1	VR in Defence applications	20	
2.3.1.2	Simulation of solar light harvesting	21	
2.3.2 Cu	stomised Serious Games and game sensors for stroke rehabilitation	32	
2.3.2.1	Reviews on VR for stroke rehabilitation	32	
2.3.2.2	Customised Stroke Rehabilitation interventions	35	
2.3.3 Cu	stomised Serious Games and game sensors for PD	37	
2.3.3.1	Review of Customised Interventions for PD rehabilitation	38	
2.3.3.2	Review of Design Guidelines	41	
СНАРТЕН	R III: SERIOUS GAMES AND SIMULATIONS FOR DEFEN	CE	
APPLICA'	TIONS FOR ENERGY SUSTAINABILITY	44	
3.1 Cha	llenges and opportunities for sustainable practice in modern military	44	
3.2 The	Simulation Platform	46	
3.2.1 Ex	perimental study of system accuracy	47	
3.2.1.1	Experiment Methodology	48	
3.2.1.2	Simulation Setup	49	
3.2.1.3	Lighting Analysis Rendering Setup	49	
3.2.1.4	Lighting System Setup	49	
3.2.1.5	Material Adjustments	50	
3.2.1.6	Accuracy Study Results	50	
5.3 Imp	lementation of the light analysis platform	53	
2.2.1.1VI	The Lighting system	33 54	
J.J.1.1 2 2 1 0	Pascale of the Analysis Value colour coding	50 57	
5.5.1.2	Researe of the marysis value colour couning	57	

3.3.1.3 The Lighting analysis Render settings	57
3.3.1.4 Materials	57
3.3.1.5 The expansion script	58
CHAPTER IV: SERIOUS GAMES AND SIMULATIONS FOR DE	FENCE
APPLICATIONS FOR ENERGY SUSTAINABILITY	62
4.1 Serious Games and Simulations for Defence applications with Sustainabil	ity 62
4.1.1 Simulation Scenarios	62
4.1.1.1 Power assessment and investigation of integration areas	69
4.1.1.2 Combining quantitative methods with user-led design and feedback	71
4.1.2 Results	72
4.1.3.1 Power generation assessment	74
4.1.3 Design Guidelines for the Integration of Photovoltaic Technology on the Mo	odern
Infantry Soldier	75
CHAPTER V: SERIOUS GAMES FOR MOTOR REHABILITATI	ON 78
5.1 A rehabilitation framework for Motor deficits utilising Serious Games	78
5.1.1 Serious Games for Stroke	80
5.1.1.1 Summary of related work: challenges and opportunities	81
5.1.2 Serious Games for Parkinson's Disease	82
5.1.2.1 Summary of related work: challenges and opportunities	83
5.1.3 Emerging themes for developing Serious Games for motor rehabilitation	85
5.2 The Framework	86
5.2.1 Motion Capture System	89
5.2.2 System accuracy study: Methodology and experiment setup	90
5.2.2.1 Experimental results and discussion	91
5.2.3 Automatic Calibration and difficulty adjustment system	94
5.3 Implementation of Serious Games for Motor Rehabilitation	97
5.3.1 Pilot games for Stroke rehabilitation	101
5.3.1.1 The Virtual Teacher	101
5.3.1.2 Hitting the Spheres	104
5.3.1.3 Air Hockey Game	107
5.3.2 Pilot Games for Parkinson's Rehabilitation	109
5.3.2.1 The Rowing game	110
5.3.2.2 The Steam Mini-golf game	116
CHAPTER VI: SERIOUS GAMES FOR MOTOR REHABILITAT	ION 122
6.1 Serious Games for Motor Rehabilitation	122
6.1.2 Stroke Rehabilitation Serious Games	122
6.1.2.1 Results and Discussion	124
6.1.3 Parkinson's Disease Rehabilitation Serious Games	127
6.1.3.1 Results and Discussion	131
6.2.3.1.1 Design Guidelines	136
6.1.4 Overall discussion of the Evaluation of Serious Games for Motor Rehabilita	tion 138
CHAPTER VII: A METHODOLOGICAL APPROACH FOR SERI	OUS
GAMES DESIGN	141
CHAPTER VIII: CONCLUSION AND FUTURE WORKS	141
8.1 Thesis Summary	146
8.2 Thesis Conclusion	147
8.3 Future work	149
0.3.1 The Solar Solater scenario generator 9.2.2 Home based motor rebabilitation with monitoring and hilliting	149
o.5.2 nonne-based motor renabilitation with monitoring capabilities	150
0.4 Impact and Vision DEFEDENCES	150 150
A DENINU A DENINU	154
AFFENDIA Annondig A. The Solar Soldier simulation results - Average Light Howcord	100
Appendix A. The Solar Soldier simulation results – Average Light Harvest	100 171
Appendix C. Patient Questionnaire	180

LIST OF FIGURES

Figure 1 The overlapping interplay of pedagogy, computer science and games [5]	3
Figure 2 Venn diagram of types of Game-Based Learning [34]	17
Figure 3 Block diagram of the adopted methodology	47
Figure 4 3D model of Brunel University with the weather station location	48
Figure 5 Scenario 1 – Sunny Day	52
Figure 6 Scenario 2 – Partially Cloudy Day	52
Figure 7 Scenario 3 – Cloudy Day	53
Figure 8 Top-Down approach diagram of the development	54
Figure 9 Urban area scene	55
Figure 10 Forest scene	55
Figure 11 Military camp scene	56
Figure 12 Flowchart of the developed script	59
Figure 13 Code snippet of the script	60
Figure 14 The positioning of the virtual light sensors (front)	63
Figure 15 The positioning of the virtual light sensors (back)	63
Figure 16 Major british troops deployments (Source: British Broadcasting Corporation	on)65
Figure 17 The scenarios map	68
Figure 18 Military scene orientation	69
Figure 19 Areas of integration	70
Figure 20 Ross Kemp in Afganistan (Source: British Sky Broadcasting Group plc 20	(80
	71
Figure 21 Helmet sensor light levels for all locations, 26th of October	72
Figure 22 Block diagram of the adopted methodology	86
Figure 23 Schematic diagram of the system architecture	88
Figure 24 Accuracy experiment setup	91
Figure 25 x-axis motion data	92
Figure 26 y-axis motion data	92
Figure 27 z-axis motion data	93
Figure 28 Data flow diagram of the Motion Data Calibrator / Adaptor	
Figure 29 Code snippet	
Figure 30 Generic Data flow diagram.	
Figure 31 The Interface component	98
Figure 32 Code snippet of the Socket Listener	99
Figure 33 Code snippet of Data Process Unit	100
Figure 34 Sketch of the Virtual Teacher game	102
Figure 35 The Virtual Teacher game environment	102
Figure 36 Data flow diagram of the VR Teacher game	103
Figure 37 Code snippet of the data mapping	104
Figure 38 Data flow diagram of Hitting the Spheres game	105
Figure 39 Sketch of the Hitting the Spheres game	106
Figure 40 The Hitting the Spheres game environment	107
Figure 41 Data flow diagram of the Air Hockey game	108
Figure 42 The Air Hockey game environment	100
Figure 42 Arm reaching evercice [174]	110
Figure 44 Sketch of the Rowing game	110
Figure 45 The Rowing game environment version 0.1	112
Figure 46 The Rowing game environment version 0.2	112
Figure 47 Data flow diagram of the Powing game	112
Tigute +/ Data now diagram of the Rowing game	113

Figure 48 Code snippet of the audio cue	. 114
Figure 49 Rowing game – the development environment	. 115
Figure 50 Code snippet of the events manager	. 116
Figure 51 Rolling a ball exercise [174]	. 116
Figure 52 Sketch of Steam mini-golf game	. 117
Figure 53 The Staem Mini-golf game version 0.1	. 117
Figure 54 The Steam Mini-golf game version 0.2	. 118
Figure 55 Data flow diagram of the Steam Mini-golf game	. 119
Figure 56 Steam Mini-golf game – the development environment	. 119
Figure 57 Code snippet of the Game Events manager	. 121
Figure 58 The single-case expert playing the Virtual Teacher game	. 123
Figure 59 Expert patient playing the Virtual Teacher game	. 127
Figure 60 Expert patient playing the Rowing game	. 129
Figure 61 Expert pateint playing the Steam mini-golf game	. 130
Figure 62 Thematic Mind map of key themes from user trials	. 139
Figure 63 Generic system architecture	. 142
Figure 64 Generic Methodology for Design, Development and Research for Serious	
Games	. 144
Figure 63B Military base scene in Pristina	. 176

LIST OF TABLES

Table 1 Lighting Simulation Setup Scenarios	49
Table 2 U.K., Catterick Garrison Headquarters, N. Yorkshire (54.375, -1.708). Light	
intensity in Lux	67
Table 3 Kosovo, Pristina (42.5, 20.9). Light intensity in Lux	67
Table 4 Iraq, Baghdad (33.33, 44.444). Light intensity in Lux	67
Table 5 Sensor classification per scene	73
Table 6 Overall classification of sensors	74
Table 7 Overall classification in terms of power generation	75
Table 8 Demographic information of the patient experts	123
Table 9 Demographic information of the single-case study expert	123
Table 10 Participants characteristics	129
Table 11 Participants Quotes	131
Table 12 Summarised feedback and suggestions for the Rowing game	132
Table 13 Summarised feedback and suggestions for the Steam mini-golf game	135

LIST OF PUBLICATIONS

JOURNALS

Paraskevopoulos, I. T., Tsekleves, E., Craig, C., Whyatt, C., Cosmas, J. P. (2014), Design Guidelines for developing customised Serious Games for Parkinson's Disease rehabilitation using bespoke game sensors, in *Entertainment Computing, Elsevier*

Tsekleves, E., **Paraskevopoulos, I. T.**, Warland, A., Kilbride, C. (2014), Development and preliminary evaluation of a low cost VR-based stroke rehabilitation platform that uses the Wii technology, in *Disability and Rehabilitation: Assistive Technology Journal*.

Paraskevopoulos, I., Tsekleves, E. (2012). Simulating the integration of photovoltaic technology on the modern infantry soldier using modelling and simulation: scenarios and guidelines. *The Journal of Defense Modeling and Simulation: Applications, Methodology, Technology*, 1548512912458194.

BOOK CHAPTERS

Paraskevopoulos, I., Tsekleves, E. (2014). Solar soldier: virtual reality simulations and guidelines for the integration of photovoltaic technology on the modern infantry soldier. In *Simulation and Modeling Methodologies, Technologies and Applications* (pp. 141-154). Springer International Publishing

Tsekleves, E., Warland, A., Kilbride, C., **Paraskevopoulos, I.,** Skordoulis, D. (2014). The use of the Nintendo Wii in motor rehabilitation for virtual reality interventions: a literature review. In *Virtual, Augmented Reality and Serious Games for Healthcare 1* (pp. 321-344). Springer Berlin Heidelberg.

CONFERENCE PROCEEDINGS

Paraskevopoulos, I., Tsekleves, E. (2013). Use of gaming sensors and customised exergames for Parkinson's Disease rehabilitation: a proposed virtual reality framework. In *Games and Virtual Worlds for Serious Applications (VS-GAMES), 2013 5th International Conference on* (pp. 1-5). IEEE. **Best IEEE member paper award**

Warland, A., Kilbride, C., Tsekleves, E., Skordoulis, D., **Paraskevopoulos, I.** (2012). ReWiiRe (Research in Wii Rehabilitation): user involvement in the development of a personalised rehabilitation system for arm re-education after stroke. *International Journal of Stroke*, 7(Suppl.), 27-27.

Tsekleves, E., Skordoulis, D., **Paraskevopoulos, I**., Kilbride, C., Warland, A. (2012). Personalised stroke rehabilitation intervention using open source 3D software and the Wii Remote. Plus. *Proc. 9th Intl Conf. Disability, Virtual Reality & Associated Technologies (ICDVRAT)* Laval, France, 10–12 Sept. 2012. pp. 515-518

Paraskevopoulos, I., Tsekleves, E. (2012). Simulation Of Photovoltaics For Defence: Applications Power Assessment and Investigation of the available integration areas of photovoltaic devices on a virtual infantry man. *International Conference on Simulation and Modeling Methodologies, Technologies and Applications (SIMULTECH).* July 2012. **Best Student paper award**

Paraskevopoulos, **I.**, Tsekleves, E. (2011) Simulation of Photovoltaics for Defence and Commercial Applications by Extending Existing 3d Authoring Software - A Validation Study. *International Conference on Simulation and Modeling Methodologies*, *Technologies and Applications (SIMULTECH)*. Netherlands, 28-31 July 2011

Tsekleves, E., Skordoulis, D., **Paraskevopoulos, I.,** Kilbride, C. (2011). Wii your health: a low-cost wireless system for home rehabilitation after stroke using Wii remotes with its expansions and blender. In *Proceedings of The 8th IASTED International Conference on Biomedical Engineering (Biomed 2011), Innsbruck, Austria* (pp. 2011-723).

LIST OF ABBREVIATIONS

3D	Three-Dimensional
3DSMD	3D Studio Max Design
AA	America's Army
ADHD	Attention Deficit Hyperactivity Disorder
BIPV	Building-Integrated PhotoVoltaic
COTS	Commercial Off-The-Shelf
DoF	Degrees of Freedom
DSTS	Dismounted Soldier Training System
EPSRC	Engineering and Physical Sciences Research Council
GBL	Game-Based Learning
HCD	Human-Centred Design
HCI	Human Computer Interaction
IR	InfraRed
LED	Light-Emitting Diode
MDA	Mechanics – Dynamics - Aesthetics
MDS-	Movement Disorder Society's Unified Parkinson's Disease Rating
UPDRS	Scale
МоСар	Motion Capture
MoD	Ministry of Defence
NHS	National Health System
PD	Parkinson's Disease
PIPV	Product-Integrated PhotoVoltaic
PTSD	Post Traumatic Stress Disorder
PV	PhotoVoltaic
ROM	Range Of Movement
SD	Sustainable Development
SG	Serious Game
UK	United Kingdom
UPDRS	Unified Parkinson's Disease Rating Scale
VG	Video Game
VR	Virtual Reality

ABSTRACT

The latest advances of Virtual Reality technologies and three-dimensional graphics, as well as the developments in Gaming Technologies in the recent years, have stemmed the proliferation of Serious Games in a broader spectrum of research applications.

Among the most popular areas of application are public services such as Defence and Health, where digital technologies realise new challenges and opportunities for research and development of Serious Games and for a variety of contexts. As with all games, the user engagement is elevated and apart from the entertaining aspect, Serious Games serve as a novel and promising alternative experience to knowledge transfer. Furthermore, Serious Games bring to the end user and the overall society a series of attractive benefits. These benefits include safety, cost-effectiveness, increased motivation and personalisation.

Hence, this Thesis aims to investigate novel approaches of developing Serious Games that utilise the recent advances of Virtual Reality and Gaming Technology and facilitate the aforementioned benefits. The process of design and development of the novel tools and applications follow an iterative manner and are driven by the review of the available literature as well as end-user feedback.

STATEMENT OF CONTRIBUTION

The constant advances of digital economy, specifically of Virtual Reality and Gaming technologies, impose new challenges and opportunities in the development of novel applications for Serious Games. The domains of Serious Games span in a broad range, with the public domain services of Health and Defence among the most popular.

The novel platforms, applications and frameworks are presented in this Thesis, via the utilisation of the state-of-the-art in three-dimensional authoring and gaming technologies that benefit the society in the aspects of sustainable practice in Defence and motor rehabilitation in Health.

The novelty of the work presented in this Thesis includes:

- A bespoke virtual reality simulation platform that allows the simulation of light intensity and yield of power generation for military scenarios with sustainability using photovoltaics as renewable energy sources. The platform is the first to facilitate the light analysis of a virtual environment incorporating movement and animated virtual avatars. It is also re-usable as a product, as the user can easily create new scenarios by importing new virtual environments and animation clips into the platform or by adjusting the daylight system to any possible condition.
- The guidelines for incorporation of the photovoltaic technology on the uniform and equipment of the modern infantry soldier. These guidelines are produced by the range of simulations conducted for the evaluation of the simulation platform as well as qualitative data acquired by the liaising with military experts.
- A framework for development of games for home-based motor rehabilitation utilising virtual reality and gaming technology. The system architecture is implemented in a layered approach to facilitate re-usability of the developed components. The system components have been re-used in a variety of case studies and for different game applications. Furthermore, the system proposed is the first

with the potential to remotely provide accurate biofeedback to the clinician as shown with an experimental study.

 The evaluation of the developed games for home-based motor rehabilitation produced a set of themes and design guidelines for the development of such games. These guidelines enable future game designers to develop games for motor rehabilitation accommodating all the requirements of the targeted cohort of users and with a patient-centred approach.

The proposed research is a substantial step forward in the field of Serious Games that benefit society in the aspects of sustainable practice in Defence and motor rehabilitation in Health by offering innovative uses and applications of technology.

The work conducted for this Thesis is that of the author's, unless otherwise stated.

CHAPTER I: INTRODUCTION

Abstract:

This chapter provides a definition of terms related to the topic of the Thesis and briefly presents the potential application domains of serious games. It offers an overview of the research background and motivation by providing the benefits of serious games to the endusers. Furthermore, the dimensions of serious games and important issues of their development are discussed. Finally, this chapter introduces the scope of this Thesis by presenting the overall structure and brief individual chapter contents. The structure of the chapter is visualised in Diagram 1 below:



Diagram 1 Chapter Structure

1.1 Research Background

Definitions:

"Game: a physical or mental contest, played according to specific rules, with the goal of amusing or rewarding the participant.

Video Game: a mental contest, played with a computer according to certain rules for amusement, recreation, or winning a stake.

Serious game: a mental contest, played with a computer in accordance with specific rules, that uses entertainment to further government or corporate training, education, health, public policy, and strategic communication objectives." [1]

"Gamification: is the use of game design elements in non-game contexts."[2]

Serious Games (SGs) are being applied with distinction in many disciplines and fields including services of general interest as Defence, Health and Education. The latest

advances of Virtual Reality (VR) technologies and three-dimensional (3D) graphics, as well as the developments in the game consoles and technologies in the recent years, have stemmed the proliferation of SG applications in new areas and therefore realising opportunities for a broader spectrum of research applications. Such advances include the development of high-end graphic processors with vast computational power and consequently more robust graphic engines, and novel game sensors such as the Nintendo Wiimote, Microsoft Kinect, the very recent Leap motion, STEM© system by Sixsense and Google glass or Oculus Rift which enable Virtual Presence. Due to that fact, the prospective application of SGs for knowledge transfer has expanded to more areas which in the past seemed to have no connection with computer games of any kind.

As the formal definition proposed in [1] dictates, SGs support training and knowledge transfer in areas as, but not restricted to education, health, engineering, politics, religion or militia. Along the same lines, Michael and Chen [3] define SGs as "games that do not have entertainment, enjoyment, or fun as their primary purpose", although they highlight this fact only to manifest that there is another aim apart from entertainment in those games. Additional terminology describing SGs can be found in the literature, including gamebased learning (GBL), immersive learning, game-based training, game-based simulation, exergames or edutainment. Furthermore, SGs are embodied in a variety of types depending on the form of knowledge to be transferred as well as the particular field of interest. Therefore, within academic principles there are debates on the grasp of SGs and their taxonomy. Crookall [4], in his article for the 40th Anniversary Symposium of Simulation and Games in 2010, clarifies the definitions that have been used over the years in the field and attempts to discuss using a philosophical tone, where the line is drawn. The conclusion of this article is that, Simulation/Gaming or Serious Games (both abbreviated as SG hereafter), can be proclaimed a scientific discipline on its own entity; not just a research subject. The author also suggests that the definitions and niceties can be dealt with later as the potentials are far more important to explore and develop.

An important issue when developing SGs is to consider the aspects and dimensions of them. As a very sophisticated and highly complex set of games that need to deliver knowledge apart from enjoyment, SGs have other dimensions apart from the technical game engineering aspect. According to Martens et al [5], the other aspects of major importance that needs to be taken into account when developing a SG are the pedagogical dimension and the game aspect in terms of the game mechanics. According to the Mechanics – Dynamics – Aesthetics (MDA) game design framework of Hunicke et al.[6], game mechanics are the formal rules of the game, in other words the set of rules to define the game or gameplay and are being discussed in more detail in the following chapters. The underlying didactics of a SG that will serve towards the knowledge transfer are supported by the technical structure of the game. In the same study, Marten et al. also conclude that only if all three aspects are present in a game only then it can be referred to as game-based learning system, or SG. Furthermore, they give a schematic diagram (Figure 1) of the interplay of the three aspects and according to how much these overlap, the SG type is defined.



Figure 1 The overlapping interplay of pedagogy, computer science and games [5] An alternative approach to a SG stratification is that of Prensky [7]. The author in his work

classifies the games according to their complexity in "mini-games" and complex games.

Minigames are very common and trivial whereas complex games require significantly more time to learn and master. The complexity of the pedagogical and technical aspects, as well as how these overlap, define the game type of SGs.

1.2 Research Motivation

The potential applications of SGs have boosted the business sector to which it has been applied, which according to Ben Sawyer [8], co-founder of the Serious Games Initiative, in 2007 was worth \$20 million and according to Mike Hayes [9], CEO of UK-based Serious Games International, will worth \$2.5bn by 2015. Those figures reflect the manifested success and exploitation of SGs in such a wide range of applications and fields. Moreover, the figures suggest the levels of acceptance of SGs by the targeted groups for each individual market.

Nevertheless, scientific research, such as that presented in this Thesis, is not always driven by the industrial facts and numbers. The motivation for researching in this particular field lies within another motivation for developing SGs and that is no other than the benefits they offer to the end-users and the overall society. SGs, apart from their obvious aim to deliver knowledge transfer through an enjoyable and motivating manner, provide a number of other beneficial features. One of them is the replication of a hazardous environment and the simulation of the desirable tasks. For example, a virtual military training or surgical training SGs provides a degree of safety in much greater order of magnitude as compared with a real scenario environment which would involve very high risks which could even result in fatalities. This approach also enables the recording, review, analysis and reenacting of the performance of the player even in scenarios of failure without any risk. Another advantage of SGs is the cost-effectiveness of the learning and training procedures. These are reduced to a minimum due to effective resource management accomplished by the use of SGs in terms of manpower hours or consumables. Furthermore, there are no geographical boundaries and that expands the potentials even more, realising telemonitoring and home-based solutions or simulations of remote geographical locations. Moreover, high level of fidelity introduced by SGs facilitates transference. This way, the attachment of the player with the virtual environment is achieved which by turn increases the engagement and motivation. Another very important aspect of SGs is that they can be modified by making them bespoke according to requirements of each application. This fact results to maintaining the engagement and motivation always at high levels. A review of research studies indicates that computer games and SGs in particular increase engagement and motivation as well as enhancing learning and skills acquisition [10][11]. Therefore, SGs are found to be very beneficial whilst supporting unconventional approaches of learning and training and increasing motivation. According to Prensky [12], game technology might increase the motivation of oneself as game-based learning systems are at least more fascinating than the conventional learning paradigms. This fact tackles the negativity that pre-existed on the impacts of playing video games (VGs) and realised them as a novel alternative way of learning [13]. In their study, Wouters et al. present extensive evidence that SGs have the potential to be a more effective way to learn [14]. However, the same study indicates that not all game aspects play significant role in the effectiveness of the game. Furthermore, this paper proposes that there should be a coordination of the learning targets with the game type. Thus, the measures of SGs effectiveness depend on their aim, design and integration as well as the field of application. For example, the measure of effectiveness of an educational SG is in terms of the knowledge transfer agenda whereas a therapeutic game is effective according to the rehabilitation targets and so on. Each application domain will require different methods of design, development and evaluation. It is, therefore, self-evident that the range of areas of application prescribes a diverse palette of methods and tools to be utilised for the research and development of SGs.

Concluding, the motivation behind this research has been made clear by presenting the benefits of SGs to society. These benefits comprise a set of factors that increase the quality of life of the people using them by a diversity of means to a variety of ends again depending on the particular field of application.

The discussion above also raises new challenges and opportunities for development of novel approaches to SGs in the presence of new platforms, frameworks and humancomputer interaction (HCI) techniques. The aforementioned advances of VR technologies and game consoles and sensors demonstrate a series of those new opportunities and are the cornerstone of the future directions for SGs development.

Motivated and stimulated by the same discussion, this Thesis reports the work conducted in the boundaries of a number of research studies; namely, the Solar Soldier project, ReWiiRe project and Games for Parkinson's Disease (PD) rehabilitation project. The Solar Soldier project is a project jointly funded by the Ministry of Defence (MoD) of the United Kingdom (UK) and Engineering and Physical Sciences Research Council (EPSRC) of UK. The ReWiiRe project (which stands for Research on Wii Rehabilitation) is funded by the National Health System (NHS) of UK. Finally, the Games for PD rehabilitation project was funded by Brunel University and the work was conducted in collaboration with Queen's University in Belfast.

All studies investigated the development of novel applications, tools and frameworks of SGs for each of the corresponding areas of application. The areas of application of the projects are Defence with sustainability simulations and Motor rehabilitation respectively.

A simulation platform that extends the capabilities of a commercial 3D authoring tool to simulate the perceived daylight on an animated agent and in a variety of environmental and lighting conditions for the use of renewable energy sources in Defence applications is the subject of the Solar Soldier project. The use of gaming sensors as motion capture (MoCap) systems and bespoke SGs in VR rehabilitation of motor impaired persons with neurodegenerative or neurophysiological conditions, such as PD or stroke respectively, is investigated in the ReWiiRe and Games for PD rehabilitation studies.

More details about the individual studies, challenges and research implications can be found in Chapters III and V that follow.

1.3 Aims and Objectives

The following sections present the aims and objectives of the Thesis.

1.3.1 Aims

Motivated by the research background and discussions above, the aim of the Thesis can be summarised into the following:

The overall aim of this Thesis is to investigate the use of VR tools and Game Technologies for the development of novel SGs, applications and tools of Defence and Health applications.

The investigated domains and particular studies, conducted in order to accomplish the above, involve individual aims that are outlined in the specific chapters for each study domain (Chapters III and V). In Chapters III and V, the reader can find more details about the challenges, methodologies and implementations of each study.

1.3.2 Objectives

To accomplish the above aim, a series of studies were conducted and as a result a number of SGs were developed and evaluated utilising novel approaches and frameworks. The specific aims and objectives of these studies are presented in the following Chapters (Chapters III and V). However, the novel platforms developed to reach the overall aim of the Thesis entails the following:

- Research of the state-of-the-art of the field and definition of research challenges.
- Experimental study of accuracy of the proposed technology application
- Design and development of the systems, applications and tools by utilising existing VR or VG software, or by extending their capabilities.
- Simulations and pilot testing
- Formation of recommendations and design guidelines based on the results of the evaluation through the simulations and pilot testing.

The purpose of the review of the research literature is to define the state-of-the-art of the field and furthermore, define the research challenges and opportunities for further development.

The purpose of the experimental study of accuracy is to assess the accuracy and efficacy of the proposed systems in order to conclude on the feasibility of the technology application as well as the quality of the results that are collected in a further stage.

The design and development phase includes the design of the system architectures and frameworks, the development of the SGs and the simulation platforms that have the purpose of assessing the specific aims of the individual studies.

The purpose of the simulations and pilot testing is to evaluate the proposed system, applications and tools and to discuss any conclusions that can be drawn by the results of the evaluation.

Finally, the formation of recommendations and design guidelines collates and summarises the results and emerging themes into reusable set of guidelines, for each study, that any professional and practitioner of the field can utilise when designing and developing similar SGs.

1.4 Thesis Structure

The Thesis has been organised in 8 chapters with a more detailed structure by chapter content that follows. Chapter I presents a brief introduction of the overall research background and provides the motivation, aims and objectives of the research.

Chapter II offers a review of the application of SGs in a series of domains. Firstly, there is an overview of SGs and their domains as an introduction to the subject, aiming to a general audience. Following is a more focused review of the available literature, related to the specific areas of the Thesis which are SGs for Defence with Sustainability and SGs for Motor Rehabilitation.

Chapter III provides an insight to the problem description and research challenges, methodology, experimental study of system accuracy and system implementation of the Solar Soldier project which investigated the use of VR simulations to produce guidelines for the integration of PV technology on the modern infantry soldier.

8

Chapter IV describes the methods that were utilised to evaluate the VR simulation platform developed for the Solar Soldier project and described in Chapter III. The results of the evaluation are discussed and summarised in design guidelines.

Chapter V presents the problem description and research challenges methodology, experimental study of system accuracy and system implementation of the SGs developed for Health-related applications, in particular for the SGs for rehabilitation of motor deficits.

Chapter VI describes the methods utilised to evaluate the Serious Games for motor rehabilitation developed and presented in Chapter V. The results of the user trials are presented and discussed. Through that discussion, there are emerged themes and design guidelines for developing games for motor rehabilitation.

Chapter VII presents a methodological framework for Serious Games design that derives from the evaluation of all the platforms and games developed for this Thesis. This chapter also summarises the key observations, challenges and recommendations that were derived from the applications developed.

That discussion also funnels the conclusion of this Thesis, which is offered by Chapter VIII as an overarching synthesis. Furthermore, the future work that is proposed in the area is presented, as well as what is envisaged for the future of the development of SGs in a wider perspective of applications.

Below the reader may find a schematic diagram (Diagram 1.1) of the structure of the Thesis as per visual aid of the workflow. At the beginning of each chapter, there is also a diagram explaining the contents of the individual chapter in more detail.



Diagram 1.1 Thesis Structure

Summary

Knowledge transfer has been significantly transformed in recent years by the introduction of VGs in learning which in this particular embodiment are termed SGs. The benefits of their use are numerous and very advantageous to society. The recent advances of VR and gaming consoles realise the potentials of further development for SGs with novel approaches to support new tools and frameworks, or explore new areas of application. The potentials of the domains under investigation of this Thesis are in Defence and Health areas. A number of novel tools, applications and frameworks for these areas are presented in the following chapters. This Chapter also offers the Aims and Objectives of the Thesis as well as the structure of the chapters.

CHAPTER II: LITERATURE SURVEY ON SERIOUS GAMES

Abstract:

This chapter introduces a range of SG domains and presents the state-of-the-art in these areas. This chapter also offers acquaintance to a broader audience by offering essential grounding and background to a possibly foreign field. A historical background of SGs is followed by specific sections that focus on the domains related to this Thesis and present an extended review of the available literature for these domains. SGs have expanded to a very broad range of subjects and thus this section does not intend to serve as a comprehensive text, rather a focused introduction to selected domains relevant to the Thesis. For further readings the reader is advised to see [3] and [8]. The structure of the chapter is visualised in Diagram 2below:



Diagram 2 Chapter Structure

2.1 Historical origins

The definitions of the SG concept have been presented in the previous chapter. As a standardised terminology does not exist and there is still debate on the definitions among scholars of the subject, this Thesis will follow the oxymoron term Serious Games as defined by [1] and [3]. However, what can be realised is a historical retrospect of the concept. A question that rises when one should attempt to historically define the origins of SGs is: when was the first time that a VG was used for purposes other than merely entertainment [15]. Nevertheless, a research in the literature indicates that the term SG is used long before the evolution of VGs and according to Djaouti and Alvarez [15] with a meaning close to the current. The book of Clark Abt [16] entitled "Serious Games" provides a series of non-digital games to be used in education. Another example, again presented in the study of origins of SGs by Djaouti and Alvarez [15], is the book of Jansiewicz [17] where a non-digital game supports the education on the subject of politics. Nevertheless, the current

trends and penetration of digital technology in the market nowadays has restricted the definition of SGs to only VGs and digital forms. Therefore, a historical throwback of SGs, as defined currently, has to match the moment of time when the first digital game was used for purposes other than entertainment. Although that can be blurry and hard to define, the work of Djaouti and Alvarez [15] sheds light on the matter with an extensive presentation of early original VGs used for serious applications and for a variety of domains. The reader is encouraged to refer to that source for more information on the matter. Furthermore, Zyda's [1] massively cited publication is titled "From Visual Simulation, to Virtual Reality, to Games" which outlines the historical sequence of the field. Zyda, in the same publication, mentions the importance of VR into the games industry and the parallel relationship of the research fields that influence each other. The author also suggests that VR researchers should align with the research field of SGs as this interdisciplinary action will enable a wider audience. More than 40 years after the initial SGs, the number of commercial or research-related games developed is immense. An inventory of the genre would even be impossible if would one include all domains and application areas. This is another indication of the importance of the area and the popularity it gained in the recent years. As the range of domains and applications is so broad, this Thesis will only present the state-of-the-art and current trends in the domains closely related to its scope. Those, namely, are Education, Healthcare, Defence and Ecology-Sustainability. Thus, following is a presentation of the aforementioned domains with a series of examples that illustrate the conceptual embodiments of SGs for each of those areas.

2.2 Overview of the application domains

SGs have been known for being very engaging and fun in the process of learning and knowledge transfer. Therefore, SGs render as a very useful tool to be used in any educational environments and audiences, varying from a primary school classroom to a corporate induction training facility or military training environments. Nevertheless, their effectiveness is not established, as according to a series of scholars, there is lack of empirical evidence and consistent research methods [18],[19], [20], [21]. There is a certain level of agreement in the literature that this is due to philosophical issues. For example, the

lack of consensus on terminology and definitions and the overlap of domains or the lack of a taxonomical stratification of game attributes and their relation with the outcomes of SGs are some of the reasons of inconsistency in the research area of SGs. Once the outcomes become measureable and quantified, one would be able to conduct research on the effectiveness of any type of SG (or gamified experiences in general, as a future step) and decide as to whether that game or learning experience provides a suitable environment for knowledge transfer and is worth migrating from conventional types of learning to this novel approach. However, the theoretical basis of SGs is still in nascent phase and research on each specific case-study of SGs is bringing the field one step forward as the results of each study expands the knowledge and increases the span of available literature. Although there is criticism on the extent of empirical evidence on the effectiveness of SGs, there is a range of studies that demonstrate positive results on the usefulness of SGs in education [11], [14], [22], [23], [24]. A very significant factor that affects the acceptance of SGs is the rapid development of VGs technologies and multimedia platforms available which have penetrated the market over the last 20-30 years. This factor is the reason behind the high computerisation of our world and the exposure of children and young adults to technology form a very young age. According to [12], the generations that were born in this digitised era can be referred to as the "Net Generation". However as explained above, this acceptance is only a hint of the success of SGs and does not provide an a priori proof of effectiveness. It clarifies, though, why the recent generations are very receptive to SGs as novel educational tools. Moreover, the growth of the multimedia industry and rapid development of novel software and hardware tools, enable the customisation of SGs to match each specific audience's requirements, specifications and outcomes. Innovative ways of HCI, realised by the use of game sensors such as the Nintendo Wii, Microsoft Kinect, Sony PlayStation Move or Google Glass and Oculus Rift, outline unprecedented game design guidelines [20]. These novel approaches to design of SGs and their characteristics [21], [25] will maximise the effectiveness and facilitate the optimisation of the desired outcomes.

As mentioned above, this section presents a series of successful examples and state-of-theart for the areas related to the scope of the Thesis. This presentation serves to provide a background presentation to a broader audience and moreover to proceed one step forward to identifying the overlaps of the area leading to an establishment of SGs as a scholarly field. The domains examined, as mentioned before, are the ones closely related to the theme of the Thesis.

2.2.1 SGs as an instructional paradigm in education

SGs suggest an alternative, fun and engaging means of learning and transference. Therefore, their application in any kind of instructional education is apparent. This section presents the application of SGs in the instructional paradigm with a series of games for a variety of scholar subjects. These subjects can range from school modules as Maths, Science, Language and History to undergraduate modules or even corporate and business training.

The development and proliferation of 3D multimedia and virtual-worlds platforms, such as Active Worlds, has served as an online repository of assets for SGs development in a number of case studies. Active Worlds is a free VR platform where users can design 3D worlds collaboratively and are available to any user of the platform. Active Worlds have helped to provide a subject-free development of immersive games for any domain. Dickey, in his research [26], presents two case studies using Active worlds for distant education of undergraduate modules. Apart from the use of online virtual worlds, there are a vast number of custom-made games (or custom expansion set for commercial games) for instructional learning with the leaders of the field being the educational games. In their recent review of trends in SGs for education Young et al. [20], present a subject-based meta-analysis of the educational SGs. The studies reviewed include maths games such as "The logical Journey of Zoombinis" by Harris et al. [27] where the authors concluded that children's team interactions are improved by goal-focused gameplay. Another example is "Dimenxian/Evolver" by Mayo [28], which compared the learning outcomes among students playing the game against conventional lectures and found that they increased by

15

7.2%. Along the same lines, the study of Kebritchi et al. [29], again presented in the review of Young et al. [20], found the maths game "DimensionM" to be beneficial in terms of performance as compared to conventional educational approaches. Yet another study that showed better performance among students playing the SG named "ASTRA EAGLE" showed improved performance of the gaming groups as compared to the non-gaming peers [30]. Another interesting study is the one by Annetta et al. [23], cited by both [19], [20] in their reviews, where the authors investigate the effectiveness of SGs on learning outcomes of genetics. The use of the "MEGA" set of games in this study showed increased engagement nevertheless it did not show significant improvement in performance. Therefore, the authors here also suggest a more thorough investigation of isolated aspects of SGs. A game-based learning framework is proposed by El Rhalibi et al. [31], [32] and in a paleontological background as an application paradigm. The authors propose a constructivist theory, model and application for the design and development of SGs which introduce dinosaurs' science in VR as an application. The learning research investigates how to respond to current challenges of providing engaging learning experiences utilising the technological advances of today.

A bigger-scale project, including 18 countries and 983 classrooms, is the "Quest of Atlantis" as part of the Atlantis Remix Project, which immerses children of school ages in educational tasks [33]. The extensive research on that project provides evidence of the usefulness of games in learning. However, the lack of consistent investigation of the matter on a bigger scale and not isolated in theme-specific studies renders the evidence insufficient for one to conclude on the effectiveness of SGs in instructional learning.

The conclusion to be drawn by the review in the literature regarding SGs and education is that although SGs show a tendency to be exceedingly engaging and fun, there is a shortage in empirical evidence on their effectiveness and connection between the SGs and performance. This fact can be extrapolated to instructional learning in general as the principles are similar. In the author's opinion, the lack of taxonomy of game attributes is more crucial to that end, as the quantification of outcomes cannot be realised unless the elements of the game are clearly defined. Another aspect towards a consistent taxonomy is the stratification of the types of games for education. In their work, Tang et al. [34], present a taxonomy of games for learning. They describe the digital games as an entity and differentiate them from games for learning in the aspects of rules, gameplay and culture. Furthermore, they present the existing terms describing games for learning and classify their relations and scopes in the following Venn diagram (Figure 2).



Figure 2 Venn diagram of types of Game-Based Learning [34]

2.2.2 SGs for Sustainability promotion / awareness

The ever-increasing need for power of today's energy-starving society is widely recognised nowadays. Nor is the need to seek for alternative and sustainable solutions for suppressing this "hunger". There is a widespread recognition of the necessity for an ecologically sustainable environment. Hence, it is an essential requirement to disseminate Sustainable Development (SD) to all aspects of society. This is becoming more and more crucial since the depletion of fossil energy deposits and the increasing need for renewable, "green" resources.

At this point, the advantages of SGs seem very fitting to the aforementioned scope and have the potential to become extremely beneficial towards that target. As presented in Chapter I, the benefits of SGs enumerate among others the increased engagement and fun elements as well as their immersive nature and cost and resources effectiveness. These elements enable SGs to play a significant role as a message spreading method to activate ecological awareness, or to address the development of sustainable practices and policy strategies through simulations.

The list of customised games for SD, by a quick research on the available literature, is extremely long which for many observers can be rather surprising. The popularity of such a genre of games lies beneath the grounding elements of SGs. Games for Good is a term that is frequently used to refer to SGs with such targets and SGs for SD could not be more fitting to such a description.

An online search for this genre results in a vast collection of games, either research related or of a commercial nature. The website *ecogamer.org*, lists a huge variety of green games on subjects ranging from agriculture to energy and conservation or natural disasters and pollution games [35]. Another online repository that enlists SGs for SD is the inventory of the Serious Games Association. The online directory stratifies the games according to each market, with Games for Good being the niche for SD games [36]. In regards to academic research on the subject, two recent surveys have been identified that classify SGs for SD. Katsaliaki and Mustafee [37] list 35 SGs for SD among which include: Stop Disasters!, Climate Challenge, Electrocity, EnergyVille, 3rd World Farmer and Catchment Detox. The games were classified according to the player's role and game objective as well as the type of game, the graphic technology utilised, availability and target age. Another methodological approach towards a taxonomy of SGs for SD is that of Liarakou and Sakka [38]. The authors here classified SGs for SD according to the learning and pedagogical aspects and usability levels. The authors identified 34 SGs (a small group of which also belonged to the collection of games found by Katsaliaki & Mustafee [37] and organised them according to a theme. A subset of 18 games appeared with climate change as their theme, whereas 10 of them were concerned with the urban environment. The remaining games were concerned with other themes such as the issues of the deprived world, resources and waste management, agriculture and natural disasters.

An example of the application of the recent trends of gamification is the MEECO project of Vara et al. [39] where a social networking platform is utilised towards an engaging service for SD. MEECO is a platform where people can interact and share their daily "green" actions engaging in a sort of challenge to become more sustainable. The more "green" actions one performs, including recycling and mobility, the more points he/she will gain in a pursuit of sustainable awareness and consciousness.

A further investigation of the available SGs for SD shows that apart from the games to raise awareness and disseminate SD, there are tools and simulation platforms to plan and assess the use of available green technologies. These customised SGs serve towards the exploitation of sustainable best practices and strategies as well as the familiarisation of the player with the new green technologies. The report of Wein and Labiosa [40] for the United States (U.S.) government presents two cases of simulations for SD, namely the "Delta Skelta" and the "Age of Ecology" SGs. Both cases utilise agent-based simulation methodologies to simulate environmental planning, natural hazards and socioeconomic and ecological interactions while addressing resource-management issues. Another example is the "3D Wild Land Fire Simulation", a fire propagation simulator developed for the U.S. Department of Agriculture [3]. The simulator is scenario-based and is used for training and prevention planning purposes. Finally, another case of a simulation game is the project entitled "Solar Soldier" which was conducted within the purposes of this Thesis. The simulation platform developed within the boundaries of this study is a military simulation for SD and aims to assess the incorporation of photovoltaic (PV) technology on the uniform and equipment of the modern infantry soldier through the use of scenario simulations. Further details on the study are presented in a following chapter (Chapter III) of this Thesis.

The aim of such simulations is to form a set of guidelines and provide further insights on the use, assessment, evaluation or effectiveness of green technologies and sustainable policies.

2.2.3 SGs in the Defence domain

The advantages of the utilisation of SGs for training and game-based learning could only be beneficial in the military context and as such the attraction of SGs in the Defence domain is rather prevalent. SGs have been widely used for different reasons in military education, varying from decision making and battle simulations to military history and business management. A group of advantages that facilitate and promote the popularity of SGs in the military domain include:

Cost effectiveness as regarded in consumables and operational manpower, optimised skill acquisition of personnel, replication of a wider range of scenarios in terms of operational environments, interactions and equipment, recording and assessment of the sessions into vignettes for further analysis and reference, high level of immersion and operational security that in some cases could be otherwise realised and the increased motivational and engaging aspects of SGs that results in more effective training. [41], [42], [43]. Prensky [44] also argues that there is a perfect match between game players and army recruits; another reason behind the popularity and success behind the military cohort of SGs. The same author redirects the reader to the webpage of U.S. Military games for over 50 examples of the games employed by U.S. Department of Defence [45]. Another group of games used again by the U.S. army can be found on the website of U.S. Army [46]. The games presented there, comprise a group of games with various outcomes, from target practicing to full combat simulations. Among them, the arguably most popular military SGs ever produced, America's Army (AA) [3]. AA is the official online game launched in 2002 by U.S. Army to deliver a training platform reflecting aspects as skills, technology and career development of a U.S. Army soldier but at the same time attract new recruits as it is publicly available. On the other side of the pond, the U.K. army is not left behind. According to [42], the UK Ministry of Defence has adopted the use of SGs as an important asset for training, following the 2008 report on the strategy for simulations. The modification of commercial games has been taken into account to adapt and customise scenarios on demand. An initially commercial, of-the-shelf (COTS) game, namely Harpoon 3, was developed by Australian Defence Department for use in naval operational and tactical training [3]. Examples of COTS products used for such purposes are the Virtual BattleSpace 2 (largely used by UK army according to [42]), and Call of Duty, Delta Force or expansion packs for Civilization III. It is, therefore, made clearly evident that the parameterisation of the scenarios and the customisation of the VR training environment, according to the individual case and the specific requirements, are of essential importance for the SGs in the military domain. This is facilitated either via the development of bespoke SGs to match the individual criteria and specifications or by customising COTS games as mentioned above. The potential disadvantages on the use of SGs in the military context lie beneath the high costs of the development phase and the restrictions to simulate physical, physiological or psychological effects (e.g. post-traumatic stress etc.) [42]. For combat training related applications, the potential use of game sensors as MS Kinect or Google Glass and Oculus Rift to facilitate a more immersive environment takes the military SGs to the next level. The Dismounted Soldier Training System (DSTS) of the U.S. Army is a great example of immersive VR platform for training. Head-mounted displays, wearable electronics and motion sensor systems, which are technologies directly aligned with the technology used in modern COTS game sensors, are used to transfer the user to a virtual environment where everything can happen according to scenario requirements and operational deployments [47]. The U.K. MoD, on a similar project, has launched a VR platform for the training needs of parachute troops. A fully immersive approach with stateof-the-art equipment, including mock parachute stand and head-mounted displays, is supposed to bridge the gap between early stage training and live jumps by providing a safer environment for the operating officers[48].

Apart from combat and operational training and as aforementioned, the military interests extend to other areas; one of which is SD. Hence the project Solar Soldier, presented in more detail in a flowing chapter (Chapter III), was conducted with the funding of the U.K. MoD to assess the issue of huge load and bulky equipment that the modern infantry soldier has to carry in operations, most of which consists of batteries. The use of renewable energy sources and SD is becoming a popular subject and of great interest in modern military.

2.2.4 SGs in the Healthcare domain

A very promising and significant domain for the utilisation and exploitation of SGs, commonly admitted, is that of Healthcare. There many potential applications in that domain, in a great diversity of application types and embodiments. A review in the available literature [8] reveals a massive number of research studies employing SGs in health-related applications and it is considered a very fast growing and significant area of research. According to Ben Sawyer, founder of the Serious Games Initiative, (see [8]), the domain of Healthcare will be the prevalent subject domain for SGs in the forthcoming years. In the case of this domain, the popularity and attractiveness to researchers and developers is due to multiple reasons. One of the reasons is that along with the exploitation of the recent advances in graphics and 3D multimedia, the field of health-related SGs offers a perfect opportunity to exploit the utilisation of the hardware-related advances such as the development of innovative game sensors and motion capture devices. As mentioned again before, among the champions of these developments are the Nintendo Wiimote¹, MS Kinect², Google Glass³, Oculus Rift⁴, Leap motion⁵ and STEM[©] system⁶ by Sixsense. The reverse engineering of a number of those sensors enabled the customisation and tailoring of games for individual requirements and specifications, thus aiming to the specific motor abilities of different patient cohorts.

The breadth of the range of categories that SGs are used in the Healthcare domain is another indication of their success within that field. According to the reviews of Ulicsak [42] and Susi et al. [8] the subject areas of SGs within healthcare can be summarised:

2.2.4.1 Exergames

The term exergames is a portmanteau of the words exercise and games. Hence it is obvious that their aim is to provide a physical exercising environment whilst keeping the player engaged and entertained. A series of COTS games met huge acceptance and success with

¹ http://www.nintendo.com/wiiu/accessories

² http://www.xbox.com/en-GB/Kinect

³ https://www.google.com/glass/start/

⁴ http://www.oculusvr.com/

⁵ https://www.leapmotion.com/

⁶ http://sixense.com/wireless

the most popular being the Wii Fit⁷ [49], [50] and Dance Dance Revolution⁸ [10]. The platform of NIKE+⁹ and the mobile applications that come with the motion sensor is a recent example of gamified exercising. The term exergame has been used in research studies regarding rehabilitation of motor degenerative diseases. Nevertheless, these types of application are included in the context of rehabilitation games in this Thesis. Exergames are defined as the games to promote the physical activities and healthy lifestyle and on the same time entertain and engage the player. [51]

2.2.4.2 Educational and training simulation and games

The training of personnel in the health care domain can be extremely cost-ineffective as well as hazardous in many cases. The training of new surgeons is a classic example of such case. The risk of training implications is very high and an immersive environment offers a great solution to that issue. A systematic review for this genre of SGs is the study of Graafland et al. [52] where the authors present in detail 30 games, 17 bespoke and 13 COTS that are supposed to deliver skills relevant to medical personnel. The subjects range from simulations of training in acute and critical training, to triage training, to training of operation steps in immersive learning environments.

Nevertheless, the educational SGs in Health domain are not restricted to skills training. There are a considerable number of games in the field that aim to inform the public on various related issues. Two examples are FatWorld (see [42]) and Hungry Red Planet (see [8]), which are games aiming to inform about obesity, malnutrition and healthy eating styles.

2.2.4.3 SGs used as tools for diagnosis, management, recovery and rehabilitation of cognitive and motor impairments

A broader section of Healthcare SGs is dedicated to diagnostic and management tools as well as training of motor and cognitive skills to overcome the impairments caused by different diseases and medical conditions. Examples of such diseases are, in terms of

⁷ http://wiifit.com/

⁸ http://www.ddrgame.com/

⁹ https://secure-nikeplus.nike.com/plus/

popularity, stroke, PD, post-traumatic stress disorder (PTSD), Alzheimer's disease etc. The SGs developed in this category can serve as a training platform for cognitive skills as memory enhancement or analytical skills development, or for motor skills with provision of biofeedback to the physician for a more sophisticated management of the therapy. There are exceedingly too many a game developed within this subsection, nevertheless surveys such as [53] provide a classification and taxonomy of rehabilitation games, and [54] where the authors present a methodological review of the bespoke rehabilitation platforms utilising the Nintendo Wiimote. The use of bespoke COTS game sensors and customised games is common feature in the majority of the studies and SGs presented in both the review studies. Two studies conducted within the scope of this Thesis have utilised bespoke (as PD) or neurophysiological (as stroke) conditions. The game sensor used is the Nintendo Wiimote and a series of mini-games are presented in the chapters to follow. Furthermore, a more elaborate review of the literature with regards to that specific subject is presented in section 2.3, Related Work.

Another popular application is the diagnosis of several conditions both cognitive and motor-related. Susi et al. [8] cite S.M.A.R.T. BrainGames for children with attention deficit hyperactivity disorder (ADHD) and a COTS game, Full Spectrum Warrior for PTSD.

2.3 Related Work

The aims of this Thesis, as described in Chapter I, consist of the use of bespoke COTS software and hardware game technologies for the development of SGs in the domain of Defence and Health. Therefore, this section serves as the elaborate presentation of the literature related to the aforementioned aim. This is accomplished in order to ensure that the work carried out for this Thesis is novel. The following subsections offer an in-depth outline of the work done by other scholars in each of the aspects related to the research conducted for the accomplishment of this Thesis. As the subjects examined by this Thesis are related to the research studies conducted in its boundaries, the related works are organised by subject (namely Defence simulations with sustainability and serious games
for motor rehabilitation) and a brief presentation of each study is offered before each review of available literature. More details on the studies and the challenges they are trying to administer are offered in Chapters III and IV.

Defence Simulations with Sustainability: Virtual Reality Simulations and Guidelines for the Integration of Photovoltaic Technology on the Modern Infantry Soldier

The type of warfare as well as the humanitarian campaigns we are engaged in nowadays uses foot soldiers a lot more, resulting in soldiers having to carry equipment that often reach and sometimes exceed fifty kilos with several of these including bulky and heavy batteries to power several of the electronic equipment that soldiers use. The challenge here is to reduce this physical burden on the modern infantry soldier moving towards a battery-free solution. Thus, the power generation assessment is of vital importance and as the physical environment is not available for evaluation of the technology and its integration on the uniform and equipment of the soldier, there is the need of using virtual environments to tackle the aforementioned challenge.

2.3.1 Simulation of product-integrated photovoltaics (PIPV) for Defence applications

The use of PV technology for the harvesting of renewable energy is a reality and is widely employed today. However, this is mainly focused towards house and industry energy harvesting. Recent development in thin and flexible materials mean that photovoltaic technology can be integrated into wearable computing and expanded to other commercial as well as Defence applications. Following recent advances in the field of thin and flexible materials, the use of PIPV for light harvesting and electric power generation has received increased attention today. Nevertheless, one of the key issues for the adoption of such technology is the correct placement (this depends on factors such as location, weather, etc.) of PVs on buildings, and other infrastructure as well as humans in order to maximise energy generation. VR simulations are a very good tool to fill the gap and provide with a solution by providing guidelines on the assessment of the incorporation of PV technology. As the area of interest is Defence applications, other available simulation platforms in that domain can provide a valuable insight to the matter. Moreover, simulations of light harvesting as well as studies on the incorporation of PV technologies on commercial products shed light and provide background research on the same issue.

2.3.1.1 VR in Defence applications

VR can be utilised for military applications to perform a wide range of simulations. These range from cognitive and behaviour simulations in battle to ergonomic simulations, all serving the improvement of the welfare of the modern soldier. These simulations have to be conducted in a virtual framework often consisting of assets that offer 3D graphical representations of terrains, human avatars and objects, as well as weather and daylightaugmented systems. All these elements create a Virtual World on a computer-based simulation environment. This is of significant interest and importance to research, as it offers a very useful alternative reality, especially for situations such as this where actual experiments are not feasible or dangerous to conduct in real life [55], [56]. More precisely, Chryssolouris et al.[57] have conducted research in the area of human ergonomics in an assembly line and Reece [58] has studied the movement behaviour of soldier agents on a virtual battlefield. Furthermore, the Santos [59], [60] project offers a virtual platform for human ergonomics in military environments and Shiau and Liang [61] present a real-time network VR military simulation system comprising of weather, physics and network communications. A similar study of Yin et al. [62] investigated the use of VR in unmanned aerial vehicle electronic warfare including functions of data link simulations, environment electromagnetic simulations and simulations of tactical exercises. Blount et al. [63] have introduced the aspect of physical fitness into simulations for infantry soldiers and others, such as Cioppa et al. [64] and Bitinas et al. [65] have worked with agent-based simulations and their military applications, focusing on human factors in military combat and noncombat situations, respectively. Apart from these aforementioned articles, there is a recently published three-volume edition containing extensive literature on VR and applications: The PSI handbook of virtual environment training and education: developments for the military and beyond [66]. The second volume of this text contains subjects such as 'Mixed and augmented reality for training', 'Evaluating virtual environment component technologies' and 'Enhancing virtual environments to support training'. The aforementioned literature focuses mainly on simulating human factors and ergonomics, either in the production line or in military environments. However, the applications of VR human-centred simulations are not restricted to ergonomics. The aspect of Human-centred Design (HCD) that this article examines is the integration of renewable energy devices on the human vesture, and in particular the integration of PV technology on the uniform or equipment of the modern infantry soldier in terms of light capture efficiency.

2.3.1.2 Simulation of solar light harvesting

Currently, the main focus of PV technology and its corresponding simulations has been on building and infrastructure applications. The recent developments in the area of PV devices, as described in the reviews of Parida et al. [67], and Chaar et al. [68] along with the introduction of thin films and flexible materials for light absorption [69], have attracted the focus of harvesting renewable energy to human-centred applications as well. The study of the performance of PIPV of Reich et al. [70] is twofold: firstly, to investigate the performance and electrical characteristics of the PV device itself; secondly, to study the effectiveness of light harvesting, which is also the main focus and aim of our work. The effectiveness of light harvesting depends on the interaction of the device with the environment, as well as on the type of integration of the PV technology on the product (e.g. attached on clothing, embroidered or woven into the fabric). The environmental conditions would require the modelling of daylight and shading in a 3D authoring and simulation tool, whilst the integration guidelines would require simulated scenarios and results that would infer the most effective method of integration.

i. Daylight and shading modelling.

With regards to daylight modelling, there have been numerous studies on methods to maximise solar system outputs as per the review of Mousazadeh et al. [71]. Moreover, there are studies on the shading effect of the environment, which investigate the effects of random shading on PV energy production [72]–[75]. Apart from research studies, there have been major developments in the corresponding software industry, with very intelligent and complex packages developed for daylight simulations, including 3D Studio Max

Design (3DSMD) by Autodesk, which is the software, utilised in this study. 3DSMD was chosen mainly because it comprises a toolset for animation and because it includes the feature of light analysis of a 3D scene, which is essential for a HCD project such as this. 3DSMD also offers extension capabilities through its embedded programming language, Maxscript. It can thus be used to semi-automate the procedures as described in the work of Paraskevopoulos and Tsekleves [76]. The results of the light analysis of 3DSMD have been validated by Reinhart et al. [77] and Paraskevopoulos and Tsekleves [76] and the software has been used in a number of other studies regarding light harvesting for PV technology [70], [78]–[81].

The work of Reich et al. [70], [81], [82] describes a method to simulate energy yield of PV cells integrated on consumer products, as for example a solar-powered wireless mouse, in indoor scenarios. They used a customised model of lighting system with 365 light sources so they could manipulate the solar position and have a representative 365¹⁰ sky zones. The 3D modelled environment and raytracing techniques were employed to export the PV energy yield.

Reinders et al. [78]–[80] have also utilised 3DSMD to develop a tool named VR4PV that simulated irradiance for PIPV in virtual environments. This tool enables the assessment of power harvest for PIPVs in indoor and outdoor applications and for complex and tilted surfaces and taking into account shadows.

Finally, the work of Muller et al. [83] is another study that examines indoor irradiance for PV technologies through simulations. The software packages utilised in this case are DAYSIM and Radiance for daylight and artificial light measurements correspondingly. The simulations were contrasted with actual irradiance measurements and then the both software were validated. The use of such kind of software that is not CAD derives a

¹⁰ Representing the solar days

restriction of the complexity of the 3D model and thus is not recommend for complex objects and tilted and curved surfaces.

The above studies have attempted to face the challenge of predicting the power yield of PIPV or building integrated PV (BIPV) technologies for indoor and outdoor applications. Nevertheless, all of them have focused on simulations where the PV device was in a static position and none of these has used simulation to analyse the effects of movement on PIPV light harvesting. Furthermore, no previous work has offered any conclusions or guidelines on the design aspects of wearable PV devices in terms of the light-capturing efficiency of mobile agents.

ii. Integration of PV technology on commercial products.

Although the integration of PV technology on commercial products is not a new idea, the emergence of flexible and thin-film materials has extended the possibilities of integration into more products with a smaller scale factor, which can be portable. However, until recently and as stated by Mestre and Diehl [84], there have been no guidelines for the integration of PV technology on products in the context of either human comfort or efficiency of energy harvesting. The authors of this study researched on a series of sustainable technologies, including PV, and their integration into consumer products. Although they offer a set of design guidelines, their conclusion is that there is a need to develop practical knowledge on how to design sustainable products. The integration of PV on garment and clothing goods has been dealt with in the work of Schuber et al. [85]. The authors of this study examine the integration of PV technology in terms of deposition, energy forecast and power management. The energy yield is extensively investigated with experiments and the data obtained is analysed to derive the extent of areas of PV required to supply a range of electronic devices with sufficient energy. This cohort of experiments is not applicable to real-life scenarios and thus is mostly restricted to indoor applications and small devices. With regards to the design aspects of the integration of PV technology on clothing, Schubert and Werner [86] have also presented an overview of flexible solar cell technologies applied to wearable renewable sources. This, however, focuses only on

the material aspect of PV technology. A report authored by Ashok Kumar [87] presents his work on Powered Clothing and wearable electronics. This study offers a section on the intricacies of power generation and supply and refers to it as the most significant challenge of such technologies. It is proposed that various renewable sources are combined to emerge a novel approach of power generation with a seamless sourcing. The author also suggests this approach as a recommended solution for the bulk and heavy equipment that the soldiers have to carry in military operations and usually mostly consists of batteries. The use of flexible solar cells in military applications is also suggested with the use of camouflage for further safety.

Strategies of incorporation of flexible PV have been the subject of the work of Krebs et al. [88]. The authors point out that the selection and integration of the PV technology is of paramount importance and directly affects the effectiveness of the energy solution. The authors suggest and investigate two possible integration techniques, one as an add-on substrate and one as textile-level integration. The evaluation of the techniques is performed under the standard testing conditions with simulated sunlight. The results of these experiments were mainly focused on the mechanical properties of the materials. Along the same lines, the research of Singh [89] examines the flexible PV textile solutions for smart applications. This book chapter presents the state-of-the-art in flexible PV textiles again in a mechanical perspective, focusing on the materials and processes of incorporation. An overview of the emerging technologies of solar powered electronics is the review article of Jia et al. [90] summarises the advances of the field and is offered as a reference to all practitioners and researchers of the field.

In their research study, Schubert and Werner [86] reference Gemmer, who has performed experimental investigation on light harvesting under different daylight scenarios and has calculated energy yield for various user profiles, for example a 'regular clerk', an 'outdoor construction worker' and a 'night shift nurse'. In the system presented in a following chapter (Chapter III), it is proposed that these profiles can be very easily modelled (3D avatars and motion capture) and simulated (light analysis tool, 3DSMD) for all various

30

light conditions (daylight system, 3DSMD) and encompassing environments (3D terrain models). The outcomes of such simulations infer the design guidelines of the most efficient manner of integration of PV on clothing in terms of light harvest.

The work of Reinders [91] examines in depth the options for PV systems and portable devices and presents their advantages and drawbacks. Among the drawbacks, one indicates the lack of PV technology penetration in our society and market. This is mainly due to limited knowledge of this technology by product designers and manufacturers, restricting in turn the extension of applications for this technology.

The work, conducted in the Solar Soldier project and presented in this Thesis, aims to fill in this gap by deploying design guidelines and a simulation platform to ascertain the integration of PV technology on military garments or equipment initially and commercial products in the future. As already mentioned in the introduction of this Thesis, the use of VR simulations is a prerequisite for military applications, mainly due to the hostile and extremely hazardous environment.

Serious Games for home-based rehabilitation of motor control

The rehabilitation requirements of various motor impairments, due to neurophysiological and neurodegenerative diseases, are complex and the prescribed exercises depend on the severity of the condition and the nature of the disease. The length of the rehabilitation regime is such that the patient has to endeavour to perform the exercise program long after the hospital discharge or even, in some cases, follow it life-long duration. A serious issue that arises in such long-haul rehabilitation interventions is the issue of abandonment. People lose their engagement with the regime and their motivation to perform the prescribed physiotherapy. The elements of fun and engagement, that SGs offer, are elements that realised them as a potential solution for home-based rehabilitation tools. The issues of congested clinics and budget suppressions is another reason towards the homebased SG solutions for motor rehabilitation of diseases such as stroke and Parkinson's which are the most prevalent motor degenerative diseases among the elderly. This section will focus on and present the available literature on this specific area. There are two subsections; one for neurophysiological diseases such as stroke, and one for neurodegenerative diseases such as PD. Research studies on the use of game sensors and bespoke games for rehabilitation of such motor degenerative diseases are presented and discussed in each section.

2.3.2 Customised Serious Games and game sensors for stroke rehabilitation

In recent years a vast number of studies have examined the use of VR in motor rehabilitation for stroke. Only within the last few years, there have been a number of research reviews that enumerate the wide range of studies dealing with the emerging treatment approach of SGs for motor rehabilitation specifically on stroke disease [54], [92]–[97]. One can realise that the volume of research conducted in this subject is extensive and thus, SGs appear to be a promising intervention alternative which increases the motivation and engagement of the patient.

2.3.2.1 Reviews on VR for stroke rehabilitation

An extensive, systematic Cochrane review published by Laver et al. [92], gathered and, after the exclusion procedures, meta-analysed a total of 19 studies of VR stroke rehabilitation interventions summing 565 participants in 11 countries and between 2004 and 2010. The significance of reviewing the available literature lies beneath the conclusion on whether the use of such kind of technologies and interventions are actually effective, to evaluate that effectiveness and further develop design guidelines for future practitioners and game designers. Laver et al. [92] compare VR interventions to conventional physiotherapy or no therapy at all. They primarily focus on upper limb, gait and balance and global motor rehabilitation. With regards to upper limb function, which is also the main focus for the study conducted for this Thesis, they examine the effects on outcomes for arm function and activity. They reference 8 studies that involved upper limb training and of the total studies that were examined, two used the COTS PlayStation EyeToy, one the Nintendo Wii and three the GestureTek IREX which is a COTS interactive rehabilitation and exercise system. The rest of the studies examined in this review used bespoke VR platforms. The authors conclude that there is a gap in terms of empirical evidence of the effectiveness of VR interventions and more randomised control trials (RCT) should be conducted. It is also recommended that researchers and designers should adopt the pilot testing as their methodology to assess usability and validity of the end-product. They also note that most of the studies so far, although a lot in number, are mostly proof-of-concepts and feasibility studies. Nevertheless, they state that the field is developing and very promising.

A review of COTS available gaming systems for the rehabilitation of the upper limbs in stroke was recently published by Thompson et al. [95]. The authors state that the empirical evidence for game-based rehabilitation alternatives is growing fast and there is a demand for constant research synthesis and meta-analysis in order to conclude on the effectiveness of such interventions and consequently move towards the commercialisation of such a genre of SGs. The inclusion criteria of the aforementioned review derived to a sum of 19 studies employing 215 stroke patients. In yet one more review, the authors state that there is lack of an evidence base on the effectiveness of game-based interventions. Finally it is suggested that researchers and designers should account for the complexities of stroke rehabilitation by incorporating qualitative data in the process of design and implementation.

The state-of-the-art in the field of VR and motor rehabilitation post-stroke is the subject of yet another and very recent review by Fluet and Deutsch [97]. The review includes studies that compare the VR intervention with the conventional physiotherapy practices as well as studies comparing VR presentation and for upper limb and gait rehabilitation. They only included studies since 2010 and the results enumerated 50 studies from which only 10 fulfilled the inclusion criteria. That fact shows that the amount of research in this subject is vast, yet not empirically supported. The conclusion of the authors is that the studies of this field should focus on investigating the efficiency of VR intervention in comparison to conventional rehabilitation practices as well as the comparison of different approaches in VR interventions between each other. This research direction will lead to the refinement of the technologies and their application to VR rehabilitation interventions for stroke recovery.

33

Finally, Virtual Environments and COTS games in stroke rehabilitation, is the review subject of Lohse et al. [94]. The authors of this review study try to review and meta-analyse studies of the efficiency of bespoke VR interventions as well as COTS games for post-stroke motor rehabilitation. The assessment of available literature resulted to a total of 24 studies matching their inclusion criteria. This study is the first to compare the outcomes of the customised interventions to those of COTS games applied to stroke motor rehabilitation. The results of the meta-analysis showed no significant difference between the bespoke interventions and COTS games, however the authors mention that the studies employing COTS games were too few and of very small scale to draw a safe conclusion. The analysis of this review also concludes that the efficiency of VR interventions is strong for customised platforms and promising for the COTS games. Finally, it is pointed out that VR interventions showed an initial benefit compared to conventional therapy regimes.

The review article of Pietrzak et al. [98], questions if COTS games is the future for upper limb stroke rehabilitation. They include a total of 13 studies where the use of COTS game consoles as the Nintendo Wii, PlayStation EyeToy and CyWee were used for post-stroke rehabilitation. They found the Wii to be the most prominent console used and the one to appear the most beneficial for the patients. The authors conclude that the use of COTS games for upper-limb rehabilitation, post-stroke, is again lacking in evidence of effectiveness. However the use of COTS game sensors is encouraged as it is a promising alternative to conventional physiotherapy for stroke rehabilitation.

Therefore, it is made evident, through the aforementioned reviews of the available literature, that VR technologies applied to the rehabilitation of diseases such as stroke is beneficial and should be further examined as empirical evidence is lacking in the field. Other work that needs to be performed in the field is, according to Rahman and Shaheen [99], is the identification of the patients that benefit from each type of VR intervention, or in other words the personalisation of the VR regime to the individual needs of the each patient and rehabilitation prescribed. Hence, the use of customised interventions is the mostly appropriate solution towards that end. This fact is also supported by the reviews of Ortiz-Catalan et al. [100] and Bleakley et al. [101]. In these reviews of VR technologies

34

for motor rehabilitation, the authors mention that customised tasks and personalised experience with adjustable levels of difficulty have been described as advantageous features by the majority of reviewed studies as they optimise the motivation and engagement of the patient.

2.3.2.2 Customised Stroke Rehabilitation interventions

As mentioned in the previous section, a high number of studies have investigated the use of COTS game consoles in the context of motor rehabilitation for stroke disease. The results have shown positive indications in terms of motivation and engagement but it is not clear and there is no evidence on their efficiency. Furthermore, the requirement of adaptation and personalisation of the game to match the specifications and requirements of the cohort of players with motor impairments is suggested by the majority of the aforementioned studies and reviews. The predominant game console that has been utilised either as a COTS machine or bespoke system is the Nintendo Wii and paraphernalia as the Wiimote or the Balance board. A number or recent literary review surveys have also investigated the experimental use of the Nintendo Wii in different clinical settings and collected patient and therapist responses on its use [102]–[104]. These highlight the importance of promoting better accessibility and more widespread use of affordable home-based systems, which on one hand motivate clinical users and on the other hand provide the required motion-sensing fidelity needed to offer VR based rehabilitation. They also indicate a number of issues with the commercial Nintendo Wii system as well as current Wii-based VR interventions. According to Lange et al. [104] accurate tracking of the patient is one of the primary challenges for VR based systems and according to Reinkensmeyer and Boninger [102] issues of remote progress assessment and remote interaction between therapist and patient have to be resolved. Also the amount of information, visual/audio feedback, instructions as well as length and speed of the exercises are often inappropriate for the stroke rehabilitation population [103]–[105]. It was also noted that several patients had difficulty holding the Wii remote [104] due to poor distal limb function [106]. Lewis and Rosie [107] highlight the importance of user control and therapist assistance and the need for supporting social interaction through the VR intervention.

The customised Wii-based VR interventions and SGs for post-stroke rehabilitation of upper limbs can be stratified into three categories depending on the technical methodology employed for customising the Wii remote technology, namely acceleration data from the Wiimote only, Wiimote and LEDs as an Infra-Red camera and a hybrid of the two. The vast majority of customised Wii remote stroke interventions employ one or two Wiimotes with a reverse engineered Application Programming Interface (API) to capture the patient motion by reading the Wiimote accelerometer data. A number of algorithms have been suggested and implemented by various research groups to filter these data and map them to motion to drive either a set of exercises or custom-built games [108]–[110]. Since all of these approaches obtain the position of the Wiimote in space using the acceleration data (that is the change in the linear acceleration as the patient moves the Wii remote in space) they suffer from a Degree of Freedom (DoF) limitation. More precisely these solutions offer accuracy only in 2-DoF as the acceleration data can only determine the pitch and roll movement. Such systems are therefore more appropriate for gesture-based interventions rather than one-to-one mapping (kinematic animation) of movement onto the VR environment.

Another very common approach for the Wiimote customisation is to use a pair of Wiimotes with the light-emitting diode (LED) sensor bar or custom LEDs as an Infra-Red (IR) camera [111]–[115] to build a low-cost motion capture system. Reflective markers or LEDs are usually placed on patient's arm or hand and as the exercises are executed the range of motion is captured and mapped onto the system display. The limitation in this approach is that each Wiimote can detect up to four LEDs in space, thus restricting the range of movement (ROM) and set of exercises that the patient can execute. Also care must be placed on the angle of the Wiimotes to reduce occlusion of LEDs as the patient moves his/her arm. This approach requires the patient to be seated and in front and close to the computer monitor in order to capture movement. Also the use of LEDs often requires extra power supply modules to be attached on patients.

Wilson [116] and Martín-Moreno have proposed [117] a hybrid solution that incorporates the two aforementioned approaches in an effort to increase accuracy and the number of DoFs. However this approach also suffers from a limited field of view of employing the Wiimote as an infrared camera making it unsuitable for tracking larger motions often required in therapy [110].

Williamson et al [105] have proposed a Wii-based system that merges acceleration with gyroscope data and the Wiimote's IR. This system is aimed at motion recognition for sporttype of games (American football in this case) and does not need kinematic analysis and mapping of user motion on a 3D avatar as for stroke rehabilitation. The acceleration and gyroscope data fusion algorithm developed requires the use of the Wiimote's IR to compensate for the gyroscope drift and movement corrections. Their results indicate improved motion recognition when compared to acceleration data alone but loss of accuracy when IR is out of sight.

2.3.3 Customised Serious Games and game sensors for PD

The utilisation of SGs for motor rehabilitation is widely investigated and a vast number of research studies examine the use of VGs and VR for the rehabilitation of motor impaired individuals suffering from a broad range of diseases, including PD. According to the review of Holden [118], VR training appears to have a certain number of advantages in contrast to conventional physical rehabilitation. The same review summarises the studies conducted in the use of VR for motor rehabilitation in the following points. Firstly, it is established by the reviewed literature, that people with disabilities in fact appear able to acquire motor skills within virtual environments. Moving on, the motor tasks performed in the virtual world can be transferred to the real world in most cases. VR has been found beneficial with regards to motor skills when compared to real-world physiotherapy. Furthermore, there was no reporting of negative symptoms (i.e. cyber-sickness) on the patients in any of the reviewed studies. The potentials of VR have been examined by Cherniack [119] in a rehabilitation context for the elderly with cognitive disorders such as PD, stroke or Alzheimer's disease. The same study enumerates the application types of VR and reviews

the available literature on each of them to conclude that VR-based rehabilitation frameworks for such diseases can offer a more comprehensive, flexible and safer environment that will increase the engagement of the subjects with the interventions.

More specifically, in the context of PD, the use of customised SGs and commercial offthe-shelf (COTS) game sensors is also examined by a number of scholars trying to identify the effectiveness of this type of interventions. Sensors as the Nintendo Wiimote, Microsoft's Kinect, Sony PlayStation Eye and even simple webcams are among the COTS mostly employed in the aforementioned studies. Among those sensors and game platforms, the Nintendo Wii is the one that has been predominately used, due to its Wii Fit title popularity in rehabilitation studies. The study of Sugumaran & Prakash [120] suggests that there is a straightforward connection between simply playing games and dopamine release, as the latter occurs with every kind of physical exercise. Another research report [121] indicates that apart from dopamine release, physical activity is shown to improve the wellbeing of patients. Some other advantages of using Wii as a rehabilitation platform are due to the console's portability, as the therapy sessions can occur in a much safer and comfortable environment (e.g. patient's home). Furthermore, as stated in the works of Jung et al. [122] and Dhillon et al. [123], the use of Wii promotes the motivation, especially of the elderly, as well as participation in physical activities and social interaction.

A more advanced clinical trial as performed by Pompeu et al. [50] has shown that PD patients using Wii Fit for certain periods of time improved significantly in Unified Parkinson's Disease Rating Scale (UPDRS) -section 2, which refers to the Daily Living Activities outcome measure. In a single-case research study by Zettergren et al.[124] the patient showed improvement in gait speed, balance functional mobility as well as cognitive aspects, such as depression (in all measured pre-test and post-test using clinical assessment scales for each aspect).

2.3.3.1 Review of Customised Interventions for PD rehabilitation

Commercially available games and peripherals like the Balance board, the Wiimotes and MS Kinect have been widely used for the assessment and rehabilitation of motor dysfunctions for PD patients and have been found to improve the patients' well-being as

seen in some of the aforementioned studies. Nevertheless, the use of off-the-shelve games has been shown to be limited or completely inapplicable when it comes to patients with severe conditions [125]. Specifically, the commercial games may be too difficult for the patient to play with and thus induce frustration and reduce motivation. Furthermore the therapist has no control on the ROM that is appropriate for each PD patient, limiting the utility and usefulness of such interventions for the purposes of a rehabilitation session. From a therapist's perspective [126], off-the-shelve games lack the required feedback in order to track performance or customise the set of exercises or level of difficulty required to match the individual patient's requirements. A specific example of such a case is presented in the article of Vuong et al. [125] for PD. The specific symptoms of the disease are taken into account as well as the most common and widely accepted protocols of rehabilitation for PD. The multimedia system, developed in the aforementioned research, consisted of an infrared web-camera as motion sensor and instructed physiotherapy exercises assisted by audio and visual cues.

Customised games for PD rehabilitation were designed in the study of Assad et al. [127], where the authors present a number of design principles for PDR games. Similarly to Vuong et al. [125], they propose, the combination of "Training Big" with visual and auditory cues, which are correspondingly the most effective physiotherapy and perception support mechanisms for PD. The methodology presented in Assad et al. [127] is using a marker-based webcam motion capture approach utilising Sony PlayStation Eye camera and a set of mini-games customised according to the PD rehabilitation protocol.

Another research study utilising web-cameras as motion tracking system is the research of Cammuri et al. [128]. The system presented in this paper suggests the use of video-based analysis for real-time performance of physical exercises. The platform used is an opensource hardware and software platform, namely Eye-Web, and the authors propose and develop a multimedia system for PD rehabilitation using this platform and custom games. In the paper of Lilla et al. [129], the authors indicate the effect of synchronised music beats to the patients' movement in the attempt to train motor skills via customised computer games. The sensor utilised in this case is MS Kinect. It was found that there was a tendency to achieve higher scores when the musical cues were on.

The use of Kinect as a support therapy for PD is also examined in the work of Sanmartin et al. [130]. This study focuses on the arm reaching movement and the use of motion sensors to record for further analysis. It is a marker-based solution as the authors propose the use of coloured gloves and colour based algorithms as a tracking system.

Another study that utilises the Kinect to monitor and analyse the physiotherapy exercises prescribed to a PD patient is the study of Martins et al. [131]. This research presents a tool for the analysis of a range of exercises and movements by providing biofeedback in the essence of range of motion of the specific limbs and speed, acceleration of movement. As this tool intends to validate the performance of the exercises, it can potentially be used by both the patient and the clinician.

A bigger scale, real-time rehabilitation system is proposed by the authors in the work of Yu et al. [132]. This approach involves multiple infrared MoCap cameras, a wall projector and visual and audio controllers for the audio-visual cues. The design of such system follows the principles set by prior studies as: usability in clinical area, "training BIG" protocol, accuracy of motion capturing and measurements, repetitive and variable nature of exercises, engaging and motivational and performance adaptable. This approach is a robust and precise solution to PD rehabilitation. It nevertheless suffers from portability issues, as there is the requirement of a dedicated space for the system to be installed restricting the options only to clinical-based and not home-based solutions.

Finally, Su et al.[133] propose another solution to increase the motion capacities of the upper limb of PD patients. Their study suggests the use of VR projection-based system coupled with an affordable and precise motion capture system (Patriot by Polhemus Inc.) utilising electromagnetic field motion capturing and a Wii Balance board. The use of polarised glasses supported immersion to improve the VR experience and a customised game was employed (catching virtual targets). The effects of speed of VR ball catching were examined and found to be related to the speed performance of the upper limbs of the

patient. The manipulation of the speed by the therapist is suggested to be beneficial for the PD patients.

2.3.3.2 Review of Design Guidelines

When designing games for rehabilitation, it is very important to keep in mind the variations in patient ability, both cognitive and motor. Not all exergames or cognitive tasks training platforms are suitable for the scope of PD rehabilitation which is generally the case for all cohorts of patients and all sorts of diseases and disabilities. The study of dos Santos Mendes et al. [49] confirmed the hypotheses that PD patients would show deficits in comparison to healthy elderly depending on the requirements of the game. Among the 10 WiiFit games tested for the purposes of this study, PD patients had normal learning and retention compared to the control group of the elderly in 7 of them but had lower performance on five. Furthermore and after some training, PD patients still could not improve in three of the games tested, whereas the control group showed improvement. These results indicate the significance of tailoring and customisation of the games and their demands to align with the requirements of the targeted group. Moreover, the lack of literature on evaluated guidelines for PD SGs further endorses the need for such guidelines to facilitate adaptation and personalisation. As stated in the paper of Göbel et al. [134], the main issue existing in SGs in Sports and Health is the lack of concepts for personalisation and the strong need of engagement with and sustainable use of SGs. The need of adaptation and calibration is also examined by Geurts et al. [135] where the authors presents four SGs tailored for patients with spasticity and motor degeneration. Their study derived to a series of results which are translated into game design requirements and guidelines. Apart from the variations between healthy and impaired players, the authors point out the huge disparity among the motor skills of the patients. Even with the same individual there can be variation of movement due to fatigue or relapses. Therefore, automatic calibration and dynamic adaptability within the game is an important aspect of the design. The same issue is raised in the study of Palacio et al. [136] where the authors present design guidelines for videogames to promote leisure activities for older adults. They point out the gap in the gaming industry that focuses on younger populations and the designs are tailored for a younger demographic, excluding older adults. The design guidelines that the same study suggests is the use of big letters and buttons to accommodate easier interaction with the elderly and the lack of skills evaluation to avoid the players' frustration in case he/she/they cannot achieve the objective of the game. The non-reward nature of gameplay and its influence has been studied in the work of Lima et al. [137]. The brain activation associated by non-reward games is hypothesised to examine the influence in motor ability and executive function in PD patients and showed positive outcomes. It is demonstrated that the winners acutely showed improved memory and motor skills. Another research article that investigates the development of design guidelines for SGs in rehabilitation is the research of Bouchard et al. [138]. The authors develop a set of guidelines for developing SGs for Alzheimer patients. The main four aspects of the design are the appropriate choice of challenges within the game, the design of suitable interactions for the cognitively impaired, adoptability and dynamic calibration of difficulty levels and effective design of visual and auditory cues for the aid of cognitive training. The same points, individually or in groups, are found as design guidelines in studies that focus on PD motor rehabilitation as well.

The key design elements for VR games for movement rehabilitation are summarised in the review of Lewis & Rosie [107]. The review outlines the studies that have developed SGs for rehabilitation in a user centric manner, in terms of ascertaining the user's perspectives and feedback into a central point. The key patterns found in the reviewed studies (mostly stroke and cerebral palsy interventions) as well as the authors' experience in rehabilitation lead to a set of key design suggestions and themes. The sense of control by the player, the feedback in terms of score and success experience and the element of adoptability to constantly challenge the player to reach their maximum capabilities are among the most dominant themes. The authors mention that although their review is restricted in stroke and cerebral palsy it is high likely that the principles will be fit to other cases (e.g. PD) and recommend the further investigation of user-requirements' capture to facilitate the maximum effectiveness and engagement with SGs for rehabilitation.

Summary

The application of SGs knows no restrictions in terms of relevant fields. The most popular fields of application are presented, as a general overview of the field for a broader audience, in the first part of this Chapter. Moving on, the literature available related to the work of this Thesis is presented and organised focusing on the specific subjects of this Thesis. Firstly, the related available work done on VR simulations for sustainability and PIPV is offered. The works on simulations of solar-light harvesting, and in particular daylight modeling and the integration of PV technology in a Defence background and on commercial products is investigated and analysed. Finally, the literature review of SGs and VR platforms for motor rehabilitation is presented with a focus on the customisation of interventions for stroke and PD utilising bespoke game sensors such as the Nintendo Wiimote.

CHAPTER III: SERIOUS GAMES AND SIMULATIONS FOR DEFENCE APPLICATIONS FOR ENERGY SUSTAINABILITY

Problem Description, Proposed Solution and Implementation

Abstract:

This chapter offers a more detailed insight on the subject of Serious Games and Simulations for Defence applications for Energy Sustainability by presenting the work conducted under the Solar Soldier project. The project examines how digital economy and in particular Virtual Reality Simulations accommodate the development of guidelines for the integration of photovoltaic technology on the modern infantry soldier. Following the related work of other researchers from various fields presented in a previous chapter, this chapter particularly emphasises on the methodology adopted for the design and development of a VR simulation platform to assess the aforementioned problem. The structure of the chapter is visualised in Diagram 3 below:



Diagram 3 Chapter Structure

3.1 Challenges and opportunities for sustainable practice in modern military

Despite modern advances in military technology, the infantry soldier continues to play a significant role in Defence. In the age of stealth jets, nuclear munitions and guided weapons, it is still the infantry soldier that examines and secures a location to ascertain whether the target area is cleared and the enemy is defeated. The modern infantry soldier utilises the electronic technology and resources available today, in order to penetrate into hostile and difficult terrain, where armoured vehicles cannot penetrate and overcome the enemy. The power requirements of such electronic technology, critically essential for the modern soldier, are much higher when compared to the power requirements of a civilian counterpart. Furthermore, the environment of operation is far more hostile and challenging than those of the civilian applications and the loss of power may endanger the infantry soldier's life. That is the main reason behind the massive overload of batteries constituting nearly 25% [139] of the overall equipment load (including lethal, survival and

communication). Owing to the aforementioned fact, there is an uncontested restriction of maneuverability and operational range, as well as a significant burden, both physical and cognitive. The recent advances in the field of sustainable energy, and particularly the innovative flexible and wearable photovoltaic (PV) technologies, could offer a potential solution to this issue, by removing, or reducing to a great extent, the use of batteries. The Solar Soldier project, which is partly funded by the Defence Science and Technology Laboratory (DSTL) of the MoD in the UK and the EPSRC, investigates this research challenge. The work conducted within the project is presented in this chapter, which focuses on how one could integrate the PV technology epitomising the Solar Soldier concept from a human interface and design perspective. The objectives of this challenge are the following:

- To assess the incorporation of the PV technology on the uniform and equipment of the infantry soldier;
- To accurately measure and evaluate its effectiveness (amount of light captured under various scenarios) taking into account usability (human comfort, intuitiveness).
- To assess the effectiveness of each area (amount of power generated under various scenarios) as well as to investigate the areas that yield the same power values all over their extent for further research on usability (human comfort, intuitiveness).

The review of the available related work, presented in Chapter II, has ascertained that the abovementioned aims are not fulfilled by any other research study up-to-date. Moreover, there is a need for a cost-effective, safe and realistic VR environment for PIPV simulations that will enable the assessment of that technology in order to determine the best practice of integration and thus tackle the problem of the bulk and heavy equipment that the modern infantry soldier has to carry. The use of VR simulations for PIPV on the infantry soldier and for a variety of military scenarios is a novel application of VR simulations and SGs proposed by this Thesis. When designing PIPVs, the challenges that arise are how and where to integrate and furthermore how effective the integration will be in terms of energy

harvest. The proposed platform of this Thesis utilises VR simulations to address the energy harvest of the PIPV integrated on the infantry soldier and simulates the power generation under a variety of operational movement styles (animated avatar) and military environments. This challenge that has not been investigated by any other research so far and this Thesis is the first to examine it to the best of the author's knowledge.

3.2 The Simulation Platform

The problem statement presented in section 3.1 requires the employment of a virtual framework that is able to conduct a number of experiments and collect measurements, which are impossible to collect due to the hazardous nature of the real environment. Thus, to enable those experiments a simulation platform is developed with a step-by-step wizard to assist the user to choose between the various options of equipment, type of landscape and daylight conditions. The methodology employed for the development of the simulation platform, fulfils the requirements set by the objectives is Modelling and Simulation (M&S) [140]. The application of M&S presented in this chapter is aimed at applying an existing feature of a 3D authoring commercial software, Autodesk's 3DSMD, by extending its capabilities and applying it to simulation of daylight for sustainable energy applications of military interest. The lighting analysis system of 3DSMD will be employed in a virtual military environment framework. Nevertheless, before the development of the simulation platform, the software plugin is validated against real sunlight measurements in order to facilitate the accuracy required for the purposes and conditions of the study. Only after the accuracy has provided proof of concept, was it possible to move on with the design of the light sensors. These are offered in the design assets palette of the software and are attached to specific areas of the soldier's uniform and equipment to assess the incorporation of PV technology. The Block Diagram of Figure 3 illustrates the overall adopted methodology of the Solar Soldier project.



Figure 3 Block diagram of the adopted methodology

This chapter, as depicted in the Chapter structure, covers the problem description, the proposed solution and the implementation of the simulation platform. Within the proposed solution, the experimental accuracy study is presented in section 3.2.1 below along with the results as a proof of concept and an experimental study of system accuracy. The evaluation of the simulation platform as well as the results and discussion are described and offered in a following chapter (Chapter IV).

3.2.1 Experimental study of system accuracy

The experimental study of system accuracy for the 3DSMD lighting analysis tool, especially in outdoor conditions, forms an essential and a significant task in the project, since the aim is to ascertain how close to the actual condition would be the proposed simulation platform. As already mentioned the software that fits the requirements and have the appropriate extension capabilities are Autodesk 3DS Max and Max Design tools, especially as the latter offers a built-in light analysis tool. The light analysis tool is based on the ExposureTM plug-in and it has been so far mainly employed for interior light analysis by architects and interior designers. There is also a validation study conducted by Reinhart and Breton [77] at the National Research Council of Canada (NRCC) but this also

mainly focuses on an interior scenario with a very brief and unsystematic validation in an external condition/scenario. Thus the utilisation of the tool for exterior use has not been widely used and there is a lack of thorough and systematic validation studies focusing on outdoor conditions.

3.2.1.1 Experiment Methodology

The experimental study of system accuracy for the 3DSMD lighting analysis tool could be satisfied only by providing two essential requirements. These are access to light intensity data measurements (measured in lux) of a particular place, with long time intervals defined (e.g. days or months) and the precise 3D CAD model of this place. Thankfully for the purposes of this study, Brunel University runs two projects unrelated to this that fulfill the aforementioned requirements. Light intensity data are collected on a daily basis since October 2006 by the weather station of the SunnyBoy project conducted by Chowdhury et al. [141]. The aim of this project is the data acquisition and performance analysis of a PV array installed within the Brunel campus as illustrated in Figure 4.



Figure 4 3D model of Brunel University with the weather station location

The next element towards this study is the 3D model of the campus. Accurate models of this are also available from a past project of Brunel University. Therefore, the basic toolset of the study is acquired and the next step is the modification of the simulation setup.

3.2.1.2 Simulation Setup

A number of scenarios have been set up for this study. These scenarios cover all the possible required daylight conditions, which are three. Namely a sunny, a partially cloudy and a cloudy day. For this purpose, three corresponding days to those conditions were chosen. These days were selected based on weather forecasts and observation and light intensity values collected with the proper equipment (lux meter) on site during those particular dates. The investigated scenarios are defined in Table 1.

Scenario	Daylight Condition	Date
SC1	Sunny day	03-June-2010
SC2	Partially cloudy day	25-October-2010
SC3	Cloudy day	17-November- 2010

Table 1 Lighting Simulation Setup Scenarios

3.2.1.3 Lighting Analysis Rendering Setup

The light analysis system requires a virtual sensor to be designed in the scene in order to yield results for this sensor. The weather station is located on the spot indicated by the red star in Figure 4. Therefore, a lighting analysis sensor (Light Meter) is designed on the exact same spot with the corresponding area that the measurements were conducted. The lighting analysis in 3DSMD uses mental-ray to simulate physical lighting, thus the accuracy of the results lie beneath the precision of the rendering settings. The lighting analysis assistant by ExposureTM plug-in provides a Lighting Analysis Render Preset of adjustments for the parameters of the simulation. It utilises the mental-ray raytracer with the method of backward raytracing and with Raytrace and Final Gather setting enabled as referred in the Functional overview of mental ray v1.5 by mental-images [142]. The only modification to that preset, required for reasons of presentation, is the scaling of the Analysis of Value Colour Coding. The minimum illuminance value in lux is 0 which corresponds to darkness and the maximum is set to 120,000 lux which is the typical maximum illuminance value for a clear sky day [143].

3.2.1.4 Lighting System Setup

The lighting system has to be setup too. Although 3DSMD as mentioned above utilises ExposureTM plug-in for lighting analysis, there have been some studies such as the study of Reich, [70] that adopt a different approach, scaling visible light to yield energy levels. In the case of this study, the lighting system of Exposure is used by employing the Perezall Weather sky model as mentioned in both the daylight simulation guidelines by Autodesk [144], [145]. The daylight system in the setup is the mr Sun and mr Sky and the longitude and latitude of the landscape is set to 51.53285° and -0.47283° respectively, which is the exact position of Brunel University.

3.2.1.5 Material Adjustments

Another simulation parameter that has to be adjusted, according to the Daylight Simulation guide provided by Autodesk [144], [145], are the materials of the 3D objects as these define the optical properties of the objects. The only applicable materials for lighting analysis by default are the mr Architectural Design materials and the "ProMaterials". As indicated in the study by Reinhart [77] these materials are complex in terms of parameters whilst accuracy lies in the detail. In an outdoor scenario like the one examined in this experiment, the light values are far higher than an indoor scenario and shading is less, consequently the accuracy of results is not highly material dependent. Thus, materials that match the optical properties and colour of the actual architectural structures are utilised. These include the effects of reflectance of solar rays on the surfaces as accurately as possible. Furthermore, in every scene object a map is assigned, which increases the precision of the optical properties in complex materials like the camouflage and vegetation. The map assigned along with the appropriate shader will provide the object with the optical properties required.

3.2.1.6 Accuracy Study Results

The resulting values of light intensity acquired by the individual simulations described in the sections above are plotted in graphs and contrasted with the actual values acquired from the weather station of SunnyBoy project. For the three different scenarios of the study there are 6 plots compared in pairs resulting to three graphs as illustrated in Figures 5, 6 and 7. The Y axis represents the illuminance values measured in lux. The higher the lux values are the higher are the light intensity values that reach the PV panel. The X axis represents the time of the specified day. One may notice that across the three figures (Figures 5, 6, 7)

the overall time duration varies by a few hours. This is due to the specific date and season that these measurements took place. For instance the measurement of Figure 5 was taken place during summer, whereas the one in Figure 6 is in winter, where there is light for less parts of the day. The values demonstrated in the figures are taken on a five minute interval. The changes seen in the lines (peaks and drops) simply illustrate the amount of light in lux that the measurement devices record and simulated in 3DSMD. Thus if the sun is on the middle of the sky with no clouds obstructing it there is a peak (high lux value). If on the other hand there is a cloud passing between the sun and the measurement unit the lux level drops.

The results clearly demonstrate that for any outdoor daylight condition scenario the lighting analysis tool of the 3D software yields values of light levels extremely close to the actual measurements. More precisely and for sunny days as illustrated in Figure 5, the values generated by 3DSMD and those of the actual measurements are very much the same. In the case of a partially cloudy day, as illustrated in Figure 6, 3DSMD's values are very close to the actual measurements for the majority of the day's duration. There is a very small discrepancy during ten to one o'clock between 3DSMD and the actual measurements, but this is in the range of 50lux at its peak and it is thus a negligible amount to take into consideration.

For applications that require long term analysis of light levels as for example a study of light levels for PIPV at a particular place and for a given long time interval such as a whole day, the approach adopted by this study derives to qualitative results. In other words, the requirements of light level analysis for PIPV



Figure 5 Scenario 1 – Sunny Day



Figure 6 Scenario 2 – Partially Cloudy Day



Figure 7 Scenario 3 - Cloudy Day

applications can be substituted by the virtual tool offered by 3DSMD and ExposureTM. However, the complexity of this tool added by the compound procedures of rendering and analysis are an obstacle to users or scientists who wish to employ a virtual tool in their projects. That is one of the essential aims of the project to which this study is part of. Utilising the design and development of a Graphical User Interface (GUI) we are trying to solve this aforementioned issue and semi-automating and simplifying the procedures of virtual lighting analysis for outdoor applications of PIPV. The GUI will offer the end user the capability to synthesise and modify in few simple steps their landscape and weather conditions and furthermore to analyse and infer safe, as proved above, results for further statistical analysis and planning of the applications.

3.3 Implementation of the light analysis platform

Moving on to the main purpose of this project, the simulation platform is assembled according to M&S methodology very similarly to the experimental accuracy study that has been described in the preceding section; however engaging a more complex set of models and daylight conditions. The software consists of a light analysis plugin, which is the tool extended to fulfill the requirements of the present study. The main purpose of the light analysis plugin of 3DSMD is for building light assessment and is mostly used by civil engineers. The aim of this study requires light analysis of non-static sensors attached on the soldier's uniform and equipment. Therefore, a script was developed using the software's API (Maxscript) that enables the light analysis of mobile sensors. Figure 8 is the diagram of the algorithm with Top-Down development approach depicting the various components:



Figure 8 Top-Down approach diagram of the development

Following is the assembly of the simulation platform. The platform consists of an animated human avatar, namely the UK infantry soldier, various terrain types and a virtual daylight system. Therefore, the initial stage of the methodology is to acquire all virtual assets required for the scope of the modeling and then the second stage includes the manipulation of these 3D assets together. These have to be manipulated (modeling, scaling, texturing, animating) in order to fulfil the requirements of the planned simulation scenarios. Then all the assets are merged together. The 3DSMD daylight system is set up and the light sensors are designed. This completes the virtual scenes ready for simulation. The simulations output raw data is in comma separated value (CSV) form, which then can be easily transformed to spreadsheets and imported to MatLab for further analysis and presentation.

3.3.1 Modeling

The final outcome of the modeling process is a virtual military framework that includes a range of virtual terrains, a human avatar and a ROMs (animated clips) combined together to form each unique military scenario as depicted in the figures below (Figures 9, 10, 11):



Figure 9 Urban area scene



Figure 10 Forest scene



Figure 11 Military camp scene

The modeling process involved several stages, as depicted in Figure 3. In the first stage, the environment models required for the various scenarios are developed or acquired through online 3D resources. After all assets have been selected, the separate models of the human avatar and terrain models are manipulated and merged into unique scenes along with the daylight system and the other assets (virtual sensors and animation clips) to form the unique scenarios. Similarly to the setup of the accuracy test experiment, the modeling of the simulation platform has to abide with certain technical specifications:

3.3.1.1 The Lighting system

The lighting system as mentioned in section 3.2.1.4 is the daylight system of 3DSMD. The daylight system allows the designer to select the date and time as well as the location on the globe through a selection of cities or latitude and longitude value pairs. Furthermore, it provides users with the ability to adjust the sky model by choosing one of the three following options: the Haze-driven model, the Perez "All weather model" and the Commission Internationale de l'Eclairage (CIE) model. The CIE sky supports only the extreme cases of overcast and clear sky [146] whilst the Perez "All weather model" [147] supports intermediate cases also. Thus, when the weather data are available, it is more appropriate to use the Perez "All weather model", whilst when relying on typical light intensity values CIE offers a more suitable solution. The sky model [144], [145] chosen for this case was the Perez "All weather model". This is provided online by Autodesk, and

allows its user to employ the irradiance data provided by Photovoltaic Geographical Information System (PVGIS)[148]. PVGIS is an online system developed by the Joint Research Centre for Energy and Renewable Energy Units. Among the data available online, there are specific areas that are of major military interest and thus very valuable for the purposes of this study. The data utilised include a set of average daily irradiance variations for the chosen location with time intervals as low as 15 minutes during a typical day of any given month. This approach covers a wider range of variable climates.

3.3.1.2 Rescale of the Analysis Value colour coding

The lighting analysis tool provides the user with an Image overlay to the light meters with a coloured coded scale. The colour coding uses blue for zero and red for the maximum value set by the designer. Once more since it is an outdoor illuminance analysis, the typical outdoor maximum illuminance value for a clear sky is used, which corresponds to 120,000 lux [143].

3.3.1.3 The Lighting analysis Render settings

The render settings of the lighting analysis are as complex and varying as most rendering setting for any other 3D application. The setup of the rendering is the same as in section 3.2.1.3 and it follows the default settings of the Lighting Analysis Render Preset.

3.3.1.4 Materials

The number of the different material types used in our case is due to the complexity of the outdoor scenes and the high detail of both the terrain models and the human avatar. Listing each material with the corresponding properties is out of the scope of this Thesis.

The approach adopted for the materials accommodates realistic textures mapped on the 3D objects and optical properties that match the typical optical properties for the specific type of any given object. The reflectivity or transmissivity values for the virtual objects are selected according to the Illuminating Engineering Society (IESNA) standard values for building materials [149]. The typical wall reflectivity applied is 0.7, floor reflectivity is set to 0.3, ground reflectivity (albedo) to 0.2 and window transmissivity to 0.77 (single glazing) all according to IESNA standards [150]. The Exposure plug-in confines the use of material libraries only to Arch & Design and Pro-materials. Hence, the materials used for

the scene setup consist of Arch & Design materials with diffuse and bump maps from libraries of the 3D models adjusted to the aforementioned optical properties. Another strategy that could be adopted but is very demanding in terms of equipment is the actual measurement of the optical parameters of the objects and the input of those in the virtual lighting analysis tool. The measurements of reflectivity and transmissivity of the actual objects lead to qualitative results and it is highly recommended when there is access to the actual correspondent object of a virtual 3D model plus access to special equipment like a spectrophotometer. The experimental accuracy test of the lighting analysis plug-in of 3DSMD employs such an approach as the authors of the study measured the optical properties of the original room and objects they modelled in the 3D authoring engine [77].

3.3.1.5 The expansion script

The embedded light analysis tool of 3DSMD does not incorporate a feature for analysing mobile light sensors as the main application of light analysis is in the area of building engineering. For that reason, a script was developed to perform an analysis for virtual scenes containing mobile objects such as the human avatar, in our case. Figure 12 and 13 are the flowchart and a snippet of the aforementioned code correspondingly:



Figure 12 Flowchart of the developed script



Figure 13 Code snippet of the script

The script exports light data with a sample rate that the user can choose. For instance for an animation with a default frame rate of 30 frames per second (fps) and a total of 3000 frames (1 minute and 40 seconds), the user can set the sample rate of analysis to 1 second. In this case the analysed frames will be every 30 frames resulting to 100 measurements.
The measurements are then exported by the same script to a spreadsheet format and imported to MatLab for further analysis. Therefore, in this approach, a commercially available 3D authoring tool is utilised and extended by using its own programming interface and employing an M&S methodology to an application of military interest. This methodology enables the present study to be the first one to analyse mobile light sensors in a virtual environment.

Summary

The application of SGs for Defence and Sustainability, which was examined in the project, is presented in this chapter. The Solar Soldier project investigates the incorporation of the PV technology on the uniform and equipment of the infantry soldier. The aims of the project include the accurate measurement and evaluation of the effectiveness of the integration taking into account usability and the assessment of the effectiveness of each area (amount of power generated under various scenarios) as well as to investigate the areas of the body that yield the best power values for further research on usability (human comfort, intuitiveness). The development of the simulation platform is presented as well as the adopted methodologies and experimental study of system accuracy.

CHAPTER IV: SERIOUS GAMES AND SIMULATIONS FOR DEFENCE APPLICATIONS FOR ENERGY SUSTAINABILITY

Evaluation, Results and Discussion

Abstract:

This chapter constitutes the evaluation and results section of the Solar Soldier Project. It focuses on the research methodologies employed to evaluate the implemented simulation platform presented in the previous chapter. After the description of the corresponding evaluation methods, the results are presented. Following the results, there is a discussion of results and the themes and design guidelines that emerge from that discussion. The structure of the chapter is visualised in Diagram 4below:



Diagram 4 Chapter Structure

4.1 Serious Games and Simulations for Defence applications with Sustainability

4.1.1 Simulation Scenarios

The simulation platform aims to assess the incorporation of PV technology on the uniform and equipment of the modern infantry soldier as described in the Problem description section of Chapter III. A series of simulations is performed to evaluate the platform and produce quantitative results for a series of scenarios that correspond to realistic military scenarios. After the modeling of the environment and its merging with the other virtual elements, the simulation procedure is enabled. Apart from the 3D models, every scene comprises of virtual light sensors attached on various areas of the soldier's uniform. These virtual sensors are the designed light sensors, as described in the implementation section of Chapter III. The distribution and positioning of the light sensors on the uniform of the virtual soldier model is based on suggestions and recommendations after liaising with the corresponding expert of DSTL. For instance, it was noted by the expert liaison officer, that the helmet is a good candidate for integration as it is worn most of the times and in all mission types, it has direct exposure to the sunlight, however when soldiers are under shade it will not produce the required amount of energy. Another example was the backpack that the soldiers usually take off during rest periods and thus can be located in a light-exposed area while the soldier is resting under the shade of a tree. It should be noted that the total number of light sensors to be used in a simulation is restricted (this is typically 8 for most high end computers) due to the high computing power requirements of the simulated scenarios. Figure 14 and 15 depict the positioning of the selected light sensors on the soldier's uniform and helmet:



Figure 14 The positioning of the virtual light sensors (front)



Figure 15 The positioning of the virtual light sensors (back)

Eight light sensors comprise the collection of sensors, namely:

• Right Shoulder Middle

- Left Shoulder Middle
- Right Shoulder Back
- Left Shoulder Back
- Right Forearm
- Left Forearm
- Helmet
- Backpack Top

These light sensors are attached on the geometry of the human avatar, thus they follow every movement that their parent geometry performs. This is a very standard technique for animation sequences of high complexity as is this present case. The objects are linked into kinematic chains so that they follow hierarchical links and the workload of the animator is reduced. Then the parent object dictates, through internal computations, the movement of the child object in forward or inverse kinematic chains.

Several 3D models have been employed for the purposes of our simulation. These can be listed as follows:

- Terrain:
 - Urban area (Arab city model)
 - Forest scene (Pine creek model)
 - Military camp (Base model)

The terrains selected cover the basic landscape of the military deployments and operations of the British Army.

- Human Model:
 - Infantry soldier desert camouflage (Royal Anglian Desert model)
 - Infantry soldier woodland camouflage (Royal Anglian woodland model)

The merging of the models from these two categories of results was conducted in complete scenes, which include a human model and a terrain. According to the daylight system adjustments presented earlier the 3DSMD daylight system is then employed. This system consists of a sun and a compass that is used to calibrate the global position of the scene. After the design of the solar system, the scene is ready to be animated. For the purposes of this study, the animation clips that were selected were that of a walk-cycle, as the effect of each parameter is to be analysed one at a time. The walk-cycle is one of the most typical motions that a dismounted infantry soldier performs on average in most missions. Thus, in the boundaries of this study only the effects of different time, date and global location on specific landscapes (3 case studies) and specific movement styles (simple walk cycle) are

to be examined. The amount of locations to examine globally is also limitless. Since the work conducted is part of a military project of the British Royal Army, a few locations of interest for the purposes of the project were selected.



Figure 16 Major British troops deployments (Source: British Broadcasting Corporation)

Figure 16 illustrates the major deployments of British soldiers around the world. The choice of locations to examine covers various latitude and longitude ranges. Thus, the following strategic points of interest are chosen, which offer distinct locations in terms of both longitude and latitude:

- U.K., Catterick Garrison Headquarters, N. Yorkshire (54.375, -1.708)
- Kosovo, Pristina (42.5, 20.9)
- Iraq, Baghdad (33.33, 44.44)

The methodology of constructing a complete virtual environment for lighting analysis that consists of a terrain, a human avatar, animation clips and a daylight system relies on merging the single 3D models into one scene and adjusting the parameters according to the above standards. The resulting scenes must be simulated in such a manner that they would cover a wide range of times and dates. It must be noted that these types of simulations require high computational time and power due to their high complexity. The range of times would have to cover most of the effective time in the day in terms of lighting; that is some hours before and some hours after midday. The dates would have to cover all seasons at least. Therefore, the selected dates and times under examination are:

- Time:
 - o 10.00
 - o 12.00
 - o 14.00
 - o 16.00
- Date:
 - Winter (07 Jan)
 - Spring (16 Apr)
 - o Summer (08 Jul)
 - o Autumn (26 Oct)

The dates themselves are random but the months are one every four and each from every season. Moving on, as mentioned in the modelling section of Chapter III (Chapter III, section 3.3.1), the sky system employed is the Perez "All weather model" as it accommodates intermediate cases between overcast and clear sky. The 3D software, 3DSMD, requires an input value of light intensity for the simulation and analysis of light in a virtual scene employing a Perez sky model. Thus, the input of solar data is required. These can be taken from existing relevant online sources. As mentioned above the PVGIS project offers valid data online. It covers only the European and African continents. They do however cover the requirements of this study. The online calculator provides the user with monthly average solar irradiance values for a given slope. The measurements required for this study must be taken with the panels in 0 degrees as we do not want to test conventional photovoltaic panels but we need the solar irradiance as the input for our virtual scene. The irradiance data is measured according to the international system (S.I.) with watt per square meter (W/m^2) . The software uses illuminance values which are expressed in lm/m^2 or lux. The conversion of W/m^2 to lm/m^2 is a very complicated and circuitous mathematical procedure that requires the engineer to know the spectral composition of the source in order to solve the conversion formula. A publication in the scientific disciple of horticulture, conducted by Thimijan and Heins [150] provides a table with measured and solved conversion factors for different sources including the sun for various spectrum portions. The conversion factor according to these tables is 0.00804 W/lm. Thus, the interconversion of radiometric to photometric units enables the simulation with typical monthly average irradiance values as input.

The data can be easily extracted by the tool provided by PVGIS¹² online, then converted with the factor mentioned above and finally organised in tables. For the different locations we have the following results (Tables 2, 3, 4):

Table 2 U.K., Catterick Garrison Headquarters, N. Yorkshire (54.375, -1.708). Light intensity in Lux

		10:00	12:00	14:00	16:00
07_Jan	Diffuse ¹¹	6952.2	8582.0	6965.2	1990.0
	Beam ¹	3358.2	5597.0	3358.2	250.0
16_Apr	Diffuse	22823.3	23880.6	22823.3	18470.1
	Beam	22388.0	26119.4	22388.0	12686.6
08_Jul	Diffuse	27674.1	28358.2	27674.1	24253.73
	Beam	27674.1	31343.3	27674.1	17972.6
26_Oct	Diffuse	13992.5	15671.6	13992.5	8022.4
	Beam	10820.8	14303.5	10820.8	2985.1

Table 3 Kosovo, Pristina (42.5, 20.9). Light intensity in Lux

		10:00	12:00	14:00	16:00
07_Jan	Diffuse	13557.2	15422.9	13557.2	6592.0
	Beam	11442.8	15298.5	11442.8	2176.6
16_Apr	Diffuse	23368.1	26990.0	23368.1	21766.1
	Beam	33519.9	39303.5	33519.9	18532.3
08_Jul	Diffuse	27798.5	27985.1	27798.5	24502.5
	Beam	52114.4	59701.5	52114.4	32089.6
26_Oct	Diffuse	18781.0	20273.6	18781.0	12313.4
	Beam	25373.1	31592.0	25373.1	9639.3

Table 4 Iraq, Baghdad (33.33, 44.444). Light intensity in Lux

		10:00	12:00	14:00	16:00
07_Jan	Diffuse	18532.3	20398.0	18532.3	10012.4
	Beam	17786.0	23258.7	17786.0	4664.1
16_Apr	Diffuse	27611.9	28233.8	27611.9	22450.2
	Beam	43159.2	51492.5	43159.2	22077.1
08_Jul	Diffuse	28171.6	28607.0	28171.6	24129.3
	Beam	57524.9	67288.6	57524.9	32388.3
26_Oct	Diffuse	22761.1	24253.7	22761.1	15422.9
	Beam	34519.9	43283.6	34519.9	13059.7

These are the average values for the given days and times for these specific locations. These average monthly measurements only indicate the irradiance value of the given location on a specific date and time. Therefore, the units of the simulated results conform to the units

¹¹ For more info on how these values were calculated see: http://re.jrc.ec.europa.eu/pvgis/index.htm.

of the input values of the simulation. This study approach can derive a set of guidelines of PV integration on military garment and equipment to be used by the PIPV designers and practitioners. How close these values get, can only be validated by a comparison of actual light intensity measurements and simulated light intensity measurements of a mobile (animated) virtual light sensor in a virtual scene shed with the same light intensity as the real environment.

The derived simulation scenarios after all the configurations and setup are described by the following map (Figure 17):



Figure 17 The scenarios map

The differences of each terrain development imply various walking distance and angles, although a general rule was followed; to animate a walk route of a block in order to cover all orientations. The three route distances vary from 1min and 8sec (forest scene) to 1 minute and 22 seconds (urban area scene). An example of the route of the walk cycles is illustrated on the next figure (Figure 18):



Figure 18 Military scene orientation

The frame rate for the animation was the default value of 30 frames per second. For reasons of computational economy, the sample rate of the light analysis script described above was set to 2 seconds. With all these conditions each simulation cycle lasted about an hour on a desktop personal computer with an i5 processor and 4 gigabytes memory.

The average values of each scenario have to be stratified and arranged in increasing order so that a general guideline is established and can be used as future reference by wearable PV designers and engineers. This would form a guide for PV system design stakeholders who wish to position similar devices on other wearable products on human users. In order to accomplish that, we utilise MatLab and its data manipulation and graphical plotting toolsets. MatLab was chosen for its high performance and automation features that simplify the manipulation of such massive sets of data.

4.1.1.1 Power assessment and investigation of integration areas

The previous step aimed towards the analysis of light harvested on individual points on the uniform and equipment of an infantryman. Nevertheless, the requirements of the Solar Soldier project specify an analysis of the power generation capabilities for the proposed PIPV system. Therefore a further analysis of the output data from the previous section is required, in order to assess the power generation efficiency of each of the proposed areas. The extent of each area under investigation is defined as an entity by the variance. It shows with the centre of each aforementioned area, on which area the light sensor was attached for reasons of computational economy. In other words, the areas extend from the light sensor up to the point they show significant deviation. From that point and beyond, the area is taken as being another entity. Figure 19 depicts each of the proposed areas and their extent in cm²:



Figure 19 Areas of integration

The light intensity data is imported into MatLab using the appropriate developed script. Then the units are converted appropriately and further analysed to derive the power generation assessment of each area. Therefore, hitherto the concept of sensor is replaced by the concept of integration area. Using the value of the area (in cm²) and the efficiency value (5%) of the prototype PV device developed for the Solar Soldier project by our corresponding project partners, it is now possible to calculate and produce the average power graphs of each sensor (in watts) [175].

4.1.1.2 Combining quantitative methods with user-led design and feedback

The simulation scenarios illustrated in previous section (section 5.1.1) provide a realistic augmentation of the military environment in to the virtual scenes. The results of the simulations are quantitative and represent the actual values of light harvested on the designed sensors and with further analysis provide us with an insight of the power generation assessment for each and every area of integration. Nevertheless, when designing for such a complex aspect such as integration of photovoltaics on an infantry soldier, one needs to take into account the human factors and accommodate for them in the design process. Thus, in the process of assessing the integration of the PV technology, the enduser was involved and provided us with valuable input to the design. From reference material such as online videos of real-time battle (for example Ross Kemp in Afghanistan as on Figure 20 [176]) or liaising with the corresponding office of the MoD, which is the DSTL.



Figure 20 Ross Kemp in Afghanistan (Source: British Sky Broadcasting Group plc 2008)

The expert liaising officer of DSTL offered a series of suggestions and recommendations, from the most appropriate areas for integration in terms of human factors, to integration mechanisms and scenarios guidelines. The valuable input was recorded and taken into account when designing for the simulation platform.

4.1.2 Results

After the establishment of the platform's accuracy, the next step is to perform the simulations. These simulations result to the classification of the various areas on the uniform and equipment in the context of higher light intensity incident on these areas. Engineers and designers can then, by using the data of this study, have a draft blueprint of how and where to incorporate the PV device on the soldier's uniform and equipment. The results are organised in two sets of graphs; one contrasting the seasons and one contrasting the locations. There are a total of 144 scenarios simulated with 8 sensors each. For reasons of economy, only one case is presented as a single graph and the rest are organised in groups and presented in Table 5. The results, acquired by the helmet sensor in an urban area scene on the 26th of October and for all locations (Baghdad, Pristina, Catterick Garrison), are illustrated in the next figure (Figure 21):



Figure 21 Helmet sensor light levels for all locations, 26th of October

This figure consists of four plots, one for each time of day. Every plot consists of six graphs, one for each location and their corresponding average values (see legend). The fluctuations of the graphs are due to the change of orientation of the walk cycle that the virtual avatar performs as well as the shading effect of the interaction of the avatar with the virtual environment. The most significant element in this set of results is the average value that describes the quantity of light that will be harvested by the point under investigation, in this

case the top of the helmet. This value is the most important as a reference of how much light a specific area on the avatar can absorb during a normal walk cycle and under the specifications described in the methodology. That value can be used as a guide by engineers who wish to design a system as independently as possible from the current bulky and unwieldy energy storage systems. The average values for each case are organised and presented in Figures 1A-9A in Appendix A, in a manner that allows the comparison and stratification of the selected light sensors, as to which are the most important areas for light absorption. Afterwards, the results are interpreted and the classification of the sensors can be inferred by contrasting their average values by seasons and locations for each environment. The contrast of season and location parameters is presented in Table 5 below (in descending order):

	Scene			
Location	Forest Scene	Military Base	Urban Area	
Season:	January/April/July/October	January/April/July/October	January/April/July/October	
Baghdad	 Helmet Forearms Backpack Shoulders Middle Shoulders Back 	 Helmet Forearms Backpack Shoulders Back Shoulders Middle 	 Helmet Shoulders Middle Forearms Backpack Shoulders Back 	
Catterick Garrison	 Helmet Forearms Backpack Shoulders Middle Shoulders Back 	 Helmet Forearms Backpack Shoulders Back Shoulders Middle 	 Helmet Shoulders Middle Forearms Backpack Shoulders Back 	
Pristina	 Helmet Backpack Forearms Shoulders Middle Shoulders Back 	 Helmet Forearms Backpack Shoulders Back Shoulders Middle 	 Helmet Shoulders Middle Forearms Backpack Shoulders Back 	

Table 5 Sensor classification per scene

The table above indicates that the change in location or season does not affect the order in the classification immensely, although the divergence between two sensors is not constant. Hence, the classification of the sensor for each scene can be simplified to the following (Table 6):

Scene	Sensor Classification		
Forest Scene	1. Helmet		
	2. Forearms		
	3. Backpack		
	4. Shoulders Middle		
	5. Shoulders Back		
Military Base	1. Helmet		
-	2. Forearms		
	3. Backpack		
	4. Shoulders Back		
	5. Shoulders Middle		
Urban Area	1. Helmet		
	2. Shoulders Middle		
	3. Forearms		
	4. Backpack		
	5. Shoulders Back		

Table 6 Overall classification of sensors

Table 6 indicates that the optimum position, in terms of light harvest, for the integration of the PIPV on the soldier's uniform is the helmet. One can clearly see that in all locations, scenes and seasons the helmet sensor consistently yields the highest values. It is also interesting to note that the main changes in the classification of the sensors are according to location and scene but not season. More precisely we can see that the forearm is the second best location for the positioning of PIPV for the scenarios in the forest and military base scenes in Bagdad and Catterick Garrison as well as the military base in Pristina. However this is not the case for the scenario of the urban scene across each location, where the middle shoulder sensor yields the second highest value. As depicted in Table 6, the backpack sensor yields the third highest values for the majority of scenarios, such as those of the forest scene and military base in Bagdad, Catterick Garrison and Pristina. There are a few differences such as in the scenario of the forest scene in Pristina and the scenario of the urban area across all locations, where the forearm sensor comes third. We can also ascertain that the shoulder back sensor is the least optimum sensor for the integration of PIPV as it yields the lowest light values for the vast majority of scenarios and in comparison to other sensors.

4.1.3.1 Power generation assessment

The classification of light harvesting of the potential areas of integration reveals the areas that the simulations showed to be the most optimum for light harvesting. However, the power generation depends on the extent of the area itself and thus a further assessment has to be made to ensure the classification is valid. The further analysis of the simulation results in MatLab as presented in Chapter III, infer the average power values in W for each of the scenarios and for all areas. These average power values are organised and presented in Figures 1B-9B in Appendix B. The results are interpreted and the classification of the areas can be inferred by comparing the values of each season. Table 7 provides the overall classification in terms of power generation for the examined scenarios as well as the extent of each area in cm²:

Scene	Are	ea		
T (sincation	200?	
Forest	1.	васкраск	300cm ²	
	2.	Helmet	314cm ²	
	3.	Forearms	$100 \mathrm{cm}^2$	
	4.	Shoulder	70cm ²	
		Middle		
	5.	Shoulder		
		Back	60cm ²	
Military Base	1.	Helmet		
	2.	Backpack		
	3.	Forearms		
	4.	Shoulder Middle		
	5.	Shoulder Back		
Urban Area	1.	Helmet		
	2.	Backpack		
	3.	Forearms		
	4.	Shoulder Middle		
	5.	Shoulder Back		

Table 7 Overall classification in terms of power generation

Here it is made evident that the areas with the most extended as well as high light harvest capabilities are dominant in terms of power generation.

4.1.3 Design Guidelines for the Integration of Photovoltaic Technology on the Modern Infantry Soldier

As stated in the Introduction, the usability of the PV device proposed by the Solar Soldier project is examined by liaising and interacting with DSTL. This interaction derived a preliminary set of guidelines for the integration of PV on the uniform or equipment. The feedback we received from DSTL enabled us to narrow down the potential areas where PIPV could be integrated and thus reduced the number of light sensors to use in our simulations. For instance, it was ascertained that the chest and the back of the uniform areas would not constitute good candidate areas for installation of PIPVs as they are constantly occluded by the gun and hands holding it and by the backpack respectively. The second set of guidelines derived from the case studies of the three different environments presented above. Table 7 provides a draft guideline for designers and manufactures of wearable PV devices in military environments. It is clearly shown that this classification can provide guides for the positioning of such devices on the uniform and equipment of the infantry soldier for the examined areas. For instance, the helmet would be the first choice for every case and environment. This fact was more or less predictable yet not validated by any study so far. Moving on, we notice that the top of the backpack as well as the forearms as a set (right and left) qualify as important area candidates for integration. The remaining areas qualify only as supplementary areas as they show poor performance and come low in most classifications. Combining the simulation data presented in section 5.1.2 along with the feedback on the HCD ascertained from DSTL we can provide the following set of guidelines and recommendations with regards to the integration of PIPVs on the modern infantry soldier:

- i. The best places on the soldier's uniform in terms of ergonomics and power generation are the helmet followed by the backpack and forearms. These three positions will provide the PV system with constant exposure to solar radiation, which can be converted to energy even when the soldier is on the move.
- ii. It is recommended that the entire backpack is covered with PVs as, on the one hand, more PV panels can be placed there and thus more energy can be harvested at all times as the soldier can easily leave the backpack in the sun whilst resting. Although the helmet yields the highest amount of light its consistent supply of power to the solar harvesting system may be stopped in certain cases. In very warm environments of operations the soldier will seek shade under natural and man-made constructions such as trees and buildings and may even take off or sit on the helmet whilst resting. This necessitates further the need to place PVs on the backpack as it can be removed and placed under the sun.

iii. Integrating PV directly into the uniform is not recommended as this is washed in extremely boiling hot water. As fabric and nano-material technology evolves it may be possible to interweave the solar panel nano-material onto the uniform that will withstand extremely high temperatures. Until then it is recommended that the solar panels are attached onto Velcros so that the PV can be attached and detached. This would also enable the interchange of the PV positioning on the uniform according to the environment and location of operation.

Summary

The Evaluation, Results and Discussion Chapter of the Solar Soldier Project describes the evaluation methods utilised for the project investigated and furthermore offers the results and discussion of the evaluation. Moreover, there is the summary of the discussion into design guidelines for PIPV in the military sector and for integration on the uniform and equipment of the modern infantry soldier.

CHAPTER V: SERIOUS GAMES FOR MOTOR REHABILITATION

Problem Description, Proposed Solution and Implementation

Abstract:

This chapter provides an elaborate presentation on the subject of Serious Games for Motor Rehabilitation presenting the work conducted under two individual projects. Both of these projects examine the application of video games in a novel rehabilitation framework for motor training for neurophysiological and neurodegenerative diseases such as stroke and Parkinson's disease. The related literature in this area is presented in Chapter II and the summary of this work as well as the definition of the research challenges and opportunities is offered in this chapter. Furthermore, the implementation and methodologies adapted for the design and development of the Serious Games for Motor Rehabilitation are described in more detail. The structure of the chapter is visualised in Diagram 5 below:



Diagram 5 Chapter Structure

5.1 A rehabilitation framework for Motor deficits utilising Serious Games

The ageing population and the number of people affected by motor impairments are constantly increasing. Neurological conditions (such as stroke or PD) affect the elderly in large numbers and the common practice in terms of treatment has traditionally been physiotherapy and physical exercises. SGs developers have realised the application of virtual reality games in motor rehabilitation as seen in the literature review in Chapter II. These games have steadily gained popularity as they suggest a number of benefits over the conventional physiotherapy interventions. These are the customisation of the SG to match the patients' requirements, the lower costs associated with their deployment, the remote monitoring of the performance by the clinician, the increased motivation and adherence to the therapy through an enjoyable playing session. Stroke and PD are amongst the high impact diseases and are dominant in terms of population, therefore one would require to examine the number of cases worldwide.

This Chapter presents the research conducted on SGs for motor rehabilitation within two projects. A framework has been designed and developed which facilitates the use of bespoke game sensors, in this case the Nintendo Wiimote, and customised SGs for neurophysiological and neurodegenerative diseases; stroke and PD. The aims of the framework, and thus of this chapter, are the following:

- Construct a backbone system architecture that is structured with reusable components in a manner to enable flexibility in the selection of COTS game sensors for the motion capture and more options for the development of the SGs
- Produce a series of SGs for motor rehabilitation with the aid of the patients (user-led design) and evaluate their design via pilot testing (presented in Chapter VI)

The first project (entitled Research on Wii Rehabilitation, ReWiiRe¹²) was jointly funded by Brunel University and the National NHS. The scope of this project is to investigate the use of customised game sensors and SGs in motor rehabilitation of stroke patients. The second project was funded by Brunel University and the work was supported by Queens University in Belfast. This project investigated the same subject, but this time for PD patients. A series of games were developed for each of the cohorts of patients as their individual requirements were taken into account. The SGs were pilot-tested in both cases in an iterative manner to involve the patients into the design as much as possible. The patient-led design approach revealed a series of design guides that steer the development of SGs for neurophysiological and neurodegenerative diseases. The author's contribution to these studies was to investigate the accuracy of the Wiimote motion data captured by the backbone system, to develop the SGs and design the clinician platform, to run the pilot tests in collaboration with the expert clinicians and further synthesise the design guidelines

¹² ReWiiRe project website: http://www.rewiire.org.uk

through the analysis of the findings. Below is the description of the problems and summary of the literature research findings of Chapter II:

5.1.1 Serious Games for Stroke

Stroke is a life-threatening condition where the brain has inefficient blood supply and this is due to either ischemia (loss of blood flow) or haemorrhage which causes bleeding in the brain. Stroke is a leading cause of long term disability worldwide including in the UK where an estimated 152,000 new strokes occur each year, leading to an annual economic burden of more than £2.8 billion [151]. Following stroke, 70% of people experience arm weakness, with only 5–20% regaining full function [152]. Treatment interventions shown to be most effective are characterised by high intensity, repetitive and task specific properties, but there are resource challenges in providing this level of intervention. Stroke rehabilitation can be a long process; as such, much of it takes place in the home setting. However adherence to home exercise programmes (HEPs) is poor with a recent study finding 79% of patients in the community stopped doing their HEP after two days [153]. A perceived lack of support from physiotherapists was the most frequently mentioned factor associated with lack of compliance. Boredom, de-motivation and depression also play a part in patients not engaging with exercise [153].

Several stroke VR-based rehabilitation interventions using robotic systems, custom made IMUs and body sensor networks have been proposed over the years [154]–[158]. However, many of the current interventions in rehabilitation are either very expensive (i.e. robot-aided systems) or not suitable for use at home or simply very difficult to enter the household, and tagged as health-related devices. The Nintendo Wii, on the contrary, has already been in the household and is a user-friendly device that is introduced and associated to the patient more as a game sensor and less of a health device. The console has sold 101,060,000¹³ hardware units, and this fact shows the market-penetrating power and potentials of such a device. Furthermore, unpublished data by the author and colleagues

¹³ <u>http://www.nintendo.co.jp/ir/library/historical_data/pdf/consolidated_sales_e1403.pdf</u> (last accessed Apr 2014)

(ReWiiRe project) dictate that the patients and their families manifest preference to the COTs game consoles as they associate them with fun and engaging activities and not with rehabilitation. The use of medical devices is bound to be associated with the disease and although the games might be extremely fun and engaging the experience is linked with the disease through the use of the medical device. Due to this fact, the intervention is perceived more as rehabilitation intervention and less as a fun and entertaining experience with an underlying purpose. The innovative use of the Nintendo Wii could present a way to enhance community based stroke rehabilitation of the arm by providing a motivating and cost-effective way of exercising that can be remotely monitored and exercises adjusted as required by a physiotherapist.

5.1.1.1 Summary of related work: challenges and opportunities

Following the related work presented in Chapter II and aforementioned sections a list of key requirements can be drawn for the design and development of an effective stroke intervention system that employs the Nintendo Wii remote technology:

- *Full kinematic tracking of the user upper limb movement in real-time:* Current systems offer limited tracking for specific type of movements (limited by IR's line of sight). Also the direction of movement cannot be recognised accurately if the motion involves both accelerating and rotating [51, 172].
- *Remote progress assessment and interaction between therapist and patient:* None of the current Wii-based systems proposed in the literature tackles this.
- *Personal connection and instant feedback between the VR environment (e.g. 3D Avatar) and the patient movement:* Very few systems offer this; [111] uses a 2D avatar and [109] employs a 3D avatar however in both cases the patient movement is limited or not accurately represented onto the VR environment.
- Support carer/therapist assistance and provide collaborative mode of gameplay/exercise: Depending on the severity of disability and individual stroke patient range of motion, a carer or therapist often assists the patient in conducting the exercises. The vast majority of Wii-based interventions, which employ IR suffer

from a drop in accuracy and loss of data when 2 users are very close, e.g. a therapist helping a patient. Also collaborative mode of play and exercise is not supported [51].

• *Customised games and exercises appropriate to stroke users:* Most of the proposed systems offer customised games. Some offer customised exercises [51].

Therefore, taking into account all key features found in the literature up-to-date, as well as any lessons, we learn out of experience with users and pilot testing of the designed games, will form a broader design approach which incorporates all the above design requirements and guidelines. These design guidelines foster the key features of game mechanics and gameplay in order to maximise the engagement of the patient with the rehabilitation interventions. The same design guidelines will form a design practice for developers and clinicians that want to further investigate the effectiveness of SGs for stroke rehabilitation.

5.1.2 Serious Games for Parkinson's Disease

PD is a slowly progressing neurodegenerative brain disorder. PD is caused by the reduction of the neurones in the substantia nigra, the brain area responsible for the dopamine production. This fact itself provides enough evidence for the debilitating motor and non-motor symptoms of a Parkinson's patient.

It is estimated that 6.3 million people suffer from PD worldwide [159], with no discrimination to race or cultural background. According to statistics, PD patients in the USA number about 1 million [160], whilst in Europe there are over 1,2 million [161] and in the UK alone there are about 120,000 patients, according to HESonline [162] and Parkinson'sUK [163]. The age group mostly affected by PD is between 60-79 years [163], nevertheless there are cases of PD in the age range of 20-39; the "the young outset" of PD. There are four key motor symptoms associated with PD, namely tremor, postural instability, rigidity and bradykinesia [164].

Although research into finding a cure is always continuing, there is no irrevocable or radical treatment for PD up to date. There are a series of drugs that are widely used to support the management of the symptoms. Depending on the severity of symptoms there are also a

series of surgical options available. Apart from drug-based solutions, there are a wide range of therapies available for patients to deal with everyday tasks and activities. According to Parkinson's UK [163] the therapists provide to PD patients as series of options that include, but not restricted to, courses of occupational therapy, physiotherapy, diet plans and speech and language therapies. The well-being of the patient can also be increased by acupuncture, massage therapy and aromatherapy.

Physical rehabilitation is a common treatment and a management tool prescribed very often by clinicians in the recent years. The prescribed exercises target the particular symptoms and studies of their short-term effects show beneficial results. Particularly, the study of Tomlinson et al. [165], presents an extensive review of physiotherapy treatments and their assessment of effectiveness. Along the same lines, the review of Goodwin et al. [166] examined the effects of physiotherapy for PD and found evidence of physiotherapy being very beneficial to aspects of the patient's life like physical functioning and well-being. Moreover, Nimwegen et al. [167] studied the physical inactivity of PD patients by quantifying the levels of everyday activities and comparing the levels of a patient cohort with a control group. As expected the levels of PD patients' physical activity were lower than the control group. However the variance was not completely justified by the motor impairments, suggesting other reasons behind inactivity such as engagement to physiotherapy and enjoyment. Finally, the study of Farley & Koshland [168] [169] shows evidence that the training of speed with movement amplitude in the limb motor system reduces the symptoms of bradykinesia and hypokinesia. This rehabilitation intervention, which is referred to as Training BIG, applies the established and proven principles to be beneficial treatment for speech rehabilitation for PD, Lee Silverman Voice Treatment (LVST) [170], to the limb motor system.

5.1.2.1 Summary of related work: challenges and opportunities

The related work presented above enumerates the features and practices of game design for PD rehabilitation interventions as well as general design guidelines for SGs for rehabilitation. The features of the games and design guidelines presented above can be

summarised, since they have been found beneficial, to the following requirements when designing games for PD:

- *Accuracy:* The motion capture system, utilised for the purposes of a study as this, should be very accurate as the biofeedback data is to be analysed to evaluate the performance and progress of the patient [126].
- *Home-based solution:* The gaming sensors that are to be employed for the motion capture need to be commercially available and of low-cost [171].
- *Real-time biofeedback:* The rehabilitation platform should realise the potential of real-time biofeedback to the clinician through a dedicated clinician platform [126].
- *Customised Games:* The designed games should enable visual and auditory cues, and adjustable level of difficulty that can monitored remotely by the clinician. The genre of these games is also referred to by the term "exergames" which is a portmanteau of exercise and game [51].
- *PD Rehabilitation Protocol:* The designed games should be a translation of Training BIG exercises into VR games. The proposed system must be very flexible so that the translation of any exercise into a mini-game must be very easy to accomplish [125].
- *Automated system Calibration:* The system should incorporate an automatic or semi-automatic calibration system that will match the ROM of the patient with the ROM required by the virtual game player [135].
- *Feedback/reward system:* A motivating and encouraging feedback system increases entertainment and thus the engagement and involvement of the player with the game and reduces the risk of abandonment of the game and physiotherapeutic treatment [118,121].

Again in this case and similarly to the SGs designed for stroke, the design guidelines will be formed out of the aforementioned key-features and best design practices that come up through the interaction of the patients / users with the games and their involvement in the design process during the pilot-testing sessions. The game design is therefore considered as patient-led as the patients are actively involved in the design and their feedback is leading the design practice. In addition, the design initiates from the physiotherapy and also facilitates high level of adaptability. The user study for evaluating the proposed design framework is also presented in a following chapter (Chapter VI).

5.1.3 Emerging themes for developing Serious Games for motor rehabilitation

The discussions above show some similarities in the summarised challenges and opportunities for rehabilitation of motor deficits that can emerge into common themes. The research of the available literature for stroke and PD resulted to a range of requirements for designing SGs for the said diseases. These requirements are dependent on the specific requisites of the disease. However there are some points in the summaries of the literature above that seem to match. Thus there is the opportunity to draw a wider picture and collate a generic summary of requirements for motor rehabilitation of the upper limbs based on the common themes. These themes are the following:

- Home-based solution: This is requirement is essential for a variety of reasons.
 Firstly, the decongestion of the hospital environments and the lower-cost of rehabilitation is a very important advantage for all current national Health systems.
 Furthermore, the safety and comfort of one's own house is an advantage that needs no further explanation. Lastly, the association of the intervention with an entertaining experience and not with a clinical routine increases the engagement with the game.
- Remotely monitored intervention: The clinical supervision of the rehabilitation regime and adherence is a crucial element of success for the rehabilitation process. Thus, the monitoring of the performance is a requirement which, due to the homebased manner of the games, is transformed to remote monitoring.
- Accuracy: The games developed for the purpose of motor rehabilitation have twofold aim. They need to be fun and entertaining to maintain the engagement of the patient. At the same time, there needs to be the underlying rehabilitation target which is the ultimate purpose of those SGs. Therefore, the game mechanics involve

the element of accuracy of MoCap, as this data is supposed to be assessed by the monitoring tools of the clinician in order to analyse the performance.

• Customised games: Accounting all the reasons above, it is evident that there is no COTS game to fulfil these requirements. The tailored games for motor deficits will incorporate by design these elements. Moreover, other requirements that are included in the customisation of SGs for rehabilitation are the automatic adaptation to the patients' movement capabilities and motivating scoring and feedback system to support the diversity of players' age groups and emotional states.

The system proposed in this Thesis fulfils all of the requirements above and hence, it is used in both cases. The system is developed in a manner that can be easily customised to suit the specific requisites of each disease as it comprises of reusable components which can be easily readjusted to match the scenarios. More information on the system architecture and framework can be found in the following section.

5.2 The Framework

The adopted methodology and the various stages of the projects are illustrated in the block diagram below (Figure 22):



Figure 22 Block diagram of the adopted methodology

The methodology depicts the various stages of the projects. Firstly, the investigation of the accuracy of the proposed system is pursued through experiments described in section 4.2.2. After the accuracy is established, the game design and development process is initiated. The design of the games is based on the translation of existing physiotherapy exercises into games. The iterative process of design is led by the patients taking part in the evaluation of the games. Their suggestion and feedback drives the implementation and development of the mini-games as the patients will be the end-users. The game development requires a collection of 3D assets which is collated by online resources and freely available assets. Furthermore, there is the requirement of a series of scripts to operate various tasks within the game engine. More details on the scripts developed can be found in the section on the implementation (Section 4.3).

The literature review presented in Chapter II outlines the potential application for homebased rehabilitation strategies that utilise commercially available gaming sensors like the Nintendo Wii and bespoke computer games. The summary of the scattered guideline elements found in the literature as well as novel approaches for SGs for PD has informed the formulation of a design framework. The motor rehabilitation design framework, depicted below in Figure 23, consists of three layers. The first layer is a motion capture system, in our case utilising the Nintendo Wiimote. This layer's output is in the form of quaternion (hyper-complex numbers widely used in calculations of 3D rotations) therefore, the sensor can be of any type given that it can output in a quaternion motion data form. This way the proposed framework is sensor independent. However, for the purposes of our study and for the reasons of accuracy and following the technology limitations discussed in Lewis and Rosie [107], we are using the Wiimote. The second layer consists of the custom VGs layer to correspond to the rehabilitation targets and requirements. Within this layer the mapping of the motion data on the VR props (e.g. game elements, such as paddles in our rowing game) is taking place as well as the calibration, automatic difficulty adjustment and feedback to the player. The third layer is the VR instructor platform that aims at facilitating the clinician with live biofeedback and the possibility for level of difficulty adjustment. Figure 23 illustrates the schematic diagram of the layers of the proposed framework along with their interactions and the sections below describe them in more depth.



Figure 23 Schematic diagram of the system architecture

The design features of accuracy, adaptability and calibration are covered in this system architecture level by providing the appropriate hardware infrastructure. The remaining design elements will be covered within the game environment through the appropriate design framework. The game engines utilised for the purposes of the game development are two, namely Unity3D and Blender.

Unity3D is a game engine with cross-platform capabilities that tends to become the industry standard especially for indie productions and web plugins, console and mobile games. It comprises a 3D game level editor and an integrated development environment, which offers

a selection of programming languages such as C#, Java and Boo which is a Python-based language. Unity3D is owned by Unity Technologies and is offered in a free version, with the latest version to date being v4.5.2.

The second software utilised for the development of the SGs for motor rehabilitation was Blender. Blender is an open-source 3D authoring platform that incorporates a game engine as well. Within the environment, the user can design or import the 3D assets of the game as well as the game scripts in Python. The latest version of Blender to-date is v2.71.

5.2.1 Motion Capture System

The motion capture system employed for the purpose of this research comprises an off-theshelf gaming sensor, i.e. the Nintendo Wiimote, and a bespoke communication interface between the Wiimotes (up to two) and local computers via Bluetooth and remotely via the Internet. The motion data received by the terminals undergo a process of smoothing and multiplexing using a data fusion algorithm in order to achieve higher accuracy and precision. The end results are mapped into quaternion forms that translate the orientation of a constructed 3D body model and also are free from gimbal-lock. The angular rate measurements captured by the gyroscope sensor can be used to distinguish true linear motion from the accelerometer readings. The gyroscope is not free from noise, but since the measured rotation is less sensitive to linear mechanical movements and without amplifying hand jitter, both of which accelerometers suffers from, it allows capturing more complex orientation with a relatively better estimate than we would be obtained by using accelerometers alone. A sensible approach for maximising efficiency is to average or concatenate the data that comes from the accelerometer and gyroscope by using a data fusion algorithm and simultaneously, it has been accomplished to employ a smoothing algorithm to remove any excessive noise from the signals while still retaining the useful information. Filtering out and removing as much random noise as possible from the sensors' output raw information whilst retaining quality data is of fundamental importance [171],[172]. The Wiimote sensors are very responsive, but they cannot respond to the linear movement that accelerometers specialise in. Yet, as described in the above section, when a gyroscope and an accelerometer are combined, the pairing of sensors facilitates a highly

accurate one-to-one representation of the control device in 3D space. The quaternion data is forwarded locally or online and thus can manipulate a virtual 3D object on the local computer (exergame) or provide the biofeedback to the clinician's party (remote terminal). For more detailed information on the motion capture system the reader is advised to refer to the work of Tsekleves et al.[171].

5.2.2 System accuracy study: Methodology and experiment setup

Accurate biofeedback is a key target of this proposed framework. Thus, the accuracy of the MoCap system is a major challenge and it needs to be addressed so that there is a baseline measurement of our system's accuracy. An experiment has been setup to shed light on this matter. The experimental comparison of our system with an industry established accurate system such as ViconTM MoCap Studio at Brunel University provides a comprehensive methodology to achieve this purpose. The studio comprises eleven infrared cameras and a MoCap software platform, namely Blade 1.7. The software offers a calibration procedure that matches the camera positioning with the cameras in the 3D environment and reconstructs the actual scene into 3D or else matches the camera frames with the enviroment frame. The highly sophisticated MoCap studio is capable of achieving accuracy of sub-milimeters. As our system is markerless, there will be no interference between the two systems. A six-rod structure has been built to support the Wiimote and provide reference of position into the Vicon platform. For that reason, one reflective marker is attached at the end of each equidistant rods. On one of the rods, there are two markers to distinguish it as a reference. (see Figure 24 below). The structure is wooden as this material does not have any magnetic properites and will not interfere with the measurement units of the Wiimote. The resulting object within the VR environment is supposedly an octahedron with the markers acting as its vertices. The center of the octahedron is where the Wiimote is placed. As the individual vertices of the octaedron are captured and their motion data is obtained and recorded, it is fairly straightforward to calculate the motion data of the centre which has been designed to be their actual middle point as seen on Figure 24 below. The structure is supported in two stands so it can rotate freely along any axis. We tested all axes by a rotational pattern of +90, -180, +90 degrees and for 3 repetitions each time to compile a dataset that could render the system performance. The recordings are taken simulteneously of both systems for time reference.



Figure 24 Accuracy experiment setup

5.2.2.1 Experimental results and discussion

The experimental approach of the accuracy test of our proposed MoCap system is described

in the section above (Section 4.2.2). The study of the accuracy enables the biofeedback

platform to be used as a reliable tool by the therapists. The results presented below (Figures 25, 26, 27) are very encouraging as they are very close to the data gathered by the Vicon system which has an established accuracy level of sub-millimetre.



Figure 25 x-axis motion data



Figure 26 y-axis motion data



Figure 27 z-axis motion data

The movement pattern shows high resemblance between the Wiimote and the MoCap systems and the quantitative analysis of the Wiimote system data showed a range of accuracy between 5.5%-10.78% (5.5%, 9.39% and 10.78% correspondingly for each of the axis depicted in the graphs above) residual value.

This value is the Normalised Root-Mean-Square Deviation (NRMSD) which is commonly used as a formula to calculate deviation for such cases as ours:

$$\text{NRMSD} = \frac{\text{RMSD}}{x_{max} - x_{min}} (1),$$

where
$$RMSD = \sqrt{\frac{\sum_{t=1}^{n} (y_t - \hat{y}_t)^2}{n}} (2),$$

where n = number of samples for each recording.

These residual values show that our system, as shown in the graphs, is accurate compared to a commercial and sophisticated MoCap system with an acceptable accuracy level to be a valid candidate for the purposes of a biofeedback platform. Our platform is proposed as an alternative home-based rehabilitation solution with monitoring capabilities. As such, the accuracy level of the MoCap is important but not crucial as it is only to be used as an indicatory factor of performance. The clinician can conclude with reasonable accuracy if the patient's condition is improving and have a reliable estimate of the quantitative values of performance. It has to be noted here, that the error of the measurements is due to a broad

range of reasons. From magnetic fields that affect the performance of the IMUs to synchronisation errors during the data capture process, the NRMSD values are an indicator that the proposed system will give results of acceptable accuracy for a home-based and tele-monitoring system, however with certain limitations.

5.2.3 Automatic Calibration and difficulty adjustment system

As stated in the related work of Chapter II (Section 2.3), the ROM of each patient differs significantly. Moreover, this range has to match the ROM of the game prop, whether that is a virtual avatar or just a game object such as a boat oar. Patients with very restricted ROM have to be available to perform wide a ROM in a game so they are not discouraged by failure. Hence, a calibration algorithm has to be employed to resolve this issue by registering the maximum ROM and matching it with the desirable ROM in the virtual environment. That calibration can be either be performed before the game starts or whilst it is playing to dynamically adjust the ROM of the player. That is due to the fact that a lot of people tend to extend their movement after some repetitions and the appropriate warmup of the first few attempts or in accordance to the game level and excitement. If the maximum ROM is recorded and sent online to the clinician platform, then it is very easy for them to adjust the desired ROM of the virtual player and therefore adjust the difficulty of the game. This can be implemented by an impact factor set by the clinician so that for example every 10° of actual rotation of the patient's limb, the virtual object registers only a fraction of that in the game. This way the clinician can adjust the difficulty on demand and according to the specific requirements of each player.

Moreover, we propose an automatic ROM calibration that can be overridden by the clinician in case (s)he feels that they should push the limits of their patients for better results. The automatic calibration does not require any prior performing of the game movement from the patient. The principle is that while the patient performs a repetition of the exercise, the system records the ROM by the corresponding minimum and maximum values of each axis of movement. Then, it is easy to calculate the ROM and adapt it with further formulas to the desired ROM of the game prop. An advantage of this approach is that the ROM is dynamically calculated and the mapping of the movement to the virtual

environment dynamically adjusted. This way the improvement of the patient's motion will result to a higher difficulty level and thus provide motivation to reach further. All of the above can be manipulated live by the clinician as previously described. The system is easily adapted to any kind of scenario as the game designer only has to store the ROM of the targeted game object. As seen in the data flow diagram below (Figure 28), the game designer inputs and stores the value in the process of the game development and the algorithm can be applied in any kind of system. At the other end, the system is game-sensor independent, which means that there can be any kind of motion capture sensor.



Figure 28 Data flow diagram of the Motion Data Calibrator / Adaptor

Following is a snippet (Figure 29) of the code written for Unity3D game engine and was used in a pilot game (rowing game, section 4.3.2.1) presented in a following section.

```
using UnityEngine;
1
2
      using System.Collections;
 3
 4
    -public class Callibration : MonoBehaviour {
5
           public float[] angleValuesR;
 6
 7
          public float maxR = 0;
8
          public float minR = 0;
          public static float sR = 0;
9
10
          public static float offsetR = 0;
          public static float rangeR = 0;
11
12
          public float[] angleValuesL;
13
          public float maxL = 0;
14
          public float minL = 0;
15
          public static float sL = 0;
          public static float offsetL = 0;
16
17
          public static float rangeL = 0;
18
19
          // Update is called once per frame
20
     Ē
          void Update () {
21
               for (int i = 0; i<50; i++) {</pre>
22
     È
23
               angleValuesR = new float[50];
24
               angleValuesR[i] = TEST_Paddle.euler_1.y;
25
26
               angleValuesL = new float[50];
27
               angleValuesL[i] = TEST_Paddle.euler_2.y;
28
     自日
29
                   for (int y=0; y<50; y++) {</pre>
30
                       if (angleValuesR[i] > maxR) {
31
                           maxR = angleValuesR[i];
32
                       }
     È
33
                       else if (angleValuesR[i] < minR) {</pre>
34
                           minR = angleValuesR[i];
35
                       3
     ╘
36
                       if (angleValuesL[i] > maxL) {
37
                           maxL = angleValuesL[i];
```
38	-		}
39			<pre>else if (angleValuesL[i] < minL){</pre>
40			<pre>minL = angleValuesL[i];</pre>
41	-		}
42	-		}
43	-		}
44			
45			
46			<pre>rangeR = maxR - minR;</pre>
47			sR = 140 / rangeR;
48			offsetR = (sR*maxR) -70;
49			
50			<pre>rangeL = maxL - minL;</pre>
51			<pre>sL = 140 /rangeL;</pre>
52			offsetL = (sL*maxL) -70;
53			
54	-	}	
55	L }		

Figure 29 Code snippet

5.3 Implementation of Serious Games for Motor Rehabilitation

The implementation of the mini-games for motor rehabilitation of the neurophysiological and neurodegenerative diseases described above is provided in this section. The pilot SGs for stroke and PD are presented in separate sections and the processes of design and development are offered with more details. Below there is the Game Data Flow Diagram with UML notation (Figure 30) as a generic data flow diagram for any kind of game within our framework. It also illustrates the basic system components that exist in all the games presented in the following sections.



Figure 30 Generic Data flow diagram

The component of Interface, as depicted in Figure 31, is essential for all games as this is the one to communicate the motion data between the Wiimotes and the game environments, via the Internet and TCP/IP protocol. Figure 32 illustrates the code in C# written for the client side¹⁴ of the network sockets for TCP/IP networking.



Figure 31 The Interface component

¹⁴ The corresponding server side is standalone software developed in the boundaries of ReWiiRe project. For more information please visit: hhtp://www.rewiire.org.uk



Figure 32 Code snippet of the Socket Listener

Moreover, as the data sent through the network is raw and unprocessed there needs to be a component to decode the data into a useable format which in this case is the quaternions representing the movement of the Wiimotes. Afterwards and for reasons for simplicity of presentation we also transform the quaternions into Euler angles. The Data Process Unit performs this task and the code snippet is offered by Figure 33 below:

```
Data
  Process
     Unit
 70
            void dataProcess ()
 71
     白
            {
 72
                //get data from socket
 73
                data = new byte[256];
 74
                    int receive = newSock.ReceiveFrom(data, ref remote);
 75
 76
                    string rawdata = System.Text.Encoding.ASCII.GetString(data);
 77
                    string[] result;
 78
                    result = Regex.Split(rawdata, @",\s");
 79
 80
                             // get separate values of the quaternions (w1,x1,y1,z1) and (w2,x2,y2,z2)
 81
                             Single w1 = System.Convert.ToSingle(result[0]);
                             Single x1 = System.Convert.ToSingle(result[1]);
 82
 83
                             Single y1 = System.Convert.ToSingle(result[2]);
 84
                             Single z1 = System.Convert.ToSingle(result[3]);
 85
                             Single w2 = System.Convert.ToSingle(result[4]);
 86
                             Single x2 = System.Convert.ToSingle(result[5]);
 87
                             Single y2 = System.Convert.ToSingle(result[6]);
 88
                             Single z2 = System.Convert.ToSingle(result[7]);
 89
 90
 91
                    //Assign the w.x.y.z values to the corresponding quaternions and inverse.
                    Quaternion quat_1 = new Quaternion(x1, z1, y1, w1);
 92
                    //z comes before y as Unity uses Y-up world orientation system
 93
 94
                    Quaternion guat 2 = new Quaternion (x2, z2, y2, w2);
 95
                    Quaternion quat_1 inv = new Quaternion();
 96
                    Quaternion quat 2 inv = new Quaternion();
 97
                    quat 1 inv = Quaternion.Inverse(quat 1);
 98
                    quat_2_inv = Quaternion.Inverse(quat_2);
 99
                    //transform to Euler angles
101
                    euler 1 = new Vector3();
102
                    euler_2 = new Vector3();
103
                    euler_1 = quat_1_inv.eulerAngles;
104
                    euler_2 = quat_2_inv.eulerAngles;
106
                    //store culer values to o_1 and o_2 vectors so that they are accesible as separate x.y.z (pitch, roll,yaw)
107
                    Vector3 o_1= new Vector3();
                    Vector3 o_2= new Vector3();
105
109
                    o_1[0] = euler_1.x; //pitch
110
                    o 1[1] = euler 1.z; //yaw
                    o_1[2] = euler_1.y; //roll
112
113
                    o_2[0] = euler_2.x; //pitch
114
115
                    o_2[1] = euler_2.z; //yaw
                    o 2[2] = euler 2.y: //roll
116
117
118
            ł
```

Figure 33 Code snippet of Data Process Unit

The game engine utilised for the development of the games is the free version of Unity3D by Unity Technologies, version 4 as described above. This environment enables very easy import and export of 3D assets either form third party or the integrated Unity store which

is very helpful especially for mini-game development such as the ones presented in this Thesis.

5.3.1 Pilot games for Stroke rehabilitation

The design and development of the games for post-stroke rehabilitation of the upper limbs follows the summary of the reviewed literature presented above in section 4.1.2.1. Three mini-games have been developed and are presented below:

5.3.1.1 The Virtual Teacher

The first game is the Virtual Teacher, a term coined by Holden and Dyar [173]. As seen in Figures 34 and 35 below, the game scene is basic and consists of a virtual avatar of neutral appearance. The avatar has an extra limb, depicted in red colour, which is the limb driven by the prescribed motion of the clinician. The platform enables a pre-recorded motion to be mapped on the red limb, which is the "Teacher" arm, and the motion can be repeated for as many times as the clinician finds appropriate. Then, the patient has to follow the movement dictated by the Virtual Teacher and for as many repetitions. Through this platform, the clinician can prescribe virtually any range of exercises for upper-limbs and thus the potential of this gamified rehabilitation experience is endless in terms of exercise regime. Moreover, by superimposing the teacher's (therapist in our case) arm movement on the patient's arm movement and instructing the patient to follow the teacher's arm trajectory it is more likely that the patient can perform the action correctly.



Figure 34 Sketch of the Virtual Teacher game



Figure 35 The Virtual Teacher game environment

The capture of the movement is realised via two Wiimotes, each of them attached to the patient's upper and lower arm correspondingly. This way the full kinematic tracking of the

upper limb is enabled in real-time. Due to that fact, the engagement of the patient with the avatar is facilitated through the instant visual stimuli and the natural and seamless representation of the patient's movement into the VR environment.

The components of the game consist of the Interface, the Motion Data Mapper, which includes the mapping of the Clinician Prescribed Exercise Data and the Patient Motion Data, and the Virtual Avatar and Virtual Teacher Limb, which are included under the Visualisation and User Interfacing component. The data flow diagram below (Figure 36) illustrates the dependencies of the various components in the system of the game.



Figure 36 Data flow diagram of the VR Teacher game

It shows that the Wiimote motion data, which is transferred online, is fused into the game environment and then processed to be transformed into the appropriate format through the Interface component. At the same time, the prescribed exercise data is transmitted via the same path. The first set of data is then mapped on the virtual avatar and the second set on the virtual teacher limb in the Motion Mapper component. In order to map the motion data on the 3D objects, the quaternion values of each component has to be mapped on the transformation values of the 3D objects. The code snippet below (Figure 37), illustrates this mapping:

170	box1 = GameObject.Find ("Shoulder3"); //find and select the "Shoulder3" object
171	box2 = GameObject.Find("Forearm3"); //find and select the "Forearm3" object
172	box1.transform.rotation = quat_2_inv; //map the motion data.
173	<pre>box2.transform.rotation = quat_1_inv;</pre>

Figure 37 Code snippet of the data mapping

Furthermore, the game is designed in a way to facilitate the mapping of either limb and moreover in any visualisation setup. The system can be easily adjusted so it can represent the movement in a mirror mode or in a conventional respective manner (left-to-left, right-to-right mapping). The use of an angled camera was included as a game feature to visually aid the orientation of the patient's movement in 3D space.

5.3.1.2 Hitting the Spheres

The second game developed for the purposes of this project is the "Hitting the Sphere" game. This game uses again the same avatar, this time without the Virtual Teacher component. The way that the backbone VR environment is designed, it is very easy to expand so additional applications and especially games can be integrated into the exercise library by including / excluding add-on components. The "Hitting the Sphere" game is an example of that expansion. Figure 38 that follows, depicts the components of this game in a data flow diagram:



Figure 38 Data flow diagram of Hitting the Spheres game

The arm movement of the patient is again mapped on the corresponding limb of the avatar via the Motion Mapper component, depending on the case of virtual mirror or normal mapping. In this game the user moves his/her arm in order to interact with (e.g. move) the avatar's arm and hit the 3D spheres within the VR space and a predefined period of time defined by the Scoring and Timing component. Game parameters, such as the time that a new sphere randomly appears or how far from the avatar's body should it be located, as well as the arm range of motion that the avatar supports are fully adaptable dynamically, in order to cater for each patient's specific needs. These events are handled by the Game Event Manager. The game records the ROM and score, to provide the therapist with feedback but also to report back to the patient at the end of each session with the simple but yet encouraging and positive feedback messages, such as "Well done a new record was made", "Good effort", etc. in the Visualisation and User Interfacing component. The spheres can also be adjusted, as an extra expansion feature, to appear on the screen in a prescribed mode

and the distance between two consecutive spheres represents the ROM of the targeted exercise. This way the clinician can adjust the difficulty of the game manually by presetting the distances, as well as the orientations of the spheres in order to target a specific exercise, or even use the algorithm of section 4.2.3 to automatically adjust the difficulty. Figures 39 and 40 illustrate the "Hitting the Spheres" design sketches and game environment correspondingly. Again, the use of an angled camera was included as a game feature to visually aid the orientation of the patient's movement in 3D space.



Figure 39 Sketch of the Hitting the Spheres game



Figure 40 The Hitting the Spheres game environment

5.3.1.3 Air Hockey Game

The third game developed for the purposes of this project was an Air Hockey game. The patient controls a virtual puck, again through two Wiimotes attached on the upper and lower arms correspondingly, and the target of the game is obviously to score more goals than you concede. The data flow diagram of Figure 41 demonstrates the basic components used for the deployment of this game. The components are very similar to "Hitting the Spheres" game.



Figure 41 Data flow diagram of the Air Hockey game

The game itself is an open source available game¹⁵ made in Blender3D game engine. Therefore, the components of the Game Event Manager and Scoring and Timing, as well as the 3D assets were available and the author used the reusable components of the project to reverse engineer this game to match the requirements of the project. This fact shows that the adopted approach is inter-platform and can be adapted to any kind of game engine and environment, given that the source code is available. Figure 41 illustrates the components of the game and with overlaid blur are the components that belong to the original air hockey game. The opponent's puck is controlled by the local PC by the Computer Player component. The figure below (Figure 42) depicts the game environment of the Air Hockey game. The Air Hockey game is developed in order to assess the preference of the players

¹⁵ <u>http://davidwehr.webs.com/apps/blog/show/4411983-source-available-and-networking</u> (last accessed Dec 2011)

between VR exercises as gamified experiences (games 4.3.1.1 and 4.3.1.2) and conventional game concepts.



Figure 42 The Air Hockey game environment

5.3.2 Pilot Games for Parkinson's Rehabilitation

The system architecture presented above in Figure 23 facilitates the accuracy and calibration. Following is the design of the actual games. The design of such games is challenging in its own right as the requirements of the players are rather complex and diverse. Furthermore, the designer has to foster performance in PD physiotherapy and align with the targets of the physical rehabilitation to optimise the outputs of the physiotherapy. Hence, the design of the game has to allocate movement and agility that will coordinate physical exercises and not random repetitive movements for the sake of keeping active. Therefore, our approach proposes the translation of the physical rehabilitation regime to game concepts, where that is applicable. We initiate the design from the suggested rehabilitation regime for PD of PD Society of UK [174], and try to conceive game concepts for each of the individual exercises. This task can be proven to be very challenging, whereas coming up with ideas for random repetitive movements, which within a game environment will serve as a physical activity, is boundless. In addition, the design elements mentioned

in section 4.1.2.1 have to be incorporated as they were found to be effective in their own respect.

The application of the design principles mentioned above, were derived from the development of two pilot games. Two exercises were chosen from Keeping Moving [174], a guide for exercise and Parkinson's Disease by PD Society of UK, and the author's conceived two ideas to translate the said exercises into gameplay.

5.3.2.1 The Rowing game

The game's aim is for the player to control a two-paddled row boat (bilateral movement) with the challenge of reaching a specific point at a given time. The targeted exercise is shown in Figure 43.



Figure 43 Arm reaching exercise [174]

The game mechanics focus on improving users' speed, ROM, bi-lateral movement and reducing rigidity. The game is designed to collect specific biofeedback (to be measured in the game and displayed in the Clinician Platform), such as the ROM of upper limbs in Euler degrees, speed in m/sec, spontaneity of movement etc.

The game is designed to enable the user to use both arms simultaneously, in a reaching style exercise from a seated position. Visual cues appear on the screen to direct player's

movement. Audio is used to set the rowing tempo. The figures (Figures 44, 45, 46) below illustrate the sketch of the first and second version of the rowing game correspondingly.



Figure 44 Sketch of the Rowing game



Figure 45 The Rowing game environment version 0.1



Figure 46 The Rowing game environment version 0.2

The development of this game took place in Unity3D game engine and followed very similar approach as the games above. The data flow diagram that follows (Figure 47)



demonstrates the components of the game as well as the data flow among them.

Figure 47 Data flow diagram of the Rowing game

Some basic differences with the previous games are that in this case there is the element of Motion Data Calibrator / Adaptor and the component of Cueing (Audio and Visual) which goes in the Visualisation, Cueing and User Interfacing component. The calibration and adaptation of this game is in terms of the ROM of the oars. When the patient performs the targeted exercise movement, as shown in Figure 43 but from a sitting position, the registered ROM is not close to the ROM of the 3D object. Thus, there is the need of calibrating and furthermore dynamically adapting the ROM in order to give the patient the incentive to perform better. The component of Motion Data Calibrator / Adaptor is described in detail in section 4.2.3.

The component of Visualisation, Cueing and User Interfacing is responsible for visualising and giving the appropriate feedback to the user and furthermore the audio and visual cues required for the PD cohort of patients. In addition to the visual cues there are auditory cues, which help the patient to develop a spontaneous movement and increase the speed as the game progresses. This audio cue is a simple tempo, which increases as the speed of the boat / movement of the patient increases. The snippet below (Figure 48) shows the code of the sound tempo that is increasing according to the speed of the boat.

```
using UnityEngine;
 1 -
 2
    using System.Collections;
 3
 4 - public class SoundPitch : MonoBehaviour {
 5
         public int startingPitch = 3;
 6
 7
         public int timeToDecrease = 5;
 8
         public Vector3 pos;
 9
         public Vector3 velocity;
10
11 -
         void Start() {
12
             audio.pitch = startingPitch;
13
         }
14
15 🗕
         void Update() {
16
             if (audio.pitch > 0)
17
              calculateVelocity();
18
         }
19
20 -
         void calculateVelocity () {
21
             pos = transform.InverseTransformPoint
22
                 (GameObject.Find("cubeMove").transform.position);
23
             velocity = GameObject.Find("cubeMove")
24
                 .rigidbody.GetPointVelocity(pos);
25
             //print (velocity.magnitude);
26
             audio.pitch = 1+velocity.magnitude / 20;
27
28
         }
29
     }
```

Figure 48 Code snippet of the audio cue

The other components are the same apart from the Game Event Manager which is, by default, specific for every game. In this game, the Game Event Manager manages the games mechanics and in particular the triggering of events such as "the oar is going forward" or "oar is going backward" in order to trigger the corresponding visual cue i.e. "Back" and the back arrow as seen on Figure 46 above. The triggers of such events are object colliders, which mean that an event is triggered when two objects collide. For example the boat is moving forward when the oar is touching a pair of boxes (that do not appear in the scene) and in specific order.



Figure 49 Rowing game – the development environment

The figure above (Figure 49) illustrates the development environment (Unity3D) and shows the aforementioned box helpers. The patient performs the rowing and once the oar touches the boxes in the specific order (to mimic the actual movement of the rowing and increase augmentation) the boat will be pushed forward with a specific amount of force. The code for these events is depicted in the snippet code below (Figure 50):

```
public class Collision : MonoBehaviour {
 4 -
 5
         public GameObject cube;
6
7
         public GameObject cube_anim;
8
         public GameObject cubeTransform;
9
         string[] boxnames = new string[2];
10
         Vector3 pos;
         void OnTriggerEnter(Collider other) {
11 🖻
12
13
14
             cube_anim = GameObject.Find("Terrain");
15
             cubeTransform = GameObject.Find("CubeTransform");
16
17
                 if (other.collider.name == "Cube1")
18
                     {
19
                         boxnames[0] = other.collider.name;
20
                     3
21
22
                 else if (other.collider.name == "Cube2")
23
                              {
24
                                  boxnames[1] = other.collider.name;
25
                              }
```

```
28
             if (boxnames[0]== "Cube1")
29
             {
30
31
                 if (boxnames[1]== "Cube2") {
32
                     GameObject.Find("Cube_anim").animation.Play("MoveTerrain");
                     cubeTransform.transform.Translate(0,0,5);
33
34
                     print ("move by 50 metres");
35
36
                     }
37
                 boxnames[0] = null;
38
                 boxnames[1] = null;
39
             }
40
41
42 -
             } // this nesting allows the detection of the order of tiggers.
43
               //First the back (cube1) then the front(cube2)
44
               // and then the "move by 50 meters" happens.
45
         }
```

Figure 50 Code snippet of the events manager

5.3.2.2 The Steam Mini-golf game

The game's aim is for the player to rotate the valve in several repetitions in order to release steam to push the ball into the hole within a given time. The targeted exercise is shown in Figure 51. Visual cues appear on the screen to direct player's movement. Audio is used to set the rotation tempo.



Figure 51 Rolling a ball exercise [174]

The game mechanics focus on improving users' speed, rigidity, ROM, bi-lateral movement. The game is designed to collect specific biofeedback (to be measured in the game and displayed in the Clinician Platform as a future development) such as ROM of upper limbs in Euler degrees, speed in m/s, spontaneity of movement etc. The figures (figures 52, 53, 54) below depict the first and second version of the Steam Mini-golf game.



Figure 52 Sketch of Steam mini-golf game



Figure 53 The Steam Mini-golf game version 0.1



Figure 54 The Steam Mini-golf game version 0.2

The two cases presented above are two physiotherapy exercises, prescribed by the Parkinson's Disease Society, translated into mini-games that will engage the patient enabling home-base rehabilitation to become more motivating and immersive. On top of that, the clinician will be provided with real-time biofeedback realising the potential to regulate the level of difficulty of the performed action as well as to record performance data for future reference and further analysis through the proposed Clinician Platform. Similar to the game presented above, the Steam Mini-golf game follows the generic game development data flow and the diagram that follows (Figure 55) further demonstrates the data flow within the game:



Figure 55 Data flow diagram of the Steam Mini-golf game

The main difference lies within the component to manage the game mechanics, which is the Game Event Manager. In this game the 3D object manipulated by the motion data of the patient is a set of virtual hands each for a corresponding limb of the patient.



Figure 56 Steam Mini-golf game – the development environment

The figure above (Figure 56) shows the trigger boxes that do not appear in the actual game. When the player touches those boxes in a particular order (mimicking the direction of undoing the valve), the red valve starts moving towards that direction and the sound of the valve starts playing. The green arrows and the target objects are the visual cues to assist the patient with the movement requirements. There is a gauge GUI element that informs the player of the pressure release levels and once these reach a critical level, the steam is released and pushes the ball with a force towards the hole. The corresponding code is illustrated by the Figure 57 below:

```
void OnTriggerEnter(Collider other) {
20 -
21
            ball = GameObject.Find("ball");
22
             steam = GameObject.Find("steam");
23
24
25
             cubeTransform = GameObject.Find("vanaMove");
26
27
                 if (other.collider.name == "CubeRight")
28
                     {
29
30
                         boxnames[0] = other.collider.name;
31
                     }
32
33
         if (boxnames[0]== "CubeRight")
34
             Ł
35
                 print (boxnames[0]);
36
                 print (boxnames[1]);
37
38
                 if (boxnames[1]== "CubeLeft") {
39
40
                     cubeTransform.rigidbody.AddTorque(0,0,-40);
41
                     if (cubeTransform.audio.isPlaying == false)
42
                     cubeTransform.audio.Play();
43
                     strokes = strokes+1;
44
45
46
                     if (strokes>=9)
47
                     Ł
48
                         ParticleAnimator steamAnim = steam.
49
                              GetComponent<ParticleAnimator>();
50
                         steamAnim.force = new Vector3(10 , 0 , 10);
51
                         steamAnim.sizeGrow = 3;
52
                         ball.rigidbody.AddForce(100,0,100);
53
                         if (steam.audio.isPlaying == false) {
54
                         steam.audio.Play();
55
                         }
56
57
                     }
58
                     print ("Number of Strokes =" + strokes);
59 -
                     //cube anim.transform.position =
60
                     //cubeTransform.transform.position;
61
                     3
```



Figure 57 Code snippet of the Game Events manager

It is clear that there are some game mechanics that can be adjusted by the designer, through the manipulation of certain values in the code, in order to upgrade / downgrade the difficulty level apart from the calibration and adaptation algorithm. These values can be remotely adjusted by the clinician and it is a feature that will be included in the future development of the clinician tool.

Summary

The application of SGs for Motor Rehabilitation, which was examined in two research projects, is presented in this chapter. Firstly, the motivation behind this research is explained through the presentation of the problem under investigation. The findings of the literature review are summarised and form the requirements of the development. Moreover, the overall framework is presented with an emphasis on the infrastructure of the proposed system and the technical specifications of the bespoke MoCap system developed as well as the adaptation algorithm. The accuracy of the proposed system is examined and the experimental methodology as well as the results and discussion are presented. Finally, the implementation of the SGs for Motor Rehabilitation is offered and presented in detail.

CHAPTER VI: SERIOUS GAMES FOR MOTOR REHABILITATION

Evaluation, Results and Discussion

Abstract:

This chapter constitutes the evaluation and results section of the Serious Games for Motor Rehabilitation developed for the scopes of this Thesis. It focuses on the research methodologies employed to evaluate the implemented games presented in the previous chapter. After the description of the corresponding evaluation methods, the results are presented. Following the results, there is a discussion of results and the themes and design guidelines that emerge from that discussion. The structure of the chapter is depicted on Diagram 6 below:



Diagram 6 Chapter Structure

6.1 Serious Games for Motor Rehabilitation

6.1.2 Stroke Rehabilitation Serious Games

The evaluation of the stroke intervention developed for the purposes of ReWiiRe project was performed in an iterative manner. The first round of pilot testing was performed with the aid and invaluable input of three stroke survivors who assisted in the development of the prototype games. Then the improved versions of these games were tested by an expertpatient in a longer study that assessed the outcome measures for baseline ability and outcome.

Following Brunel University ethics approval, three stroke survivors were recruited to the project team and assisted with system evaluation and further development. The following table (Table 8), provides the demographic characteristics of the patient-experts and Table 9 the characteristics of the Single-Case Study expert.

Part. num	Gender	Age	Length of disease	mRS score	Motor Characteristics	Played the Wii before?
1	Male	33	18 months	3	moderate disability, with moderate gross movement at the shoulder and elbow, flickers at the wrist and hand (no grasp).	Yes
2	Female	52	16 months	2	slight disability with weak and uncoordinated movement throughout the arm and hand.	Yes
3	Female	57	14 months	4	moderately severe disability with minimal movement at the shoulder and elbow and some non-functional movement in the hand.	No

Table 8 Demographic information of the patient experts

Table 9 Demographic information of the single-case study expert

Part. num	Gender	Age	Length of disease	mRS score	Motor Characteristics	Played the Wii before?
Single case study	female	31	11 years	2	left sided hemiplegia	Yes

Participants took part in three trials using the Wii-based stroke intervention in a university therapy room (see Figures 58, 59).



Figure 58 The single-case expert playing the Virtual Teacher game

All participants were given time to familiarise themselves with the system. The field test, including familiarisation with the protocol, lasted 50 minutes. The trials are summarised below:

- i. Teacher animation trial: each participant performed three, pre-recorded upper-limb exercises in which (s)he had to follow a 3D avatar projected on a screen. Three movement patterns, involving the shoulder, elbow and forearm, were selected on the basis of functional use and reported difficulty in movement performance. Each movement pattern was performed by participants for 3 sets of 5 repetitions. The pre-recorded exercises were different for each patient in terms of speed, ROM and movement performed dependent on each participant's specific impairments.
- Ball hitting game trial: involved the use and display of the 3D avatar on the projector screen where participants tried to "hit" randomly appearing balls, (high, middle, low, central, left or right of the screen avatar). Each ball remained on the screen for 5 seconds, or less if successfully hit. The time taken to hit (or miss) 25 balls was recorded. Two sets of 25 were completed.
- iii. Air hockey game trial: centred on playing an air hockey game for 5 minutes. This involved moving a "puck" forward, backward and side to side through combined movements of the elbow and shoulder to score goals and defend their own "net" against the computer opponent. The ROM was changed to accommodate the functional motion ability of each participant.

All sessions were captured via video and observation notes were made during the trials. After completion of field testing, participants shared their comments, insights and experience about using the customised Wii intervention with the rest of the research team.

6.1.2.1 Results and Discussion

A number of themes emerged from post intervention debriefs and observation notes, namely: participant connection with the avatar, normal versus mirror view, VR environment and personalisation.

i. Participant Connection with the Avatar

Participants reported the importance of being able to see their movement mapped onto a virtual character (avatar), as it helped them in getting a direct visual feedback of the action performed at the time. Participants also reported feeling connected with the avatar:

"I actually like when I am doing it, it (referring to the avatar) feels like that's me" (Participant 1)

This was more evident in the teacher animation and hitting the ball trial. In the former, participants indicated the usefulness in being able to see the physiotherapist's pre-recorded exercise motion on the screen whilst they were trying to match it. They stressed the importance of feedback and guidance and reported that this would enable the session to take place at home instead of the clinic:

"It would be like having the physio at home with me guiding me all the way through what [the exercises] I have to do plus I can see and compare how well I am doing" (Participant 2)

It was observed, particularly in the case of participants 1 and 2, that their ROM improved marginally (i.e. they were able to extend their arms slightly further) when conducting the exercise whilst looking at the avatar compared to looking at their arm alone. In addition to this, participants highly valued the responsiveness of the system, particularly in terms of the visual feedback provided through the depiction of the arm movement on the VR environment:

"It is great! I can see my arm movement appear instantly on screen and can tell how well and far I have gone" (Participant 3)

Participants were asked to perform the exercises in the teacher animation trial while seated and standing up; the avatar was in both cases in a standing posture. Participant 2 did not mind this but participants 1 and 3 found it less helpful:

"It is easier to do exercises when he (referring to the avatar) is standing, it feels more like *me*" (Participant 1)

Participant responses indicate a direct link between patient motivation and personal connection with the avatar, especially when this is combined with instant visual representation and feedback of their motion on the screen. This was true with regards to the position and posture of the avatar with participants suggesting that if they are seated, so should the avatar. This was not the case however in terms of the avatar's personal physical characteristics with all three subjects reporting this as irrelevant.

ii. Normal versus Mirror View

Another element investigated, in both the teacher animation and hitting ball trial, was the use of normal and mirror view (similar to watching oneself in a mirror). All participants reported preference for the mirror view, as they found it more natural and they were accustomed to using a mirror when conducting exercises either at home or in the gym. It was also observed that participants performed better in terms of response time and scored higher (in the case of the hitting ball trial) when using the mirror view:

"It feels (referring to the mirror view) a more natural way of seeing things" (Participant 3)

It was not expected at the start of the trials that this would have such an impact on participant interaction with the VR environment and performance. It is therefore suggested that any VR stroke rehabilitation environment should support a mirror view, as it is perceived by stroke patients as a more natural way of viewing and understanding their movement.

iii. VR Environment Personalisation

According to participants, a key feature of the trialed activities was the ability to personalise various aspects of the VR environment. In particular, the exercise duration, ROM, gameplay / activity speed and avatar viewing angle were regarded as the most important personalisation features. Participants commented that when interacting with the teacher animation, they found it a great benefit to be able to select from a range of prescribed exercises. They reported that being able to have control over the ROM in the teacher animation, hitting ball and air hockey trials is very important in terms of accessibility, as it enables them to fit the activities to suit their needs:

"I can now play games and do things not possible for me before" (Participant 2)

"I used to play Wii bowling and golf before. This is the first time since then I have played games with the Wii again" (Participant 1)

In addition to this, participants indicated that being able to adjust the speed in the hitting ball and air hockey trial made the tasks more manageable and enjoyable:

"I found it (referring to the hitting ball trial) hard at the beginning, as it was fast for me. After changing the speed it become easier and I could follow it" (Participant 3)

This was found to be a key factor for the adoption of the trialed activities, as different stroke survivors have very different functional abilities and response times. For instance participant 3 had the slower reaction time and was slower compared to the others. Feedback from this stage resulted in the use of a mirror view of the avatar being adopted.



Figure 59 Expert patient playing the Virtual Teacher game

6.1.3 Parkinson's Disease Rehabilitation Serious Games

The framework, developed and presented in Chapter V, facilitates the flexibility of the system components to apply on a variety of motor deficits and rehabilitation requirements. The system utilised for stroke rehabilitation is repurposed and a series of new mini-games are developed to target PD patients as presented in Chapter V (Section 5.3.2). The evaluation of these games was conducted in two rounds of pilot testing. Five participants diagnosed with idiopathic PD (2 women, 3 men; mean age 68.25 ± 3.5 years; disease duration, 5.4 ± 1.5 years) were recruited. All participants were on medication throughout the course of the study. Prior to commencing the study, the severity of PD was assessed using both the Movement Disorder Society's revised UPDRS Scale (MDS-UPDRS [177]), and the modified Hoehn & Yahr rating scale ([178], see Table 10). All participants had an MMSE (Mini Mental State Examination, [179]) score above 24, indicating no serious cognitive deficits. Participants also fulfilled the following inclusion criteria: normal or corrected-to-normal vision and hearing, no history of neurological conditions other than PD, and no musculoskeletal disorders affecting arm movement. Ethical clearance was obtained from the local psychological ethics committee, and all participants gave informed consent. The number of participants is enough to collect insights to drive the design of the games as according to Nielsen [180][181] the first five users will reveal about 85% of the problems. In a prototype design like ours, this percentage is acceptable due to the duration of the project and funding. Furthermore the iterative nature (two pilots) of our testing reduces the risk of unrevealed problem. The pilots took place in a research lab and each session lasted an hour. This lab is the Movement Innovation Lab at School of Psychology, Queen's University in Belfast. Participant information was collected prior to the study. The participants had to provide us with their written consent about the study and were asked to read the participation information and sign where applicable. The Movement and Innovation Lab at Queen Belfast University had received Ethics Approval for conducting the pilot studies from its University Research Ethics Committee. The figures below (Figures 60, 61) illustrate the participants playing the games for PD.

Table 10 Participants	characteristics
------------------------------	-----------------

Variable ¹⁶	Mean ± SD
H&Y rating scale (Stage 0-4)	$1.8 \pm 0.84 \ (Mode = 2)$
Total MDSUPDRS	56.60 ± 19.36
MDSUPDRS-section I	9.40 ± 3.05
MDSUPDRS- section II	10.02 ± 4.91
MDSUPDRS- section III	33.80 ± 10.13
MDSUPDRS- section IV	4.40 ± 3.84



Figure 60 Expert patient playing the Rowing game

¹⁶ SD: Standard Deviation; H&Y: Hoehn and Yahr Rating Scale; MDSUPDRS: Movement Disorder Society's Unified Parkinson's disease Rating Scale; MDSUPDRS-I: Non-Motor Aspects of Experiences of Daily Living (max score: 52); MDSUPDRS-II: Motor Aspects of Experiences of Daily Living (max score: 52); MDSUPDRS-III: Functional Motor Examination (max score: 72); UPDRS-IV: Motor Complications (max score 24). Higher scores indicate more severe symptoms and progression.



Figure 61 Expert patient playing the Steam mini-golf game

At the first round of pilot testing, participants were shown and given the opportunity to familiarise with the two games and then asked to engage with these. In addition, the authors were given the opportunity to address any issues with the motion capture technology and get feedback on the use of the Wiimotes as MoCap devices from the patients. Each session was recorded using video cameras. Also, the motion data (including Wiimote motion capture) of each game session was recorded by the system for later analysis. This data is stored in quaternion form and for each of the degrees of freedom (yaw, roll, pitch). A questionnaire (Appendix C) was distributed after each session and a focus group took place with all study participants. The participants played each of the two games for at least five rounds with small breaks between rounds to prevent muscle fatigue. Then, they were given some time to rest and after that answer the questionnaires.

After the completion of the first pilot, all data (notes, observation video and post-study questionnaires) was analysed and the user feedback we got from the first round of pilot testing was taken into account to amend accordingly. A second round of pilot testing was arranged two months after the first session with the same cohort of participants. The beta

versions of the games, amended according to the participants' feedback, were presented and tested again by the same persons. Afterwards, the participants had to fill in the same questionnaire as in the first pilot and give us their feedback on the improved game versions. At the end of the sessions, a focus group intervened with the participants to gather as much feedback as possible through various research methods.

6.1.3.1 Results and Discussion

The data gathered from the various stages of the study, presented above, is organised, analysed and summarised in a qualitative manner. Observation notes, video recordings of the sessions and the post-study questionnaires are evaluated and conclusions are driven by that data. The table below (Table 11) summarises the quotes captured from participants.

Participant ID	Quote ID	Quote	
	P101	"sound feedback is most welcome"	
	P1Q2	"it would be useful if feedback is compared with previous sessions to show me how I am doing"	
Participant 1	P1Q3	"You can make it more challenging by having to navigate and avoid obstacles or follow a track, like in canoeing"	
	P1Q4	"it is hard relating the game [indicating the steam golf game] to a real event. Perhaps you could turn it to a sailing boat steering game"	
	P2Q1	"It feels good, it feels as the muscles here and here [pointing at the forearms on both arms] have really tighten from what we've been doingit feels as if my arms are being used"	
Participant 2	P2Q2	"This competitiveness is startling me but also keep me going"	
	P2Q3	"It seems more challenging, the pace is faster than last time" [this is due to the audio tempo incorporation, giving them the impression of a faster game pace]	
	P3Q1	"I was rowing when I was younger so I enjoy this"	
	P3Q2	"it's a very useful exercise"	
	P3Q3	"How did I do? Did I do better than [participant 4]?"	
	P3Q4	"It is much better than last time"	
Participant 3	P3Q5	"It seems to be faster than last time. The pace seems faster, I thought I was going faster" [this is due to the audio tempo incorporation, giving them the impression of a faster game pace]	
	P3Q6	"Competing with the computer would be good"	
	P3Q7	"It is good that the game is not too long so that you can rest"	
	P3Q8	"I like the movementit feels like driving. It makes more sense making in into a driving game"	

Table 11 Participants Quotes

	 P3Q9 "useful to get feedback on my performance and whether it changeduseful to know how I am over time" P3Q10 "audio_that was useful_it gives you incentive 	
	13Q10	the extra mile"
	P4Q1	"It would be good be able to play against another person"
	P4Q2	"This game [indicating the steam golf game] would work better as a driving game, since you have to move your arms like when driving"
Participant 4	P4Q3	"Would you be able to make it into a hand coordination game? That would be most useful."
	P4Q4	"I have not been able to use the Wii so this game [indicating the rowing game] is very good for upper body exercise"
	P4Q5	"It is a fairly challenging exercise [indicating the steam golf game]"
Participant 5	P5Q1	"it is interesting because I do not have to hold or click anything"
т апистрані 5	P5Q2	"it's much more demanding exercise" [referring to the steam golf game]

The games were tested individually and the results are presented for each of the games.

i. Game 1

The feedback and suggestions we obtained for the first game, namely the rowing game, can be summarised in the following points (Table 12):

Feedback	Suggestions
 Fun to play 	 Introduction of score/performance progress history
Clear Instructions	 Audio feedback on how to improve exercise technique
 Intensive exercise with a very good range of movement for upper body 	 Improve visual aesthetics and add more game graphics
 High motivation 	
 Positive comments on graphics / visual / audio cues 	

Table 12 Summarised feedback and suggestions for the Rowing game

The majority of participants enjoyed playing the rowing game. Participants overall commented that the game was fun to play and that the exercise although intensive, it contained a good range of exercise for the upper body:
"It feels good, it feels as the muscles here and here [pointing at the forearms on both arms] have really tighten from what we've been doing...it feels as if my arms are being used" (P2Q1)

"I have not been able to use the Wii so this game [indicating the rowing game] is very good for upper body exercise" (P4Q4)

The study participants were clearly engaged with the game and the agility required for the gameplay seemed to be fitting to their motor capacities.

"I like the movement...it feels like driving. It makes more sense making in into a driving game" (P3Q8)

"it is interesting because I do not have to hold or click anything" (P5Q1)

It was felt that playing the game increased the incentive and motivation for exercising and despite being intensive at times, participants were motivated to complete the game and each time to improve their score. Interesting dynamics were observed amongst our participants as some expressed enjoyment in trying to improve their personal best whereas others were more interested in competing against each other:

"How did I do? Did I do better than participant 4?" (P3Q3)

"It would be good to be able to play against another person" (P4Q1)

"Competing with the computer would be good" (P3Q6)

The targets of the game were clear and they found the exercise intensive but appropriate for upper limb training. The users also found the audio and visual cues very helpful and commented positively on the graphics and the audio feedback.

"It is much better than last time" (P3Q4)

The audio with the beat tempo was found very useful, as it was observed when looking at the captured videos and produced performance scores/times. Generally participants rowed faster in the game when the audio tempo was increased as the game progressed:

133

"audio...that was useful, it gives you incentive to go the extra mile" (P3Q10)

"It seems more challenging, the pace is faster than last time" [this is due to the audio tempo incorporation, giving them the impression of a faster game pace] (P2Q3)

"sound feedback is most welcome" (P1Q1)

Another interesting feedback came from one of our study participants who used to row prior his PD diagnosis and found this game as a useful tool to reminiscing and being able to follow his hobby:

"I was rowing when I was younger so I enjoy this" (P3Q1)

With respect to hobbies another participant provided the team with invaluable ideas on how the existing rowing game could be modified to create new games, which can cater for other personal hobbies whilst keeping the same rate and level of activity, such as a gardening game.

Also a number of improvements were suggested. The key one was the inclusion of a score/performance progress history, which allows participants to compare their score with previous game sessions and see how much they have progressed:

"it would be useful if feedback is compared with previous sessions to show me how I am doing" (P1Q2)

Another suggestion was to provide the player with feedback (possibly audio feedback) on how to improve the technique (e.g. what needs to be done to score higher). Other improvements included the use of more animations, the incorporation of a 3D character graphic on the boat and being able to play against the computer. An interesting note about the overall game design feedback received is that we would normally expect such feedback and suggestions (e.g. on graphics improvement) by a younger population. It was therefore very positive and at the same time unexpectedly welcome to have such responses. Indicating that older users have an equally fluent view when it comes to game play and experience.

134

ii. Game 2

The feedback and suggestions we obtained for the second game (the steam mini-golf game) can be summarised in the following points (Table 13):

Feedback	Suggestions
 Much more challenging exercise 	 Improve visual feedback / cues
 Not very clear instructions 	 Clockwise instead of anticlockwise movement was found more appropriate
 More challenging to accomplish the task 	 More clear instructions

Table 13 Summarised feedback and suggestions for the Steam mini-golf game

This second game, the steam mini-golf game, has proven to be more challenging than and not as popular as the rowing game:

"it's much more demanding exercise" [referring to the steam golf game] (P5Q2)

"It is a fairly challenging exercise" [indicating the steam golf game] (P4Q5)

Overall it was found that underlying game exercise was much more challenging when compared to the other game as it required arm movement across three different planes (coronal, transverse and sagittal plane). It was our intention to investigate a game with a more complex set of motor skills required in the gameplay in order to examine the limitations of translating the physiotherapy exercises into games. It seems that the patients were confused by the game concept. However once the concept was explained they were motivated to accomplish the given task and stand up to the challenge. On the one hand the increased challenge provided participants with more incentive to win the game but on the other hand it was found very difficult and tiring. The game graphics were found to be good and the clockwise movement of the arms required in the game was found to be more appropriate when compared to the anti-clockwise movement used in the first pilot version of the game. It was suggested that the mini-golf concept employed for the game design was part of the issue as it was not directly replicable and transferable to the real environment: "it is hard relating the game [indicating the steam golf game] to a real event. Perhaps you could turn it to a sailing boat steering game" (P1Q4)

"This game [indicating the steam golf game] *would work better as a driving game, since you have to move your arms like when driving"* (P4Q2)

Participants suggested that it would be better instead to have a boat or car racing game, as it would enable a clear association with an already known activity. Our discussion with participants after the pilots and their feedback and suggestions clearly indicated to us that games, which are aligned to past hobbies and activities have a greater chance of success and acceptance by people with Parkinson's.

6.2.3.1.1 Design Guidelines

This section presents a set of design guidelines and best practices. These emanate from the review of the available literature and the feedback of the prototype games pilot tested.

i. Translation of exercises to games

Exergames with challenging yet achievable goals form a great incentive and motivation for keeping active. The customisation of such games is a key element of success and significantly improves the engagement of the patients with the rehabilitation program. It was found that even in cases where participants were tired or felt slight discomfort they kept on playing the games as there were personal motivational rewards in completing them and aiming at improving their score each time. We also found that a key to a successful game design is to take Parkinson's specific physiotherapy exercises and translate them into a game. It is important that the exercise movement is not too complex and can be executed on a single plane so that the motion can be accurately captured by the Wii remotes (or other equivalent MoCap system) and depicted in the game. In addition to this it is better if the game duration is relatively short or it includes little breaks to give the player time to recover and rest.

ii. Familiar activities and past hobbies make the best games

We found that simple exercises movements work the best. Also when translating exercise movements to games it is most useful to choose actions that relate to familiar activities based on sport, hobbies, or other daily activities. This works well for part hobbies and activities that people with PD used to be engaged with prior to their diagnosis, as it provides them with an enabling and creative space where past habits can be followed and reinvented.

iii. Adaptation and calibration

It is very significant that games are automatically calibrated to the range of motion of each individual player so that the required movement and exercise level in the game is adjusted automatically to offer the player the best possible experience and level of challenge. The aim of the game should be simple but challenging with the level of difficulty gradually increasing to meet the skill level and experience of the player.

iv. Game narrative: providing playful experiences

The game narrative includes the ludic aspect of the game including visual stimuli and audio output. The importance of the game narrative in accordance to enjoyment and hence engagement with the game is very high. It is a basic element of interaction and the pleasure of the human senses should always be kept a high priority. Appealing graphics, attractive scenes, charming animations and amusing sounds are essential elements of fun and directly related to the engagement of a player. Another element that should be appraised is the clear instructions and game targets. The players seem to be frustrated when they don't know how to play a game or cannot clearly see the target of it. More particularly for PD rehabilitation games, the inclusion of visual and audio feedback is a critical element in the games as they guide the player throughout and not only improve the game experience but also cognitively aid their agility.

v. Challenge and competition

Detailed feedback should be provided at the end of each game session indicating the skill level of the player and how well they have performed not only in a single game session but also offering a comparison with previous game sessions so that a progress history can be drawn. Fostering competition through the games, whether it is against oneself, the computer or another player is important in increasing player motivation and supporting people further in keeping active. Games which facilitate gameplay amongst people with PD are particularly useful as they have the potential of fostering ongoing motivation through healthy competition and a network of support through social play. In fact social games, which involve two or more people should be pursued by researchers in this area as the benefits of such interaction can support wellbeing through a game-buddy system. In this case people with similar PD diagnosis (level, age, symptoms) can be paired to offer a positive experience and a social and community feeling.

vi. PD Patient-led versus patient-centred game design

Although piloting games and game prototypes with people with PD is a useful exercise it is not the only and best approach to be adopted. The main issues with such an approach are that games are still heavily based on game designers' preconceptions and assumptions and they are tested at a late stage in the design process. With regards to the former point, despite game designers having ample experience in the design of traditional games aimed at the entertainment and leisure market, they have very limited knowledge and often understanding of the often complex and diverse requirements of people with PD. Hence when designing games for use by people with PD several assumptions are often made. Obviously people with PD, their carers and therapists are the only experts in what works and what does not regarding their very specific needs. It makes sense therefore that people with PD and professionals are involved in the design of games. Critically this should be done from the early stages of the design process in order to reduce the gulf between actual user needs, expectations and game experience and play. Such a patient-led design approach can also potentially reduce the effort and cost of designing games tailored for people with PD as inappropriate ideas would be discarded in the early phases of design. Lastly involving people with PD in the design process will have positive benefits for them, as it will offer them a sense of duty, usefulness and contribution to something for the greater good.

6.1.4 Overall discussion of the Evaluation of Serious Games for Motor Rehabilitation

The sections above presented the pilot studies conducted to evaluate the series of minigames developed for motor rehabilitation. The results of each individual study summarise the findings of these pilots. Nevertheless, the backbone system utilised for both the projects is the same and therefore a comparison of the results will offer a more rounded discussion and summary of the benefits of SGs for motor rehabilitation.

138

The first group of games, that target the post-stroke upper-limb rehabilitation, showed the importance of accurate visualisation of the movement in relation to engagement of the patient with the virtual environment. The significance of direct visual feedback of their movement mapped on the avatar is reported as paramount for the connection of the patient with the avatar. This was similarly found in the results of the PD games as the patients pointed out that, visual and auditory stimuli are important for the engagement with the game.

Again in terms of interaction, the stroke games should be developed with the movement mapped in a mirror view mode whereas the PD games target bilateral movement. The element of visual and auditory cues is not essential in SGs for stoke but it is found as beneficial in SGs for PD. Therefore, the specific interaction and visualisation requirements depend on the type of disease.

Finally, the personalisation of the game through automatic or semi-automatic calibration and adaptation is found to be very important feature in both cases. In stroke trials, it was found that the parameters of exercise duration, ROM and gameplay speed are key features for the tailoring of the game to match the personal requirements. The PD games trials revealed similar themes with the personalisation of the game environment and theme being a popular suggestion for future games.

A summary of the key themes that arose of the evaluation of the Serious Games for Motor Rehabilitation in this Thesis are depicted in the thematic mind map below (Figure 62):



Figure 62 Thematic Mind map of key themes from user trials

It is important to point out that, although the results are very positive and promising, there is still lack of empirical data to conclude on the efficiency of the use of SGs in motor rehabilitation. However, the benefits as well as the promising results of the studies conducted in the field manifest the extreme potential and bring the field one step forward to the establishment of SGs into a useful and effective tool in the toolbox of the clinical professionals.

Summary

The Evaluation, Results and Discussion Chapter of the present Thesis describes the evaluation methods utilised for the studies investigated and furthermore offers the results and discussion for each of the individual research projects included in this Thesis. The pilot testing of the SGs for stroke are offered first, with the results and discussion following. The discussion of the results emerges a series of themes for SGs for stroke rehabilitation. Following is the pilot testing of the PD games. Again, the results and discussion follow the description of the pilots. Moreover, there is the summary of the discussion into game design guidelines. Finally, Section 5.2.4 provides a more rounded discussion and summary of the benefits of games for rehabilitation in general.

CHAPTER VII: A METHODOLOGY FOR RESEARCH, DESIGN AND DEVELOPMENT OF SERIOUS GAMES

Abstract:

This chapter presents a generic methodological approach for research and development of Serious Games. It focuses on the phases of design and development, from capturing specifications and requirements, to initial design and prototype, to user-centred design and evaluation. Then the key recommendations derived from the case-studies presented in previous chapters are summarised. The outline of the chapter is depicted on Diagram 7 below:



Diagram 7 Chapter Structure

The applications, platforms, tools and games presented in the previous chapters (Chapter III and V) aimed to tackle relevant societal challenges and bring disruption through innovation in the respective fields. Their research, design, development and evaluation cycles, however, bear some similarities and as such there can emerge an overarching methodology for research, development and evaluation of Serious Games for various disciplines. The methodology can be applied equally in Defence, Health, Education or any other field that Serious Games can be proven an effective tool for provision of innovative solutions to problems. Before that, though, there needs to be a generic system architecture to accommodate the use of disruptive technologies in a mix and match manner toolbox. To accomplish that one needs to distinguish and organise a conceptual layer model. The system architecture consists of some sort of technology to be utilised and as such there needs to be a Technology layer. Data collection is essential component of the system architecture as equally; there needs to be performance assessment and user profiling in applications such as rehabilitation in health, learner's profile and assessment in education, military personnel

training etc. Furthermore, this data is visualised to assist professionals or artificial intelligence administering the interventions to make decisions on performance and progression.



Figure 63 Generic system architecture

Therefore, a data layer that is able to achieve this target is essential component of the generic system architecture. A third layer is where the entity is developed, that be a simulation platform, a game, a tool or an application and for any given domain. The entities should be distinct and separated from the other layers in order to be flexible to follow advances and disruptions that can potentially be introduced in the other layers. As an

example, the system architecture applied for Serious Games for motor rehabilitation (Figure 23) should be flexible to adapt with any input hardware technology for motion capture. In that manner, the developed applications will have a longer life-cycle as they will be able to adopt in advances of technology to a longer extent. Figure 63, above, illustrates the generic system architecture organised in the three different layers and with a word cloud within each layer describing but not limited to potential components and application areas for each layer.

After the generic system architecture is defined and outlined, the generic methodology for research, design, development and evaluation of Serious Games can be defined as well. As mentioned above there were similarities in the processes described in previous chapters within this Thesis. Figures 3 and 22 depict the adopted methodologies for the respective studies. It is easily noticeable that the two methodologies are analogous in a great deal of ways. As the outcomes of the development are Serious Games, it is also noticeable that both development processes conform to industrial processes of game development; namely Planning, Pre-Production, Production and Post-Production to a great extent but with some differences. The differences in process and approach lie in the differences between conventional entertainment games and Serious Games as the latter encompass an underlying purpose other than pure entertainment. Thus, from the work conducted in this Thesis there can be an emerged methodology for research, design, development and evaluation of Serious Games. The tasks can be broken down in steps that are organised in a coherent manner which is also compliant to the industrial stages of game development as described above. The extra tasks deriving from the underlying non-entertainment purposes are weaved in the process. Figure 64 below describes the tasks for the generic methodology for development and research of a Serious Game, domain-independent.

Planning	 Define research objectives and requirements Research target demographic and / or field of application Define technical specifications and underlying training, didactic targets and methods Design system architecture according to layer model (Figure 63) 	
Pre-Production	 Verify technology; compare against gold standards Design concept (gather assets, concept sketches, UI prototype design) Develop prototype Evaluation through pilot testing the prototype with expertusers (usability, engagement, interface) 	
Production	 9. Refine development in iterative manner 10. Finalise development 11. Test the final version for quality assurance and adherence to plan 12. Evaluation through pilot testing the final version with expert-users (usability, engagement, interface) 	
Post-Production	Post-Production 13. Evaluation trials for effectiveness (rigorous evidence) 14. Evidence-based conclusion and Postmortem	
	Serious Game with evidenced effectiveness	

Figure 64 Generic Methodology for Design, Development and Research for Serious Games

The phases of development are organised in Planning, Pre-Production, Production and Post-Production and the tasks are defined for each phase. In the planning phase, the researcher needs to define the research objectives and specific requirements of the application domain and Serious Game in respect. Moreover, there needs to be an in-depth research of the targeted cohort, in case there is a demographic. Next, the underlying purpose other than pure entertainment needs to be defined and its aims to be addressed. Next step is to compose a system architecture according to the layer model of Figure 63. This

concluded the Planning phase as there is all the information required to proceed with the prototype development.

The Pre-Production phase is where the prototype is developed. Firstly, there needs to be a verification of the architecture that is actually appropriate to be used in the context of the study. That can be a validation of accuracy as in Chapter III, section 3.2.1 and Chapter V, section 5.2.2 where there was experimental validation of accuracy for the two technologies used in the studies of this Thesis. Following is the design of the concept and development of a prototype. A user-centred approach is employed in this case of Serious Games, as per a design evaluation method with a focus on usability, engagement and acceptance. Once the prototype is fully developed and tested with expert-users, the development stage can be initiated in the Production phase.

The production phase encompasses an iterative approach where the prototype is refined according to feedback gathered from user trials and finalised in order to be trialed for the second time. As such the same cohort tries the second version of the prototype and for the same aspects. The data captured from the user trials can emerge to design themes and guidelines for future reference as in Chapters IV and VI of this Thesis. The guidelines are scalable and be used by other professionals, researchers and practitioners of the field and as such are defined a contribution to Science.

Finally, the post-production phase is dedicated to a rigorous evaluation on the effectiveness of the developed application, tool, platform or game. A larger scale evaluation has to be conducted to conclude on the efficiency of the innovative intervention and the effectiveness it is able to bring in the respective domain. A Post-mortem in the case of Serious Games can be metaphorically the dissemination of the technical report and results to an academic and technical audience for further exploitation.

Summary

This chapter presented a generic system architecture approach and a methodology for research, design and development of Serious Games, irrespective of the field of applications. The tasks for research, design and development are organised and presented in detail and are also recommended as a methodological framework that can be applied in any scenario for research and development of Serious Games.

CHAPTER VIII: CONCLUSION AND FUTURE WORKS

Abstract:

This chapter provides the summary of all chapters included in the Thesis and it provides the reader with a conclusion of author's work. Moreover, this chapter offers a discussion on future developments of the presented work as well as future work on the subject. The diagram below (Diagram 8) illustrates the structure of the chapter:



Diagram 8 Chapter Structure

8.1 Thesis Summary

This Thesis consists of five chapters. The first chapter provided an introduction to the research background and motivation as well as the aims and objectives of the Thesis. It offered the definitions of the terms related to the topic and also briefly discussed the potential application domains of SGs and other important issues of their development. Finally, Chapter I presented the Thesis structure with a short description of the Chapter contents and organisation.

The application of SGs spans across a wide variety of fields and is a novel tool and a modern trend to knowledge transfer and learning. Therefore, Chapter II presented the most popular domains of SGs providing the reader with an introductory text to the subject. The historic origins of the field are also presented in this chapter. Following the introduction of the field with the aforementioned presentation of the most popular domains, a comprehensive review of work related to the aims of the Thesis is presented. The areas under investigation and their related work that is available in the literature are presented, in order to provide the reader with an elaborate presentation of the existing state of the art of the subjects. Therefore, related work for the subject of SGs, VR simulations and guidelines for the PIPV and for the subject of SGs for home-based rehabilitation of motor control are presented in a more comprehensive literature survey.

Moving on, the specific problems identified through the research of the literature are presented in Chapters III and V.

Chapter III presents the research work performed for the Solar Soldier project which investigated the use of SGs and VR simulations to ascertain the integration of PV on the uniform and equipment of the modern infantry soldier. The Chapter describes the problem and also presents the proposed solution. The methodology of the proposed solution is offered along with the experimental accuracy test of the system employed and following is the implementation of the simulation platform.

Chapter V presents the work performed in two studies, namely the ReWiiRe project and the Games for PD rehabilitation project. These studies investigated the use of bespoke SGs and game sensors for the rehabilitation of motor control for neurophysiological and neurodegenerative diseases, such as stroke and PD respectively. Chapter IV describes the problems and also summarises the challenges and opportunities that arise by the literature review for each disease and the emerging common themes. Then, the proposed system with the adapted methodology, the experimental accuracy study of the system is presented. Finally the implementation of SGs for each disease is presented.

In Chapters IV and VI, the reader can find the evaluation of all the aforementioned systems and platforms as well as the results of the simulations and user-testing trials. The discussion of the results produces a series of emerging themes and guidelines, which are also presented in this chapter after each individual evaluation section.

Chapter VII presents a generic methodological approach for research and development of Serious Games and discusses the key observations, recommendations and challenges derived from the evaluation of applications and tools developed.

8.2 Thesis Conclusion

This Thesis has investigated the innovative use of VR and Gaming technologies in the development of SGs in the public domain service areas of Defence and Health.

More particularly, the use of VR and 3D simulations in the integration of PIPV on the uniform and equipment of the modern infantry soldier was the first subject investigated. This study has proposed an innovative virtual simulation framework that mimics closely the military environment for the purposes of investigating the integration of PIPV technology on the infantry soldier by analysing and measuring the effectiveness of light capture on various areas of the uniform and equipment of the virtual soldier. The platform is the first to facilitate the light analysis of a virtual environment incorporating movement and animated virtual avatars. It is also re-usable as a product, as the user can easily create new scenarios by importing new virtual environments and animation clips into the platform or by adjusting the daylight system to any possible condition.

The examined case studies covered several basic military environments as well as the several potential areas of integration of the PV device after interacting with the army personnel. After performing numerous simulations, the resulting data were organised and presented in such a manner enabling the classification of the examined areas in order of power generation efficiency. The derived overall classification infers preliminary qualitative and novel guidelines for any designer or practitioner of wearable military applications.

The second subject was the investigation of the innovative use of bespoke SGs and game sensors in a home-based rehabilitation system for motor deficits due to neurophysiological or neurodegenerative diseases such as stroke and PD. The conducted studies presented a novel system, developed to offer a customised VR-based rehabilitation intervention by using low-cost off-the-shelf game sensors (the Nintendo Wii remote) and a number of bespoke games. The presented system offers full kinematic tracking of the user's upper limb movement in real-time and the games were designed with a patient-centric approach with a selection of physiotherapy exercises having been translated into game experiences and playful activities. A set of emerging themes and design guidelines were extracted by the evaluation of the games through iterative pilot testing. It is intended that the proposed guidelines inform future researchers and game designers in the production of SGs that stimulate and engage people suffering from neurophysiological or neurodegenerative diseases with their physical therapy.

It is evident that the recent evolution of VR and game technologies has an engaging effect on the application of knowledge transfer. Although there is still lack of empirical evidence on the effectiveness of SGs in the process of knowledge acquisition, the promising results of a broad range of studies has shown that the users tend to engage with SGs and thus they are rendered an ideal candidate for novel and alternative learning experiences. Therefore and until there is sufficient evidence on the effectiveness and not only promising results, the scholars are recommended to keep investigating the use of the advances of VR technology into the development of next generation of SGs for any of the Health rehabilitation application domains.

8.3 Future work

Based on the design and implementation of the proposed work and the corresponding results of the evaluation, this section provides a set of discussions regarding additions and improvements as well as the author's vision of the future.

8.3.1 The Solar Soldier scenario generator

Chapter III discussed the development of a platform that simulates and analyses the light harvest on the uniform and equipment of the modern infantry soldier. The scenarios investigated for the purposes of the evaluation, covered a basic range of military scenarios and focused on the British army deployments throughout the world. However, the scenarios of real military operations are countless and the development of a scenario generator is proposed. The author envisages a user interface that any designer or practitioner could use to generate a military scenario. A database of 3D models of terrains, soldiers, uniforms and equipment would be available as resources. Furthermore, a collection of animated clips covering a variety of military related movements is also suggested. Then, the user of the scenario generator would have to go through a step-by-step wizard to create a bespoke scenario. The end product would be a 3D scene that would encompass a terrain, a virtual soldier with selected equipment, animation clips and regional daylight configurations. This development would simplify the scenario creation and utilising online free 3D object databases the expansion of scenarios is limitless. The proposed system would also be responsible of adjusting and transforming the 3D objects into the same units and scale and for merging the animation clips.

8.3.2 Home-based motor rehabilitation with monitoring capabilities

Chapter V proposed a framework for home-based motor rehabilitation with monitoring capabilities. The architecture employed for the development of the system facilitates the online monitoring through the biofeedback layer as seen in Figure 23 of Chapter V. The motion data captured by the patient is transmitted online to both the local terminal and the remote clinician's terminal. This enables the live monitoring of the performance and furthermore is able to foster the manual difficulty adjustment by the clinician. The game parameters like speed, time etc. can be adjusted remotely and this enables the clinician to keep the patient constantly challenged through the increase of the difficulty levels. Moreover, the clinician platform would have the capability of visualising the movement on a 3D avatar as well as providing with accurate measures of the ROM of the limbs, speed of movement, number of repetitions and time. Additionally, the platform would be able to record all the sessions per patient in a safe storage and for further evaluation of the performance of each patient. Furthermore, the limitations of the gaming sensors as capturing devices are possibly covered by the advantages of another game sensor. For example the limitation of the MS Kinect to capture rotational movement can be covered by the use of a Wiimote or vice versa, the depth data of the Kinect can aid the 3D positioning of the Wiimote. This fact raises new challenges in the development of SGs for motor rehabilitation as the use of collaborative game sensors has to be examined.

8.4 Impact and Vision

The research conducted for the accomplishment of this Thesis is driven by delivering a series of benefits to the society that aim to enhance quality of life, health and environmental awareness. The transformational impacts of this research belong in the areas of Energy and Sustainability and Health, quality of life and ageing population. The use of innovative technologies such as virtual reality and gaming technology and the development of bespoke SGs for Defence with sustainability and Motor rehabilitation will improve the societal

aspects mentioned above. More particularly, the use of the Solar Soldier simulation platform provides a safe environment to test and develop the application of renewable energy sources thus impacting the environment. Moreover, the use of renewable energy sources such as photovoltaics eventually impacts the financial aspect of the modern Defence policies by reducing the costs allocated for batteries. The use of SGs for motor rehabilitation, apart from the direct impact of improving a patient's well-being, will decongest the hospitals and health centres through the home-based rehabilitation platform with remote monitoring capabilities. Thus, apart from the human-centred impact there is benefit in the potential reduction of healthcare costs with the application of this novel rehabilitation practice.

Moreover, the research outreaches to the next generation of researchers, practitioners, scientists and scholars of the field as well as other related fields by providing frameworks and scalable applications that can be applied to other disciplines. The methodological approach of the technology application can be transferred to a variety of disciplines and research fields such as Education or Sports. Finally, to increase the capabilities of this research, a series of articles have been authored and disseminated in international publications (please see List of Publications section) to maximise the effect of dissemination.

The prospect of implementing the future work presented above as well as the scalability of the research reveal potentials of advanced research and exploration in future projects and multiple disciplines. The advances of technology with regards to VR and games are rapid and realise more opportunities for research in the field by applying the novel hardware and software in the SGs curriculum. The vision of the author in the field of SGs, and in particular for the domains of Defence and Health, embraces the whole spectrum of the technological advances that, in the near future, will lead to an augmented reality practically indistinguishable from the real world and always for the benefit of people and overall society.

REFERENCES

- [1] M. Zyda, "From visual simulation to virtual reality to games," *Computer (Long. Beach. Calif).*, no. September, pp. 25–32, 2005.
- [2] S. Deterding, D. Dixon, R. Khaled, and L. Nacke, "From game design elements to gamefulness: defining gamification," *Proceedings 15 Int. Acad. MindTrek Conf.*, pp. 9–15, 2011.
- [3] D. Michael and S. Chen, *Serious Games. Games that educate, train, and inform.* Boston,MA: Thomson Course Technology, 2006.
- [4] D. Crookall, "Serious Games, Debriefing, and Simulation/Gaming as a Discipline," *Simul. Gaming*, vol. 41, no. 6, pp. 898–920, Jan. 2011.
- [5] A. Martens, H. Diener, and S. Malo, "Game-Based Learning with Computers -Learning, Simulations, and Games. Springer Berlin," *Trans. Edutainment.*, vol. vol. 5080., pp. pp172–190, 2008.
- [6] R. Hunicke, M. LeBlanc, and R. Zubek, "MDA: A formal approach to game design and game research," *Proc. Challenges Games AI Work. Ninet. Natl. Conf. Artif. Intell.*, pp. 1–5, 2004.
- [7] M. Prensky, "Complexity matters. Mini-games are Trivial but 'Complex' Games Are Not," *Science*, vol. 45, no. 4, Feb. 2005.
- [8] T. Susi, M. Johannesson, and P. Backlund, "Serious games: An overview (technical report)," 2007.
- [9] C. Dring, "Mike Hayes helps launch Serious Games studio," 2012. [Online]. Available: http://www.mcvuk.com/news/read/mike-hayes-helps-launch-seriousgames-studio/0103962. [Accessed: 30-Mar-2014].
- [10] E. a. Boyle, T. M. Connolly, T. Hainey, and J. M. Boyle, "Engagement in digital entertainment games: A systematic review," *Comput. Human Behav.*, vol. 28, no. 3, pp. 771–780, May 2012.
- [11] T. M. Connolly, E. a. Boyle, E. MacArthur, T. Hainey, and J. M. Boyle, "A systematic literature review of empirical evidence on computer games and serious games," *Comput. Educ.*, vol. 59, no. 2, pp. 661–686, Sep. 2012.
- [12] M. Prensky, *Game-Based Learning*. New York, New York, USA: McGraw Hill, 2001.
- [13] S. I. de Freitas, "Using games and simulations for supporting learning," *Learn. Media Technol.*, vol. 31, no. 4, pp. 343–358, Dec. 2006.
- [14] P. Wouters, C. van Nimwegen, H. van Oostendorp, and E. D. van der Spek, "A meta-analysis of the cognitive and motivational effects of serious games.," *J. Educ. Psychol.*, vol. 105, no. 2, pp. 249–265, 2013.
- [15] D. Djaouti and J. Alvarez, "Origins of serious games," *Serious games Edutainment Appl.*, pp. 1–22, 2011.
- [16] C. Abt, Serious Games. New York: University Press of America, 1970.

- [17] D. R. Jansiewicz, *The New Alexandria Simulation: A Serious Game of State and Local Politics*. New York: Canfield Press, 1973.
- [18] K. Squire, "Video game-based learning: An emerging paradigm for instruction," *Perform. Improv. Q.*, vol. 21, no. 2, pp. 7–36, 2008.
- [19] C. Girard, J. Ecalle, and a. Magnan, "Serious games as new educational tools: how effective are they? A meta-analysis of recent studies," J. Comput. Assist. Learn., vol. 29, no. 3, pp. 207–219, Jun. 2013.
- [20] M. F. Young, S. Slota, a. B. Cutter, G. Jalette, G. Mullin, B. Lai, Z. Simeoni, M. Tran, and M. Yukhymenko, "Our Princess Is in Another Castle: A Review of Trends in Serious Gaming for Education," *Rev. Educ. Res.*, vol. 82, no. 1, pp. 61–89, Feb. 2012.
- [21] W. L. Bedwell, D. Pavlas, K. Heyne, E. H. Lazzara, and E. Salas, "Toward a Taxonomy Linking Game Attributes to Learning: An Empirical Study," *Simul. Gaming*, vol. 43, no. 6, pp. 729–760, May 2012.
- [22] S. de Freitas, K. Kiili, M. Ney, M. Ott, M. Popescu, M. Romero, and I. Stanescu, "GEL: Exploring Game Enhanced Learning," *Procedia Comput. Sci.*, vol. 15, pp. 289–292, Jan. 2012.
- [23] L. a. Annetta, J. Minogue, S. Y. Holmes, and M.-T. Cheng, "Investigating the impact of video games on high school students' engagement and learning about genetics," *Comput. Educ.*, vol. 53, no. 1, pp. 74–85, Aug. 2009.
- [24] P. Backlund and M. Hendrix, "Educational Games–Are They Worth The Effort?," 5th Int. Conf. Games Virtual Worlds Serious Appl., pp. 1–8, 2013.
- [25] D. Charsky, "From Edutainment to Serious Games: A Change in the Use of Game Characteristics," *Games Cult.*, vol. 5, no. 2, pp. 177–198, Feb. 2010.
- [26] M. Dickey, "Three dimensional virtual worlds and distance learning: two case studies of Active Worlds as a medium for distance education," *Br. J. Educ. Technol.*, vol. 36, no. 3, 2005.
- [27] A. Harris, N. Yuill, and R. Luckin, "The influence of context-specific and dispositional achievement goals on children's paired collaborative interaction.," *Br. J. Educ. Psychol.*, vol. 78, no. Pt 3, pp. 355–74, Sep. 2008.
- [28] M. J. Mayo, "Video games: a route to large-scale STEM education?," *Science*, vol. 323, no. 5910, pp. 79–82, Jan. 2009.
- [29] M. Kebritchi, A. Hirumi, and H. Bai, "The effects of modern mathematics computer games on mathematics achievement and class motivation," *Comput. Educ.*, vol. 55, no. 2, pp. 427–443, Sep. 2010.
- [30] F. Ke and B. Grabowski, "Gameplaying for maths learning: cooperative or not?," *Br. J. Educ. Technol.*, vol. 38, no. 2, pp. 249–259, Mar. 2007.
- [31] A. El Rhalibi, M. Merabti, R. Yun, and D. Liu, "Game Based Learning Framework for Virtual 3D Dinosaurs Knowledge," 2011 Dev. E-systems Eng., pp. 419–424, 2011.

- [32] A. El Rhalibi, M. Merabti, R. Yun, and D. Liu, "Game based learning with Homura 3D, for dinosaurs life teaching," in *2011 IEEE Consumer Communications and Networking Conference, CCNC'2011*, 2011, pp. 332–336.
- [33] S. Barab, P. Pettyjohn, M. Gresalfi, C. Volk, and M. Solomou, "Game-based curriculum and transformational play: Designing to meaningfully positioning person, content, and context," *Comput. Educ.*, vol. 58, no. 1, pp. 518–533, Jan. 2012.
- [34] S. Tang, M. Hanneghan, A. El Rhalibi, and A. El-Rhalibi, "Describing Games for Learning: Terms, Scope and Learning Approaches," in *Proceedings of the Game Design and Technology Workshop and Conference (GDTW 2007)*, 2007, pp. 98– 102.
- [35] Ecogamer.org, "Green Game: Environmental Games | Ecogamer.ORG," 2013. [Online]. Available: http://ecogamer.org/environmental-games. [Accessed: 30-Mar-2014].
- [36] SGA, "Serious Games Directory," *Serious Games Association*, 2012. [Online]. Available: http://www.seriousgamesdirectory.com/. [Accessed: 30-Mar-2014].
- [37] K. Katsaliaki and N. Mustafee, "A survey of serious games on sustainable development," *Proc. 2012 Winter Simul. Conf. (WSC)*, pp. 1 13, 2012.
- [38] G. Liarakou and E. Sakka, "Evaluation of serious games, as a tool for education for sustainable development," in *EDEN Annual Conference*, 2011, pp. 96–110.
- [39] D. Vara, E. Macias, S. Gracia, A. Torrents, and S. Lee, "Meeco: Gamifying ecology through a social networking platform," in *2011 IEEE International Conference on Multimedia and Expo (ICME)*, 2011, pp. 1–6.
- [40] A. Wein and W. Labiosa, "Serious Games Experiment toward Agent-based Simulation: U.S. Geological Survey Open File Report," Reston, Virginia, 2013.
- [41] M. Popescu, M. Romero, and M. Usart, "Serious Games for Serious Learning Using SG for Business, Management and Defence Education," *Int. J. Comput. Sci. Res. Appl.*, vol. 03, no. 01, pp. 5–15, 2013.
- [42] M. Ulicsak, "Games in Education : Serious Games.," A FutureLab Literature Review, 2010. [Online]. Available: http://media.futurelab.org.uk/resources/documents/lit_reviews/Serious-Games_Review.pdf. [Accessed: 30-Mar-2014].
- [43] A. Alexander and T. Brunyé, "From gaming to training: A review of studies on fidelity, immersion, presence, and buy-in and their effects on transfer in pc-based simulations and games," *DAWARS Training Impact Group*, 2005. [Online]. Available: http://www.aptima.com/publications/2005_Alexander_Brunye_Sidman_Weil.pdf. [Accessed: 30-Dec-2013].
- [44] M. Prensky, "Digital game-based learning," *Comput. Entertain.*, vol. 1, no. 1, p. 21, Oct. 2003.
- [45] Department of Defence Games Community, "Department of Defence Games Community," 2013. [Online]. Available: http://www.dodgamecommunity.com/. [Accessed: 30-Dec-2013].

- [46] GoArmy Games, "GoArmy Games," 2013. [Online]. Available: http://www.goarmy.com/downloads/games.html. [Accessed: 30-Dec-2013].
- [47] BiStudio, "Bohemia Interactive," 2011. [Online]. Available: http://www.bistudio.com/. [Accessed: 03-Mar-2014].
- [48] DailyMail, "Troops set to take virtual plunge: MoD launches £500,000 parachute jump simulator," 2011. [Online]. Available: http://www.dailymail.co.uk/sciencetech/article-1361500/MoD-launches-500-000parachute-jump-simulator.html. [Accessed: 30-Dec-2013].
- [49] F. A. dos Santos Mendes, J. E. Pompeu, A. Modenesi Lobo, K. Guedes da Silva, T. D. P. Oliveira, A. Peterson Zomignani, and M. E. Pimentel Piemonte, "Motor learning, retention and transfer after virtual-reality-based training in Parkinson's disease--effect of motor and cognitive demands of games: a longitudinal, controlled clinical study.," *Physiotherapy*, vol. 98, no. 3, pp. 217–23, Sep. 2012.
- [50] J. E. Pompeu, F. A. D. S. Mendes, K. G. Da Silva, A. M. Lobo, T. D. P. Oliveira, A. P. Zomignani, and M. E. P. Piemonte, "Effect of Nintendo WiiTM-based motor and cognitive training on activities of daily living in patients with Parkinson's disease: a randomised clinical trial.," *Physiotherapy*, vol. 98, no. 3, pp. 196–204, Sep. 2012.
- [51] I. Paraskevopoulos and E. Tsekleves, "Use of gaming sensors and customised exergames for Parkinson's Disease rehabilitation: a proposed virtual reality framework," in *Games and Virtual Worlds for Serious Applications (VS-GAMES)*, 2013 5th International Conference on, 2013, pp. 1–5.
- [52] M. Graafland, J. M. Schraagen, and M. P. Schijven, "Systematic review of serious games for medical education and surgical skills training.," *Br. J. Surg.*, vol. 99, no. 10, pp. 1322–30, Oct. 2012.
- [53] P. Rego, P. Moreira, and L. Reis, "Serious games for rehabilitation: A survey and a classification towards a taxonomy," in 2010 5th Iberian Conference on Information Systems and Technologies (CISTI), 2010.
- [54] E. Tsekleves, A. Warland, C. Kilbride, I. Paraskevopoulos, and D. Skordoulis, "The use of the Nintendo Wii in motor rehabilitation for virtual reality interventions: a literature review," in *Virtual, Augmented Reality and Serious Games for Healthcare 1*, vol. 68, Prof. Minhua Ma, Ed. Springer-Verlag, 2014, pp. 321–344.
- [55] S. Jarvenpaa, D. Leidner, R. Teigland, and M. Wasko, "MISQ Special Issue: New Ventures in Virtual Worlds," *MIS Q. Call Pap.*, 2007.
- [56] A. R. Chaturvedi, D. R. Dolk, and P. L. Drnevich, "DESIGN PRINCIPLES FOR VIRTUAL WORLDS.," *MIS Q.*, vol. 35, no. 3, pp. 673–684, 2011.
- [57] G. Chryssolouris, D. Mavrikios, D. Fragos, and V. Karabatsou, "Virtual realitybased experimentation environment for the verification of human-related factors in assembly processes," *Robot. Comput. Integr. Manuf.*, vol. 16, no. 4, pp. 267– 276, 2000.
- [58] D. A. Reece, "Movement Behavior for Soldier Agents on a Virtual Battlefield," *Presence: Teleoperators and Virtual Environments*, vol. 12, no. 4. pp. 387–410, 2003.

- [59] J. Yang, S. Rahmatalla, T. Marler, and K. Abdel-malek, "Validation of Predicted Posture for the Virtual Human SantosTM," *HCII*, vol. LNCS 4561, no. Digital Human Modeling, pp. 500–510, 2007.
- [60] K. Abdel-malek, J. Yang, J. H. Kim, T. Marler, S. Beck, C. Swan, L. Frey-law, A. Mathai, and C. Murphy, "Development of the Virtual-Human Santos," in *Digital Human Modeling*, 2007, pp. 490–499.
- [61] Y.-H. Shiau and S.-J. Liang, "Real-Time Network Virtual Military Simulation System," 2007 11th Int. Conf. Inf. Vis. (IV '07), no. 1, pp. 807–812, Jul. 2007.
- [62] G. Y. G. Ying, L. L. Lei, W. X. W. Xiuliang, Z. P. Z. Puzhao, and G. S. G. Shuxia, "Design and Realization of Virtual Scene System in UAV Electronic Warfare," 2009 Int. Conf. Inf. Eng. Comput. Sci., 2009.
- [63] E. M. Blount, S. I. Ringleb, a. Tolk, M. Bailey, and J. a. Onate, "Incorporation of physical fitness in a tactical infantry simulation," *J. Def. Model. Simul. Appl. Methodol. Technol.*, vol. 10, no. 3, pp. 235–246, Sep. 2011.
- [64] T. M. Cioppa, T. W. Lucas, and S. M. Sanchez, "Military Applications of Agent-Based Simulations," *Proc. 2004 Winter Simul. Conf. 2004.*, vol. 1, pp. 165–174, 2004.
- [65] E. Bitinas, Z. Henscheid, and L. Truong, "Pythagoras: A new agent-based simulation system," *Technol. Rev. J.*, pp. 45–58, 2003.
- [66] D. Schmorrow, J. Cohn, and D. Nicholson, *The PSI Handbook of Virtual Environments for Training and Education [Three Volumes]: Developments for the Military and Beyond*. Westport, CT, USA: Praeger Publishers, 2008.
- [67] B. Parida, S. Iniyan, and R. Goic, "A review of solar photovoltaic technologies," *Renew. Sustain. Energy Rev.*, vol. 15, no. 3, pp. 1625–1636, Apr. 2011.
- [68] L. El Chaar, L. a. Lamont, and N. El Zein, "Review of photovoltaic technologies," *Renew. Sustain. Energy Rev.*, vol. 15, no. 5, pp. 2165–2175, Jun. 2011.
- [69] G. Hashmi, K. Miettunen, T. Peltola, J. Halme, I. Asghar, K. Aitola, M. Toivola, and P. Lund, "Review of materials and manufacturing options for large area flexible dye solar cells," *Renew. Sustain. Energy Rev.*, vol. 15, no. 8, pp. 3717– 3732, Oct. 2011.
- [70] N. H. Reich, W. G. J. H. M. van Sark, W. C. Turkenburg, and W. C. Sinke, "Using CAD software to simulate PV energy yield - The case of product integrated photovoltaic operated under indoor solar irradiation," *Sol. Energy*, vol. 84, no. 8, pp. 1526–1537, 2010.
- [71] H. Mousazadeh, A. Keyhani, A. Javadi, H. Mobli, K. Abrinia, and A. Sharifi, "A review of principle and sun-tracking methods for maximizing solar systems output," *Renew. Sustain. Energy Rev.*, vol. 13, no. 8, pp. 1800–1818, Oct. 2009.
- [72] A. Kovach and J. Schmid, "Determination of energy output losses due to shading of building-integrated photovoltaic arrays using a raytracing technique," *Sol. Energy*, vol. 57, no. 2, pp. 117–124, 1996.

- [73] D. D. Nguyen, B. Lehman, and S. Kamarthi, "Solar Photovoltaic Array's Shadow Evaluation Using Neural Network with On-Site Measurement," 2007 IEEE Canada Electr. Power Conf., pp. 44–49, Oct. 2007.
- [74] L. Cortez and J. Cortez, "Study of the effects of random changes of solar radiation on energy production in a photovoltaic solar module," *Can. J. Electr. Electron. Eng.*, vol. 1, no. 4, 2010.
- [75] B. Celik, E. Karatepe, N. Gokmen, and S. Silvestre, "A virtual reality study of surrounding obstacles on BIPV systems for estimation of long-term performance of partially shaded PV arrays," *Renew. Energy*, vol. 60, pp. 402–414, 2013.
- [76] I. Paraskevopoulos and E. Tsekleves, "Simulation of photovoltaics for defence and commercial applications by extending existing 3D authoring software: a validation study," in SIMULTECH 2012. International Conference on Simulation and Modeling Methodologies, Technologies and Applications, 2011, pp. 384–396.
- [77] C. Reinhart and P. Breton, "Experimental Validation Of 3ds Max® Design 2009 And Daysim 3.0," in *11th International IBPSA Conference, Building Simulation* 2009, 2009, pp. 1–8.
- [78] A. H. M. E. Reinders, "A design method to assess the accessibility of light on PV cells in an arbitrary geometry by means of ambient occlusion," in 22nd European *Photovoltaic Solar Energy Conference and Exhibition*, 2007.
- [79] A. H. M. E. Reinders, "Product-integrated PV applications How industrial design methods yield innovative PV powered products," 2008 33rd IEEE Photovolt. Spec. Conf., 2008.
- [80] A. J. Veldhuis and A. H. M. E. Reinders, "Real-Time Irradiance Simulation for PV Products and Building Integrated PV in a Virtual Reality Environment," *IEEE Journal of Photovoltaics*, vol. 2, no. 3. pp. 352–358, 2012.
- [81] N. H. Reich, W. G. J. H. M. van Sark, A. H. M. E. Reinders, and H. de Wit, "Using CAD software to simulate PV energy yield: Predicting the charge yield of solar cells incorporated into a PV powered consumer product under 3D-irradiation conditions," *Photovolt. Spec. Conf. (PVSC), 2009 34th IEEE*, 2009.
- [82] N. Reich, "On the design of product integrated photovoltaic systems," Utrecht University, 2010.
- [83] M. Muller, J. Wienold, W. D. Walker, and L. M. Reindl, "Characterization of indoor photovoltaic devices and light," *Photovolt. Spec. Conf. (PVSC), 2009 34th IEEE*, 2009.
- [84] A. Mestre and J. C. Diehl, "Ecodesign and Renewable Energy: How to Integrate Renewable Energy Technologies into Consumer Products," 2005 4th Int. Symp. Environ. Conscious Des. Inverse Manuf., 2005.
- [85] M. B. Schuber, Y. Ishikawa, J. W. Kramer, C. E. M. Gemmer, and J. H. Werner, "Clothing integrated photovoltaics," *Conf. Rec. Thirty-first IEEE Photovolt. Spec. Conf. 2005.*, 2005.
- [86] M. B. Schubert and J. H. Werner, "Flexible solar cells for clothing," *Mater. Today*, vol. 9, no. 6, pp. 42–50, 2006.

- [87] K. Ashok, "Powered Clothing," *Technical Report*, 2007. [Online]. Available: http://www.fibre2fashion.com/industry-article/pdffiles/POWERED-CLOTHING.pdf. [Accessed: 15-Dec-2012].
- [88] F. C. Krebs, M. Biancardo, B. Winther-Jensen, H. Spanggard, and J. Alstrup, "Strategies for incorporation of polymer photovoltaics into garments and textiles," *Sol. Energy Mater. Sol. Cells*, vol. 90, no. 7–8, pp. 1058–1067, 2006.
- [89] M. K. Singh, "Flexible Photovoltaic Textiles for Smart Applications," in Solar Cells - New Aspects and Solutions, Prof. Leonid Kosyachencko, Ed. InTech, 2011, pp. 43–68.
- [90] D. Jia, Y. Duan, and J. Liu, "Emerging technologies to power next generation mobile electronic devices using solar energy," *Frontiers of Energy and Power Engineering in China*, vol. 3, no. 3. pp. 262–288, 2009.
- [91] A. H. M. E. Reinders, "Options for photovoltaic solar energy systems in portable products," in *Proceedings of TMCE*, 2002.
- [92] K. E. Laver, S. George, S. Thomas, J. E. Deutsch, M. Crotty, L. Ke, and D. Je, "Virtual reality for stroke rehabilitation (Review)," *Cochrane Database Syst. Rev.*, no. 9, p. CD008349, 2011.
- [93] L. P. Lisa, A. Jughters, and E. Kerckhofs, "The effectiveness of different treatment modalities for the rehabilitation of unilateral neglect in stroke patients: a systematic review.," *NeuroRehabilitation*, vol. 33, no. 4, pp. 611–20, Jan. 2013.
- [94] K. R. Lohse, C. G. E. Hilderman, K. L. Cheung, S. Tatla, and H. F. M. Van der Loos, "Virtual reality therapy for adults post-stroke: a systematic review and meta-analysis exploring virtual environments and commercial games in therapy.," *PLoS One*, vol. 9, no. 3, p. e93318, Jan. 2014.
- [95] K. Thomson, A. Pollock, C. Bugge, and M. Brady, "Commercial gaming devices for stroke upper limb rehabilitation: A systematic review.," *Int. J. Stroke*, vol. 9, no. 4, pp. 479–88, Jun. 2014.
- [96] M. C. Moreira, A. M. de Amorim Lima, K. M. Ferraz, and M. A. Benedetti Rodrigues, "Use of virtual reality in gait recovery among post stroke patients--a systematic literature review.," *Disabil. Rehabil. Assist. Technol.*, vol. 8, no. 5, pp. 357–62, Sep. 2013.
- [97] G. Fluet and J. Deutsch, "Virtual Reality for Sensorimotor Rehabilitation Post-Stroke: The Promise and Current State of the Field," *Curr. Phys. Med. Rehabil. Reports.*, vol. 1, no. 1, pp. 9–20, 2013.
- [98] E. Pietrzak, C. Cotea, and S. Pullman, "Using commercial video games for upper limb stroke rehabilitation: is this the way of the future?," *Top. Stroke Rehabil.*, vol. 21, no. 2, pp. 152–62, 2014.
- [99] S. Rahman and A. Shaheen, "Virtual reality use in motor rehabilitation of neurological disorders: A systematic review," *Middle-East J. Sci. Res.*, vol. 7, no. 1, pp. 63–70, 2011.
- [100] M. Ortiz-catalan, S. Nijenhuis, K. Ambrosch, T. B. Eerdt, S. Koenig, and B. Lange, "Emerging Therapies in Neurorehabilitation," in *Emerging Therapies in*

Neurorehabilitation, vol. 4, J. L. Pons and D. Torricelli, Eds. Berlin, Heidelberg: Springer Berlin Heidelberg, 2014, pp. 249–265.

- [101] C. M. Bleakley, D. Charles, A. Porter-Armstrong, M. D. J. McNeill, S. M. McDonough, and B. McCormack, "Gaming for Health: A Systematic Review of the Physical and Cognitive Effects of Interactive Computer Games in Older Adults.," J. Appl. Gerontol., Jan. 2013.
- [102] D. J. Reinkensmeyer and M. L. Boninger, "Technologies and combination therapies for enhancing movement training for people with a disability," *Journal of NeuroEngineering and Rehabilitation*, vol. 9, no. 1. p. 17, 2012.
- [103] D. Levac, P. Miller, and C. Missiuna, "Usual and Virtual Reality Video Gamebased Physiotherapy for Children and Youth with Acquired Brain Injuries," *Physical & Occupational Therapy in Pediatrics*, vol. 32, no. 2. pp. 180–195, 2012.
- [104] B. Lange, S. Koenig, C.-Y. Chang, E. McConnell, E. Suma, M. Bolas, and A. Rizzo, "Designing informed game-based rehabilitation tasks leveraging advances in virtual reality," *Disability and Rehabilitation*, vol. 34, no. 22. pp. 1863–1870, 2012.
- [105] B. Williamson, C. Wingrave, and J. J. LaViola, "RealNav: Exploring natural user interfaces for locomotion in video games," in *3DUI 2010 - IEEE Symposium on 3D User Interfaces 2010, Proceedings*, 2010, pp. 3–10.
- [106] L. Y. Joo, T. S. Yin, D. Xu, E. Thia, P. F. Chia, C. W. K. Kuah, and K. K. He, "A feasibility study using interactive commercial off-the-shelf computer gaming in upper limb rehabilitation in patients after stroke," *J. Rehabil. Med.*, vol. 42, no. 5, pp. 437–441, 2010.
- [107] G. N. Lewis and J. a Rosie, "Virtual reality games for movement rehabilitation in neurological conditions: how do we meet the needs and expectations of the users?," *Disabil. Rehabil.*, vol. 34, no. 22, pp. 1880–6, Jan. 2012.
- [108] R. S. Leder, G. Azcarate, R. Savage, S. Savage, L. E. Sucar, D. Reinkensmeyer, C. Toxtli, E. Roth, and A. Molina, "Nintendo Wii remote for computer simulated arm and wrist therapy in stroke survivors with upper extremity hemipariesis," in 2008 Virtual Rehabilitation, IWVR, 2008, p. 74.
- [109] M. Pälmke, H. Von Piekartz, C. Zalpour, T. Schüler, and K. Morisse, "A new perspective for Virtual Mirror Therapy: Developing a low-cost-high-convenient environment utilising the Wiimote," in 2009 Virtual Rehabilitation International Conference, VR 2009, 2009, p. 197.
- [110] G. Alankus, R. Proffitt, C. Kelleher, and J. Engsberg, "Stroke Therapy through Motion-Based Games," ACM Transactions on Accessible Computing, vol. 4, no. 1. pp. 1–35, 2011.
- [111] S. Attygalle, M. Duff, T. Rikakis, and J. He, "Low-cost, at-home assessment system with Wii remote based motion capture," in 2008 Virtual Rehabilitation, *IWVR*, 2008, pp. 168–174.
- [112] M. Milosevic and E. Jovanov, "A real-time control of multiple avatars using Wii remotes and avatar system," in *Proceedings of the Annual Southeastern Symposium on System Theory*, 2011, pp. 139–142.

- [113] D. Scherfgen and R. Herpers, "3D tracking using multiple Nintendo Wii Remotes: a simple consumer hardware tracking approach," in *Proceedings of the 2009 Conference on Future Play on @ GDC Canada*, 2009, pp. 31–32.
- [114] Y. P. Wuang, C. S. Chiang, C. Y. Su, and C. C. Wang, "Effectiveness of virtual reality using Wii gaming technology in children with Down syndrome," *Res. Dev. Disabil.*, vol. 32, no. 1, pp. 312–321, 2011.
- [115] P. J. Standen, D. J. Brown, S. Battersby, M. Walker, L. Connell, A. Richardson, F. Platts, K. Threapleton, and A. Burton, "A study to evaluate a low cost virtual reality system for home based rehabilitation of the upper limb following stroke," *International Journal on Disability and Human Development*, vol. 10, no. 4. pp. 337–341, 2011.
- [116] P. H. Wilson, J. Duckworth, N. Mumford, R. Eldridge, M. Guglielmetti, P. Thomas, D. Shum, and H. Rudolph, "A virtual tabletop workspace for the assessment of upper limb function in Traumatic Brain Injury (TBI)," in 2007 Virtual Rehabilitation, IWVR, 2007, pp. 14–19.
- [117] J. Martin-Moreno, D. Ruiz-Fernandez, A. Soriano-Paya, and V. Jesus Berenguer-Miralles, "Monitoring 3D movements for the rehabilitation of joints in physiotherapy.," *Conf. Proc. IEEE Eng. Med. Biol. Soc.*, vol. 2008, pp. 4836– 4839, 2008.
- [118] M. Holden, "Virtual environments for motor rehabilitation: review," *Cyberpsychology Behav.*, vol. 8, no. 3, pp. 187–211, 2005.
- [119] P. E. Cherniack, "Not just fun and games: applications of virtual reality in the identification and rehabilitation of cognitive disorders of the elderly.," *Disabil. Rehabil. Assist. Technol.*, vol. 6, no. 4, pp. 283–9, Jan. 2011.
- [120] A. Sugumaran and A. Prakash, "Wii can rehabilitate ?," *Clin. Pract.*, no. April, pp. 228–229, 2011.
- [121] N. Ben Herz, "The Nintendo Wii© and PD," Research report, parkinson.org, 2009. [Online]. Available: http://www.parkinson.org/NationalParkinsonFoundation/files/cd/cdf0b319-a28b-4082-bf07-64888dbd2c90.pdf. [Accessed: 21-Apr-2013].
- [122] Y. Jung, K. Li, and N. Janissa, "Games for a better life: effects of playing Wii games on the well-being of seniors in a long-term care facility," *Proc. Sixth Australas. Conf. Interact. Entertain.*, pp. 1–5, 2009.
- [123] J. S. Dhillon, C. Ramos, B. C. Wünsche, and C. Lutteroth, "Leveraging consumer sensing devices for telehealth," *Proc. 13th Int. Conf. NZ Chapter ACM's Spec. Interes. Gr. Human-Computer Interact. - CHINZ '12*, p. 29, 2012.
- [124] K. Zettergren, J. Franca, M. Antunes, and C. Lavallee, "The effects of Nintendo Wii Fit training on gait speed, balance, functional mobility and depression in one person with Parkinson's disease," *Med. Heal. Sci. Journal, MHSJ*, vol. 9, no. November, pp. 18–24, 2011.
- [125] C. H. Vuong, T. Ingalls, and J. J. Abbas, "Transforming clinical rehabilitation into interactive multimedia," *Proc. 19th ACM Int. Conf. Multimed. - MM '11*, p. 937, 2011.

- [126] J. Annema, M. Verstraete, V. Vanden Abeele, S. Desmet, D. Geerts, and I. K. U. Leuven, "Videogames in therapy : a therapist's perspective," *Proc. Sixth Australas. Conf. Interact. Entertain.*, pp. 94–98, 2010.
- [127] O. Assad, R. Hermann, D. Lilla, B. Mellies, R. Meyer, L. Shevach, S. Siegel, M. Springer, S. Tiemkeo, J. Voges, J. Wieferich, M. Herrlich, M. Krause, and R. Malaka, "Motion-Based Games for Parkinson's Disease Patients," in *Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics)*, J. C. Anacleto, S. Fels, N. Graham, B. Kapralos, M. S. El-Nasr, and K. Stanley, Eds. 2011, pp. 47–58.
- [128] A. Camurri, B. Mazzarino, G. Volpe, P. Morasso, F. Priano, and C. Re, "Application of multimedia techniques in the physical rehabilitation of Parkinson's patients," *J. Vis. Comput. Animat.*, vol. 14, no. 5, pp. 269–278, Dec. 2003.
- [129] D. Lilla, M. Herrlich, R. Malaka, and D. Krannich, "The influence of music on player performance in exergames for parkinson's patients," in *Entertainment Computing - ICEC 2012 Lecture Notes in Computer Science*, 2012, pp. 433–436.
- [130] G. Sanmartín, J. Flores, and P. Arias, "Motion capture for clinical purposes, an approach using primesense sensors," in *Lecture Notes in Computer Science* (*including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics*), vol. 7378 LNCS, Francisco J. Perales, R. B. Fisher, and T. B. Moeslund, Eds. 2012, pp. 273–281.
- [131] T. Martins, V. Carvalho, and F. Soares, "Development of a system for monitoring and tracking of physiotherapeutic movements in patients with neurological diseases," *3rd Port. Meet. Bioeng. (ENBENG), 2013 IEEE*, no. February, pp. 3–6, 2013.
- [132] W. Yu, C. Vuong, and T. Ingalls, "An Interactive Multimedia System for Parkinson s Patient Rehabilitation," in *Virtual and Mixed Reality - Systems and Applications, Part II, HCI 2011*, Shumaker and Randall, Eds. Springer-Verlag, 2011, pp. 129–137.
- [133] K.-J. Su, W.-J. Hwang, C.-Y. Wu, J.-J. Fang, I.-F. Leong, and H.-I. Ma, "Increasing speed to improve arm movement and standing postural control in Parkinson's disease patients when catching virtual moving balls.," *Gait Posture*, vol. 39, no. 1, pp. 65–69, Jul. 2013.
- [134] S. Göbel, S. Hardy, and V. Wendel, "Serious games for health: personalized exergames," in *Proceedings of the international conference on Multimedia*, 2010, pp. 1663–1666.
- [135] L. Geurts, V. Vanden Abeele, J. Husson, F. Windey, M. Van Overveldt, J. Annema, and S. Desmet, "Digital Games for Physical Therapy : Fulfilling the Need for Calibration and Adaptation," in *Proceedings of the fifth international conference on Tangible, embedded, and embodied interaction*, 2011, pp. 117–124.
- [136] R. R. Palacio, C. O. Acosta, A. L. Morán, and J. Cortez, "Towards videogame design guidelines to promote significant leisure activities in Mexican older adults," *Proc. 4th Mex. Conf. Human-Computer Interact. - MexIHC '12*, p. 43, 2012.

- [137] A. M. A. Lima, F. de C. C. Hirata, G. S. de Bruin, R. M. S. Mota, and V. M. S. de Bruin, "The influence of playing a non-reward game on motor ability and executive function in Parkinson's disease," *Behav. Neurol.*, vol. 25, no. 2, pp. 119–125, 2012.
- [138] B. Bouchard, F. Imbeault, A. Bouzouane, B. J. Menelas, and S. Qc, "Developing Serious Games Specifically Adapted to People Suffering from Alzheimer," in Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics), no. 2, 2012, pp. 243– 254.
- [139] Defence Public Relations (ARMY), "British Army Vehicles and Equipment,"
 2010. [Online]. Available: http://www.army.mod.uk/documents/general/285986_ARMY_VEHICLESEQUIP MENT_V12.PDF_web.pdf. [Accessed: 01-Apr-2014].
- [140] A. G. Bruzzone, "Critical Issues in Advancing Modeling & Simulation," in SIMULTECH 2011 - 1st International Conference on Simulation and Modeling Methodologies, Technologies and Applications, 2011, pp. 9–15.
- [141] S. Chowdhury, P. Day, G. A. Taylor, S. P. Chowdhury, T. Markvart, and Y. H. Song, "Supervisory data acquisition and performance analysis of a PV Array installation with data logger," 2008 IEEE Power Energy Soc. Gen. Meet. -Convers. Deliv. Electr. Energy 21st Century, 2008.
- [142] Mental-images, "mental ray® Functional Overview Version 1.5," *Technical Report*, 2007. [Online]. Available: http://www.mentalimages.com/fileadmin/user_upload/PDF/mental_ray_Functiona l_Overview.pdf. [Accessed: 25-Apr-2012].
- [143] C. Halsted, "Brightness, Luminance and Confusion," *Inf. Dislplay*, vol. 9, no. 3, pp. 21–24, 1993.
- [144] Autodesk, "Daylight Simulation in 3ds Max Design 2009 Getting Started," *Technical Report*, 2009. [Online]. Available: http://images.autodesk.com/adsk/files/3dsmax_started.pdf. [Accessed: 01-Feb-2011].
- [145] Autodesk, "Daylight Simulation in 3ds Max Design 2009 Advanced Concepts," *Technical Report*, 2009. [Online]. Available: http://images.autodesk.com/adsk/files/3dsmax_advanced.pdf. [Accessed: 01-Feb-2011].
- [146] S. Darula and R. Kittler, "CIE general sky standard defining luminance distributions," *Proc. eSim*, 2002.
- [147] R. Perez, R. Seals, P. Ineichen, R. Stewart, and D. Menicucci, "A new simplified version of the perez diffuse irradiance model for tilted surfaces," *Solar Energy*, vol. 39, no. 3. pp. 221–231, 1987.
- [148] M. Sŭri, "Solar resource data and tools for an assessment of photovoltaic systems," *Technical Report: Office for Official Publications of the European Communities*, 2007. [Online]. Available: http://re.jrc.ec.europa.eu/pvgis/index.htm. [Accessed: 25-Mar-2013].

- [149] M. Donn, "Using Autodesk® 3ds Max® to Measure Real Daylight Performance," *Technical Report*, 2009. [Online]. Available: http://aucache.autodesk.com/au2009/sessions/5426/DV114-1_SpeakerHandout_Michael_Donn.pdf. [Accessed: 01-Feb-2011].
- [150] R. Thimijan and R. Heins, "Photometric, radiometric, and quantum light units of measure: a review of procedures for interconversion," *Hortic. Sci.*, vol. 18, no. 6, pp. 818–822, 1983.
- [151] Stroke Association, "What is a stroke? | Stroke Association," 2012. [Online]. Available: http://www.stroke.org.uk/factsheet/what-is-a-stroke. [Accessed: 15-Apr-2014].
- [152] DoH, "Reducing Brain Damage: Faster Access to Better Stroke Care," *Fiftysecond Report of Session 200506*, 2005. [Online]. Available: http://books.google.com/books?hl=en&lr=&id=u1hufT7bf3sC&oi=fnd&pg=PP7&dq=Reducing+Brain+Damage+:+Faster+access+to+better+stroke+care&ots=5tgp2JRrVX&sig=P5OY0mrjWee0VlSrUKbDCqvhksI. [Accessed: 03-Mar-2014].
- [153] R. Saka, A. McGuire, and C. Wolfe, "Economic burden of stroke in England," London Natl. Audit Off., pp. 1–29, 2005.
- [154] P. Fergus, K. Kifayat, S. Cooper, M. Merabti, and A. El Rhalibi, "A framework for physical health improvement using Wireless Sensor Networks and gaming," in *3rd International Conference on Pervasive Computing Technologies for Healthcare, 2009. PervasiveHealth 2009.*, 2009, pp. 1–4.
- [155] K. Liu, T. Liu, K. Shibata, Y. Inoue, and R. Zheng, "Novel approach for lower limb segment orientation in gait analysis using triaxial accelerometers," in *IEEE/ASME International Conference on Advanced Intelligent Mechatronics*, *AIM*, 2008, pp. 488–492.
- [156] M. Huang, E. Chen, W. Xu, and M. Sarrafzadeh, "Gaming for Upper Extremities Rehabilitation Categories and Subject Descriptors," *Proc. 2nd Conf. Wirel. Heal.*, pp. 9–10, 2011.
- [157] S. O. H. Madgwick, A. J. L. Harrison, and R. Vaidyanathan, "Estimation of IMU and MARG orientation using a gradient descent algorithm," in *IEEE International Conference on Rehabilitation Robotics*, 2011.
- [158] S. Sessa, M. Zecca, Z. Lin, T. Sasaki, K. Itoh, and A. Takanishi, "Waseda Bioinstrumentation System #3 as a tool for objective rehabilitation measurement and assessment - Development of the inertial measurement unit," in 2009 IEEE International Conference on Rehabilitation Robotics, ICORR 2009, 2009, pp. 115–120.
- [159] Medtronic, "About Parkinson's Disease," 2012. [Online]. Available: http://www.medtronic.com/patients/parkinsons-disease/index.htm. [Accessed: 19-Nov-2013].
- [160] Parkinson's_Foundation, "Statistic's on Parkinson's," 2013. [Online]. Available: http://pdf.org/en/parkinson_statistics. [Accessed: 19-Nov-2013].
- [161] P. Andlin-Sobocki, B. Jönsson, H.-U. Wittchen, and J. Olesen, "Costs of disorders of the brain in Europe.," *Eur. J. Neurol.*, vol. 14, no. 2, p. e9, Feb. 2007.

- [162] HESonline, "Hospital Episode Statistics on Parkinson's HESonline NHS," 2013.
 [Online]. Available: http://www.hesonline.nhs.uk/Ease/servlet/ContentServer?siteID=1937&categoryI D=868. [Accessed: 19-Nov-2013].
- [163] Parkinson'sUK, "Parkinson's prevalence in the United Kingdom," 2009.
 [Online]. Available: http://www.parkinsons.org.uk/pdf/ParkinsonsprevalenceUK.pdf.
- [164] J. Jankovic, "Parkinson's disease: clinical features and diagnosis.," J. Neurol. Neurosurg. Psychiatry, vol. 79, no. 4, pp. 368–76, Apr. 2008.
- [165] C. L. Tomlinson, S. Patel, C. P. Herd, C. E. Clarke, R. Stowe, and N. Ives, "Physiotherapy intervention in Parkinson's disease : systematic review and metaanalysis," *BMJ*, vol. 5004, no. August, pp. 1–14, 2012.
- [166] V. A. Goodwin, S. H. Richards, R. S. Taylor, A. H. Taylor, and J. L. Campbell, "The Effectiveness of Exercise Interventions for People with Parkinson's Disease : A Systematic Review and Meta-Analysis," vol. 23, no. 5, pp. 631–640, 2008.
- [167] M. van Nimwegen, A. D. Speelman, E. J. M. Hofman-van Rossum, S. Overeem, D. J. H. Deeg, G. F. Borm, M. H. L. van der Horst, B. R. Bloem, and M. Munneke, "Physical inactivity in Parkinson's disease," *J Neurol*, no. 258, pp. 2214–2221, 2011.
- [168] B. G. Farley and G. F. Koshland, "Training BIG to move faster: the application of the speed-amplitude relation as a rehabilitation strategy for people with Parkinson's disease.," *Exp. Brain Res.*, vol. 167, no. 3, pp. 462–7, Dec. 2005.
- [169] B. G. Farley, C. M. Fox, L. O. Ramig, and D. H. Mcfarland, "Intensive Amplitude-specific Therapeutic Approaches for Parkinson's Disease," *Top. Geriatr. Rehabil.*, vol. 24, no. 2, pp. 99–114, 2008.
- [170] L. O. Ramig, C. Fox, and S. Sapir, "Parkinson's disease: Speech and voice disorders and their treatment with the Lee Silverman Voice Treatment," *Seminars in Speech and Language*, vol. 25, no. 2. pp. 169–180, 2004.
- [171] E. Tsekleves, D. Skordoulis, I. Paraskevopoulos, and C. Kilbride, "Wii your health: a low-cost wireless system for home rehabilitation after stroke using Wii remotes with its expansions and blender," *Proc. 8th IASTED Int. Conf. Biomed. Eng. Biomed 2011*, p. Pages: 502–509, 2011.
- [172] E. Tsekleves, D. Skordoulis, I. Paraskevopoulos, C. Kilbride, and A. Warland, "Personalised stroke rehabilitation intervention using open source 3D software and the Wii Remote Plus," in *Proc. 9th Intl Conf. Disability, Virtual Reality & Associated Technologies*, 2012.
- [173] M. K. Holden and T. Dyar, "Virtual environment training: a new tool for neurorehabilitation.," *Neurol. Rep.*, vol. 26, no. 2, pp. 62–72, 2002.
- [174] R. Webber and B. Ramaswamy, "Keeping Moving Exercise and Parkinson's," 2009.
- [175] I. Paraskevopoulos and E. Tsekleves, "Simulation of photovoltaics for defence applications: power generation assessment and investigation of the available

integration areas of photovoltaic devices on a virtual infantryman," in *SIMULTECH 2012. International Conference on Simulation and Modeling Methodologies, Technologies and Applications*, 2012, pp. 384–396.

- [176] British Sky Broadcasting Group plc, "Watch Ross Kemp in Afghanistan Online | Sky Go," 2008. [Online]. Available: http://go.sky.com/vod/content/SKYENTERTAINMENT/Browse_by_Channel/Sk y1/content/seriesId/fd5fde775c0a7110VgnVCM1000002c04170a____/conte nt/default/promoPage.do. [Accessed: 02-Apr-2014].
- [177] S. Fahn and R. Elton, UPDRS program members. Unified Parkinsons Disease Rating Scale, Recent dev. Florham Park: Macmillan Healthcare Information, 1987, pp. 153–163.
- [178] C. G. Goetz, W. Poewe, O. Rascol, C. Sampaio, G. T. Stebbins, C. Counsell, N. Giladi, R. G. Holloway, C. G. Moore, G. K. Wenning, M. D. Yahr, and L. Seidl, "Movement Disorder Society Task Force report on the Hoehn and Yahr staging scale: status and recommendations.," *Mov. Disord.*, vol. 19, no. 9, pp. 1020–1028, 2004.
- [179] M. F. Folstein, S. E. Folstein, and P. R. McHugh, "'Mini mental state'. A practical method for grading the cognitive state of patients for the clinician," *J. Psychiatr. Res.*, vol. 12, no. 3, pp. 189–198, 1975.
- [180] J. Nielsen, "How Many Test Users in a Usability Study?," 2012. [Online]. Available: http://www.useit.com/alertbox/number-of-test-users.html. [Accessed: 31-Mar-2014].
- [181] J. Nielsen, "Why You Only Need to Test With Five Users," 2000. [Online]. Available: http://www.useit.com/alertbox/20000319.html. [Accessed: 31-Mar-2014].

APPENDIX

Appendix A. The Solar Soldier simulation results – Average Light Harvest



Figure 1A Forest scene in Baghdad



Figure 2A Forest scene in Catterick Garrison



Figure 3A Forest scene in Pristina



Figure 4A Military Base scene in Baghdad



Figure 5A Military Base scene in Catterick Garrison



Figure 6A Military Base scene in Pristina

10:00 12:00 14:00

16:00

10:00 12:00 14:00 16:00

Backeac

Backpac


Figure 7A Urban Area scene in Baghdad



Figure 8A Urban Area scene in Catterick Garrison



Figure 9A Urban Area scene in Pristina



Appendix B. The Solar Soldier simulation results – Average Power Generation

Figure 1B Forest scene in Baghdad



Figure 2B Forest scene in Catterick Garrison



Figure 3B Forest scene in Pristina



Figure 4B Military base scene in Baghdad



Figure 5B Military base scene in Catterick Garrison



Figure 65B Military base scene in Pristina



Figure 7B Urban area scene in Baghdad



Figure 8B Urban area scene in Baghdad



Figure 9B Urban area scene in Baghdad

Appendix C. Patient Questionnaire

Questionnaire: Exploring the Use of the Nintendo Wii in Rehabilitation)]	10 C 4 10	
Please tick one box for each question. Please feel free to get someone to help you fill out this if required.	5		2 M

About You

1	Age:	45-54 76-80	55-64 80-85	☐ 65-69 86 +	70	-75
2	Are you:	Male	Female			
3	Have you p	layed video gam	es before?	Yes	No No	Don't know
4	Have you u	sed the Nintend	o Wii before?	Yes	No	Don't know

Rowing game

5 Does	the rowing game have clear and simple directions? es 🔲 No 📄 Don't know
6 How e	easy or difficult was it to play the rowing game? difficult 1 2 3 4 5 Very easy 💓
7 How f	fun or boring was it to play the rowing game? Doring 1 2 3 4 5 Very fun
8 What	do you think about the game speed? oo slow Slow Just right Fast Too fast

Please turn to the next page to complete the questionnaire

1 of 6

9 How useful was the audio feedback (e.g. sounds) you received whilst playing the game?
Not useful 1 2 3 4 5 Very useful 😳
10 How useful was the visual feedback (e.g. arrows, paddle movement) you received whilst playing the game?
11 How useful was the feedback you received at the end of the game?
Not useful 1 2 3 4 5 Very useful 💓
12 Bate your overall experience in playing the rowing game
Not so good 1 2 3 4 5 Very good
13 Do you feel the the rowing game requires a lot of arm movement? Image: Too little A little Just right A lot Too much
14 Do you feel the rowing game helps you with exercising your arms? •••••••••••••••••••••••••••••
15 If you had access to the rowing game at home, would you use it to do arm exercises? Image: Yes No Don't know
16 Would you prefer holding the Wii remote to play the rowing game? Image: Yes Image: No Image: Don't know

17 Would you like to play the rowing game again?
Yes If you have answered No, please skip ahead to question 19
18 If yes how often would you play the rowing game? Image: every day Image: every other day Image: every day Image: every other day Image: every day Image: every other day Image: every day Image: every day <t< td=""></t<>
19 Do you feel that your balance is affected while you play the rowing game? Image: Pres, it's affected is affected in the positively is affected in the positive posi
20 Do you prefer playing the rowing game while seated or standing up?
21 What did you like the most about the rowing game?
22 What improvement would you suggest for the rowing game?

Rolling game

23 Does the rolling game have clear and simple directions?		
Yes No Don't know		
How easy or difficult was it to play the rolling game?		
Very difficult 1 2 3 4 5 Very easy		
25 How fun or boring was it to play the rolling game?		
Very boring 1 2 3 4 5 Very fun		
26 What do you think about the game speed?		
Too slow Slow Just right Fast Too fast		
27 How useful was the audio feedback (e.g. sounds) you received whilst playing the game?		
Not useful 1 2 3 4 5 Very useful 💓		
28 How useful was the visual feedback (e.g. arrows, wheel movement) you received whilst playing the game?		
Not useful 1 2 3 4 5 Very useful 💓		
29 How useful was the feedback you received at the end of the game?		
Not useful 1 2 3 4 5 Very useful 💓		
30 Rate your overall experience in playing the rolling game.		
Not so good 1 2 3 4 5 Very good 😳		

31 Do you feel the the rolling game requires a lot of arm movement?
32 Do you feel the rolling game helps you with exercising your arms? Strongly disagree 1 2 3 4 5 Strongly agree
33 If you had access to the rolling game at home, would you use it to do arm exercises? Image: Second se
34 Would you prefer holding the Wii remote to play the rowing game? Image: Yes Image: No Image: Don't know
 Would you like to play the rolling game again? Yes No If you have answered No, please skip ahead to question 37
36 If yes how often would you play the rolling game? Image: every day every other day once a week once a month
37 Do you feel that your balance is affected while you play the rolling game? Image: Yes, it's affected in the positively Yes, it's affected in the positively
38 Do you prefer playing the rolling game while seated or standing up? Image: Seated image: Standing image: Seated image: Standing image: Standimage: Standing image: Standing image: Standing



40 What improvement would you suggest for the rolling game?

This is the end of the questionnaire, thank you for completing the questionnaire.

6 of 6